# A TEACHER'S PERCEPTIONS OF LANGUAGE LEARNING AND SOCIAL PRESENCE IN A BLENDED LEARNING COMMUNITY COLLEGE ESL COURSE

by

Allison Omohundro
A Dissertation
Submitted to the
Graduate Faculty
of
George Mason University
in Partial Fulfillment of
The Requirements for the Degree
of
Doctor of Arts
Community College Education

# A Teacher's Perceptions of Language Learning and Social Presence in a Blended Learning Community College ESL Course

A Dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Arts at George Mason University

by

Allison Omohundro Master of Special Education Longwood University, 2006 Bachelor of Liberal Studies Longwood University, 2005

Director: Kelly Schrum, Associate Professor Higher Education Program

> Fall Semester 2019 George Mason University Fairfax, VA

Copyright 2018 Allison Omohundro All Rights Reserved

# **DEDICATION**

This dissertation work is dedicated to my family. A special feeling of gratitude to my parents, Dan and Karen Sauveur whose words of encouragement and pride carried me through my journey in education. My sister Kristin has supported me while setting an exemplar for hard work and persistence. I also want to give special thanks to my husband William Omohundro for being there for me throughout the entire doctorate program. You have been my greatest support system.

# **ACKNOWLEDGEMENTS**

Completion of this doctoral dissertation is possible with the support of several people. I would like to express my sincere gratitude to all of them. First of all, I am extremely indebted to my chair, Dr. Kelly Schrum, for her valuable guidance, scholarly inputs, and consistent encouragement I received throughout the research work. I benefitted greatly from the assistance of other committee members during my graduate work, Dr. Amy Swan and Dr. Steven Weinberger. Thanks also to my colleagues at DPES, especially Betsy Whiting, who directed me towards the practice of blended learning.

# **TABLE OF CONTENTS**

	Page
List of Abbreviations	
Abstract	
Chapter One: Introduction	
Background	
Blended Learning in Community Colleges	4
Building Community and Social Presence in Language Learning	7
Statement of the Problem	9
Purpose	11
Significance	16
Organization of Study	18
Definition of Terms	18
Chapter Two: Literature Review	22
Introduction	22
The Flipped Model of Blended Learning	23
Social Presence in Blended Learning Research	26
Design Guidelines of Blended Learning for Language Learning	29
Blended Learning in ESL Higher Education	34
Benefits of Blended Learning in ESL Higher Education	36
Challenges of Blended Learning in ESL Higher Education	38
Blended Learning in ESL Community College Higher Education	40
Teacher Perceptions of Language Learning	42
Teacher Perceptions of Social Presence	45
Relationship of the Literature to the Study	47
Chapter Three: Methodology	
Introduction	
Research Method and Design	

Population and Setting	54
Region Characteristics	55
NVCC Characteristics	56
Campus and ESL Program Characteristics	58
Participant Profile	59
Background	60
Procedures and Data Collection	61
Data Analysis	63
Researcher as Instrument Statement	65
Summary	68
Chapter Four: Findings	69
Learning Environment	70
Classroom Structure and Assignments	
Nature of Hybrid Courses	75
Organization of F2F Format	76
Explicit Instructions	77
The Use of Advertisements	79
Literacy Instruction with Short Stories	82
Language Instruction Around Song	84
LMS Utilization and Online Activities	86
Discussion Board	86
Short Stories	88
Communication Across Modes	89
Student-to-Student	89
Student-Teacher	90
Challenges	93
Technology Skills	93
Following Directions	95
Plagiarism	97
Summary	101
Chapter Five: Discussion	
Introduction	104

Summary of Study	104
Discussion of Findings	106
Components in Promoting Language Learning in Blended Learning Research	108
F2F Format	108
Learning Management System	110
Communication	112
Social Presence	117
Implications for Practice and Policy	120
Limitations	123
Future Research	124
Conclusion	126
Appendix A: Faculty Pre-Interview Questions	129
Appendix B: CoI Survey Instrument	131
Appendix C: Faculty Post-Interview Questions	134
Appendix D: Advertisement Project Rubric	136
eferences	. 138

# LIST OF ABBREVIATIONS

Blended Learning	BL
Computer-Assisted Language Learning	
Community of Inquiry	COI
English as a Second Language	
Face-to-Face Instruction	F2F
Learning Management System	LMS

**ABSTRACT** 

A TEACHER'S PERCEPTIONS OF LANGUAGE LEARNING AND SOCIAL PRESENCE IN A BLENDED LEARNING COMMUNITY COLLEGE ESL COURSE

Allison Omohundro, D.A.

George Mason University, 2019

DissertationDirector: Dr. Kelly Schrum

This dissertation will address the views of an experienced instructor of a blended learning

(BL) community college English as a Second Language environment. The research will

explore the BL community college setting from the perspective of an instructor on how

the fostering and action of social presence impacts the course. The aim of the study is to

understand how an English as a Second Language instructor plans for language learning

and fosters social presence so that ideas might be developed for the successful

implementation of BL community college environments in the future that will support

English language learning.

#### **CHAPTER ONE: INTRODUCTION**

### **Background**

A sense of classroom community is defined as the "spirit, trust, interaction, and commonality of expectations and goals" (Rovai, 2002, p.11) that occurs within a classroom environment, either virtual or physical. A sense of classroom community has been found to correlate with important educational outcomes, including retention and academic achievement (Abedin, Daneshgar, & D'Ambra, 2010; Drouin, 2008; Lamport & Bartolo, 2012; Liu, Gomez, & Yen, 2009; Pichon, 2016; Wang, 2010). Yet many English as a Second Language (ESL) students feel disconnected from their classroom learning community (Jiménez & Rose, 2010) which can have an impact on motivation, interest, and achievement (Dörnyei and Csizer, 2002). Peirce (1995) spoke in terms of investment for ESL students in the classroom:

...for a new learner of English to take the risk of using the fledgling language in spite of the fear of being misunderstood or laughed at, the learner must believe that there is a substantive payoff in language use. The user's social identity in the new language and new culture is being formed, and for the time being, the new identity is fragile (p. 17).

Instructors need to develop social communities of learners that both support students and incorporate the identities that ESL students bring to the classroom (Rance-Roney, 2009).

Many practitioners suggest that blended learning (BL) is an avenue to explore in order to boost learner engagement (Tay, 2016; Woltering, Herrler, Spitzer, & Spreckelsen, 2009; Wu, Tennyson, & Hsia, 2009) and in the process encourage ESL students to become a part of the classroom community (Senior, 2010) and develop a social presence, an ability to identify and communicate with the community (Garrison, 2009) during class interactions.

The term blended learning is generally applied to the practice of combining both online and face-to-face learning experiences when teaching students within one course. In a blended-learning environment, for example, students would meet with a teacher in a traditional classroom setting, while also independently completing online assignments for the course outside of the classroom (Oliver & Trigwell, 2005). The delivery of the online part of the course is usually through learning technologies, typically involving a Learning Management System (LMS), such as Blackboard or Canvas, and the use of synchronous and asynchronous electronic tools (Bersin, 2004).

BL is a bridge for all universities between e-learning and traditional teaching (Hubackova, 2015). Because e-learning came first, its origins are important to BL. "E-learning is all forms of electronic supported learning and teaching, which are procedural in character and aim to affect the construction of knowledge with reference to individual experience, practice, and knowledge of the learner" (Tavangarian, Leypold, Nölting, Röser, & Voigt, 2004, p. 274). The belief has been gaining ground that the most constructive way to implement e-learning is not as a singular entity, but as the integrated process of BL (Hubackova, 2015). In the case of BL, the role of the teacher is not

replaced with online instruction but rather is enhanced. The role of the instructor becomes one of facilitator rather than lecturer (Bauer-Ramazani, Graney, Marshall, & Sabieh, 2016; Han, 2015; Webb, Doman, & Pusey 2014).

An important element for fostering social presence and a sense of community in blended learning is the instructor's involvement in both virtual and physical settings. It is also important for instructors to push their students to engage in critical analyses and higher-order thinking in the online environment (Whiteside, 2015). Teachers' instruction influences students' perceptions (Asri Humaira, Rasmitadila, Widyasari, & Rachmadtullah, 2019), so the implementation and use of the different resources and tools of a course will impact students' perceptions of how important and useful these resources and tools are to their language learning (Comas-Quinn, 2011). Instructors are still navigating the models of a BL environment and are continually learning more about how to maximize student learning in a multimodal program with a learning management system, social media tools (Abdillah, 2016; Aurangzeb, 2018; Comas-Quinn, 2011; Woodley & Meredith, 2012), and traditional classroom tools (Bender, 2012). Blended learning programs can be challenging because instructors and learners are maneuvering among the different modes of BL which requires a lot of adaptability and increases the need for social presence (Whiteside, 2015).

Different from e-learning courses where students may experience the lack of presence of a teacher and other classmates, a BL design offers features of e-learning, and incorporates components from traditional education that promote student success, such as group work and one-on-one discourse with a teacher (Hotle & Garrow, 2015). McGuire

(2016) noted fully online instructors have great difficulty in facilitating quality interactions among students in asynchronous online discussion forums, but they also mentioned challenges of interacting with their students individually, which accords with previous studies of faculty perception of online instruction (Chiasson, Terras, & Smart, 2015). As Deggs, Grover, and Kacirek (2010) noted, faculty's perceptions of their effectiveness in an online class are challenged by students, particularly if the faculty does not engage the students in discussion or if they are slow in providing timely and relevant feedback (p. 698).

An instructor who is implementing BL in their classroom could benefit from understanding the definition and origin of BL as well as the varied models and which models best fit their learning outcomes for a specific course. More attention needs to center on the practitioner's use of BL regarding their perception of effective instruction for social presence and language learning. In addition, focus needs to shift in BL research to community colleges where instructors tend to face more challenges in the implementation and delivery process than in 4-year institutions (Crawford, Barker, & Seyam, 2014; Crookston & Hooks, 2012).

# Blended learning in community colleges.

Today's community colleges are responsible for improving social mobility despite growing economic inequality while also working to make the U.S. more globally competitive (Century Foundation Press, 2013). Two-year colleges face the task of educating students who have the greatest academic and economic needs while simultaneously receiving lower financial support than 4-year institutions (Century

Foundation Press, 2013; Desrochers & Wellman, 2011). Practitioners need research-based information about the effectiveness of BL methods, especially at the community college level. While the terms blended learning and hybrid learning are used interchangeably at community colleges and varying definitions exist, this paper uses the definition by Rovai and Jordan (2004) in which BL is "a mix of classroom and online learning that includes some of the conveniences of online courses without the complete loss of face-to-face contact" (p. 1).

Community colleges can benefit from a hybrid of virtual and physical learning (Snart 2017). Flexibility is an important asset that community college students are seeking when choosing courses (Jaggars, 2014, Snart, 2017). Community college instructors need to design and implement courses that reflect the diverse student population. Additionally, educators need to provide support for technology on the LMS and online resource support for students when they are outside of the classroom (Snart, 2017).

The affordability of community colleges attracts many multilingual and international students seeking to improve their English language skills (Hagedorn & Li, 2017; Yueh-ching, 2016). In fact, ESL is one of the fastest growing programs in many community colleges and across all types of adult education programs (Community College Consortium for Immigrant Education, 2015). The site for this research is among the top five community colleges with the highest numbers of international student enrollments (Institute of International Education, 2017).

BL has been widely used in English language teaching since at least 2007, when Sharma and Barrett wrote about blended learning tools and applications that may lead to language learning success in the classroom if used appropriately and knowledgeably. According to Watanabe and Swain (2008), multiple studies have examined the nature of peer-peer instructional conversation and its significance to ESL learning from a socio-cultural theory perspective. ESL instructors need to incorporate a more student-centered approach, encouraging students to have conversations with each other in and outside of the classroom (Yueh-ching, 2016). The socio-cultural nature of teaching and learning in ESL classes requires instructors to take on the role of the cultural broker in ESL classrooms. Lack of awareness of students' identities associated with their unique, but connected, communities could impact a student's interest in the language practices of the classroom (Darvin & Norton, 2014). Instructors must find strategies that deal with individual differences in learners through using instruction that meet the needs and characteristics of ESL students.

Research into the area of the use of technology that has potential to boost student achievement with ESL students is relevant and needed (Blattner & Lomicka, 2012; Chen, 2016; Silverman & Hines, 2009). Diverse forms of online communication, including email messages, discussion forums, online chatting, and social networking, have supplemented the teaching and learning of second and foreign languages over the last two decades (Fotos & Browne, 2004; Kern & Warschauer, 2000; Paepe, Zhu, Depryck, 2018). Despite these varied communication styles, there is little research on the

development of social community within an online environment blended in the ESL community college setting.

# Building community and social presence in language learning.

Following the sociocultural tradition in language and learning research (Vygotsky, 1978; Wenger, 1998) in the last decades, a number of ESL researchers have begun to explore students' online language tasks by observing the interaction between learners and their sociocultural contexts (Slavkov, 2015). Learning is seen as a process of becoming a member of a certain community (Lave, 1991, p. 65), including gaining "the ability to communicate in the language of this community and act according to its norms" (Sfard, 1998, p. 6). In an educational institution, ESL students are seen to participate in "negotiated literacy" (Canagarajah, 2013, p. 61) by seeking meaningful learning and defining their group role based on the benefits and limitations of the contexts in which they find themselves, including cultural norms, social expectations, and institutional ideologies, as well as students' goals, personal histories, and familial obligations (Fuchs, 2016; Jang & Jiménez, 2011; Yu and Lee, 2016). ESL research has begun to explore the advantages that online communication provides to adult ESL learners in order to make language learning meaningful and sustainable (Coryell & Chlup, 2007; Hampel & Stickler, 2005; Paepe et al., 2018). The studies examined involve primarily online courses and lack a range of ethnically and culturally diverse participant groups. Further research needs to concentrate on BL courses in conjunction with the ways in which ESL instructors consider the sociocultural background of each student when planning for collaborative activities.

Digital communications have been used in second language teaching since the early 1990s. Chapelle (2001) stated nearly two decades ago, "Everyday language use is so tied to technology that learning language through technology has become a fact of life" (p. 1). Presently, Kessler (2018) observes, "technology use has become so ubiquitous in our daily lives that the absence in our classroom is quite noticeable" (p. 206). The principal idea of BL in second language education is to achieve a more effective and efficient way to promote second language teaching and learning by combining two different but complementary modes: computer-assisted language learning (CALL) technologies and face-to-face (F2F) interaction (Hong & Samimy, 2010). Computer assisted language learning (CALL) is thought to be able to fulfil the principles of language teaching by providing more communicative practices with the use of technology (Mishan, 2005). In contrast to the traditional classrooms where face-to-face discussions are often held, the open-access online discussions enable every member to think critically about their posts before submitting them to classmates and instructors. It is also important, however, to draw on Warschauer (1996) to mark a distinction between the research conducted on classes with computer-assisted language learning (CALL) and blended courses. CALL is the collection of instructional tools, not instruction deliveries, and while it is possible for a BL course to incorporate CALL technologies, CALL by itself cannot constitute a blended course.

The aim of this study was to carefully examine the use of synchronous and asynchronous assignments in a BL course, using CALL tools, for different types of ESL learners, in order to enhance their English language acquisition and foster social

presence. Specifically, this research focused on how an experienced ESL instructor perceived the use of technology and course materials in a blended class to involve students in face-to-face and online discussions and foster a social presence while promoting language learning. While research in ESL and collaboration in academic contexts has shed light on learning language, blended learning environments, in which ESL students engage in English and make it meaningful to them, have received far less attention. With a social presence perspective, this qualitative case study examined the discursive practices of a blended community college ESL classroom and of its online discussion forums and assignments with the goal of learning what instructional practices are most and least effective in promoting language learning.

#### **Statement of the Problem**

ESL students are currently the fastest growing population of students in the United States (Barr, Eslami, Joshi, Slattery, & Hammer, 2016; Kanno & Kangas, 2014; Dobbins & Rodriguez, 2013). In 2006, Fu and Matoush reported, "the number of English language learners in the U.S. over age five has grown from 23 million to 47 million, or by 103 percent" (p. 10). ESL students enrolled at colleges and universities have diverse educational experiences and linguistic abilities. In addition, ESL students have unique motivations for learning English. Meeting ESL students' diverse needs can create instructional challenges in the classroom.

In community colleges, instructors face challenging work environments such as additional workloads for adjuncts, a need to teach a wide variety of courses, and lack of time for professional development (Brock et al., 2007; Calcagno, Bailey, Jenkins, Kienzl,

& Leinbach, 2008). Without knowledge of best practices and proper guidance on how to foster a social learning community, ESL instructors cannot adequately strive for successful language learning among their students. Technology offers a way to address the diversity of student needs within one class by allowing teachers to customize instruction in online platforms, often with limited burden on the teacher.

According to Graham, Allen, and Ure (2005), there are three main reasons for choosing BL: (1) improved pedagogy, (2) increased access/flexibility, and (3) increased cost effectiveness. In the review of the literature, it is apparent that blended learning means different things to different people and that a definition of BL "depends upon the context and purposes of the developers" (Dewar & Whittington, 2004, p.4). The reasons for implementation from Graham et al. (2005) seem to be speaking to the burdens that community college teachers face in their classrooms. These instructors as developers have an obligation to work within a minimal budget (Crawford et al., 2014; Crookston & Hooks, 2012), provide instruction to non-traditional students (Mullin, 2012), and maintain education derived from current best practices, which includes technology (Li, 2013). Studies indicate that community college English language teachers have more of a burden in developing courses under those stipulations due to learning various technological skills, staying updated in the field of English language teaching, and meeting the needs of an increasingly diverse generation of learners (Webb et al., 2014).

There is limited research on not only how the community college ESL teacher can blend both practices (Ingerson, 2011) successfully in a classroom, but also what factors impact the satisfaction and success for both themselves and the learner. Also, while a few

studies have looked at teachers' perceptions of blended learning and social presence (Ndlovu & Mostert, 2018), even fewer, if any, have examined whether teachers have a good understanding of how social presence is formed effectively in the learning community of an ESL community college environment. The challenge for ESL instructors is fostering effective social presence in the online portion of a BL setting where paralinguistic cues such as facial expression and intonation are often absent. The present study, therefore, aimed to examine the attitudes and perceptions of a teacher on F2F and online assignments meant to develop social presence and promote language learning in an ESL community college BL course.

This study is built on the previous studies of blended learning and social presence in a Community of Inquiry Framework (So & Brush, 2008; Jusoff & Khodabandelou, 2009). Despite the positive impact of social presence on learners' satisfaction with a course (Leafman, Mathieson, & Ewing, 2013), it is still unclear what role social presence plays in an ESL instructor's design and implementation process, as well as student's achievements in language learning. The previous literature also mainly investigated the role of social presence on students' satisfaction with online courses rather than in a blended course or the teacher's perspective. Further research that investigates the role of social presence in a BL community college ESL setting would be beneficial with the goal of understanding what creates social presence and promotes language learning.

# **Purpose**

This research aimed to recognize the rich social, linguistic, cultural, and academic potential that ESL learners develop when an instructor adapts curriculum and

assignments to encourage those capabilities. Practices and resources are needed to guide instructors on how technology can be used to effectively foster social presence and meet the language learning needs of ESL learners. The affective component of social presence exemplifies the use of humor, emotions, and self-disclosure (Leafman & Mathieson, 2014). The interactive element includes acknowledgement, appreciation, and return of communication among learners (Garrison, Cleveland-Innes, & Fung, 2010; Rourke, Anderson, Garrison, & Archer, 2001).

ESL students are ethnically and linguistically diverse and they bring a wide range of sociocultural backgrounds and learning styles to the class to facilitate their learning (Li, 2013). Accordingly, second language learning has begun to focus on incorporating the sociocultural context in ESL classroom settings to assist learners in their language development (Aimin, 2013; Behroozizad, Nambiar, & Amir, 2014; Lantolf & Thorne, 2006; Panahi, Birjandi, & Azabdaftari, 2013). By perceiving the language classroom as a learning community, ESL instructors can develop the construct of the "strategic classroom" that makes "maximum use of affective, cognitive, metacognitive, and social learner strategies to influence effective learning communities" (Takeuchi, Griffiths, & Coyle 2007, p. 91).

Deriving from socio-constructivism, Garrison, Anderson, and Archer's (2000)

Community of Inquiry Theory (CoI) provides a method for designing and maintaining the socially constructed nature of traditional classroom learning in virtual environments. Garrison et al. introduced their model as a "pragmatic organizing framework of sustainable principles and processes for the purpose of guiding online

educational practice," (Swan, Garrison, & Richardson, 2009, p. 5). CoI has gained momentum in instructional design and BL research that emphasizes theory to practice.

CoI has three main elements: cognitive, social, and teaching presence (Garrison, Anderson, & Archer, 2010, p. 6). The teaching presence, referring to how instructors sequence the learning activities and facilitate learning (Koh, Herring, & Hew, 2010), encompasses the design, direction, and support of student activities that can lead to a powerful learning experience (Rubin, Fernandes, & Avgerinou, 2013). Cognitive presence refers to the development of critical thinking skills (Scherer Bassani, 2011); the engagement with course concepts; and the ability to create meaning out of ideas, develop and build competence via discussion, and reflect and apply the newfound meaning (Rubin et al., 2013).

Liu et al (2009) found that social presence is a predictor of community college course retention and final grade achievement in online environments at community colleges. Social presence, an effective way to support the social and interpersonal communication required for online teaching and learning (Lowenthal & Dunlap, 2014), includes affective responses and expression, open communication with others during the course, and cohesive communicative responses (Ice, Swan, Diaz, Kupczynski, & Swan Dagen, 2010; Rourke et al., 2001; Rubin et al., 2013). Studies involving the CoI framework, specifically social presence, have largely focused on online learning exclusively. This study examined social presence in the blended environment with an ESL population at a community college.

BL has the potential to provide students a sense of belonging in the online portion of a course (Sorden & Munene, 2013) that can be provided by the fostering of social presence. There is also a strong correlation between social presence and student satisfaction (Barker, 2015; Downing, Spears, & Holtz, 2014; Hsieh, Wu, & Marek, 2017; Lee & Huang, 2018; Lyons, Reysen, & Pierce, 2012; McGuire, 2016; Sorden & Munene, 2013) that can increase student engagement. BL facilitators must promote an authentic learning community to engage students in their learning (Tu, Sujo-Montes, Yen, Chan, & Blocher, 2012). In order to retain and motivate ESL students, institutions and educators must continue to look for innovative ways to meet evolving student learning needs. While learning outcomes appear to stay constant in both F2F and BL environments (Crawford et al., 2014; Palmer, Shaker & Hoffman-Longtin, 2014; Ryan, Kaufman, Greenhouse, She, & Shi, 2016; Xu & Jaggars, 2013), students in blended-learning environments have reported increased satisfaction with a strong, collaborative community in both the online and face-to-face components (Aragon & Johnson, 2008; Dixson, 2010; Xu & Jaggars, 2013). Oliver and Trigwell (2005) have attributed the student satisfaction advantage to variation theory to explain the increase in learning as resulting from the blending of instructional approaches. Their study explored how the three factors of social presence, collaborative learning, and student satisfaction correlate with each other in BL environments.

In online collaborative learning, strategies promoting the feeling of connectedness and belonging have appeared to be critical for learners (Hara, Bonk, & Angeli, 2000; Kitchen & McDougall, 1998). As a result, social presence can be considered a social and

communication factor that is particularly critical to distance learners' perception of psychological distance with their instructor and other learners (Gunawardena & McIsaac, 2004). Closely tied to social presence is the concept of instructor immediacy (Sung & Mayer, 2012) which applies to both face-to-face and online facilitation, and is generally defined as verbal and non-verbal behavior that reduces the psychological and transactional distance (Baker, 2004) An important argument in the earlier social presence theory is that different types of communication media have different abilities to affect an individual's perception of social presence (Gunawardena & Zittle, 1997). Early in social presence research, Gunawardena (1995) noted that practitioners who engage in studying skills and techniques related to fostering social presence will have the greatest impact on student perceptions. Therefore, there is a need for research on collaborative language assignments used in blended environments to determine effectiveness for social presence and positive perceptions.

The purpose of this qualitative study was to explore the perceptions of an ESL teacher in a blended learning environment on fostering language learning and social presence. The teacher perception focused initially on why they developed assignments in the blended setting and how those assignments impacted language learning. The second focus was on the teacher's perception of how their course fostered social presence within the face-to-face and online modes. The study was conducted at a community college campus in Northern Virginia in an ESL blended course (hybrid) in the area of reading. This study sought to broaden the existing body of research in language learning, social presence, and blended learning by examining an ESL instructor's attitudes towards

the constructs and evaluation of their students' learning during a spring semester at the community college level by answering the following research questions:

RQ1: What is the approach of an experienced community college English as a Second Language teacher to structuring a blended learning course to promote language learning?

RQ2: What are the teacher's perceptions of assignments that most effectively fostered social presence and promoted language learning in an English as a Second Language blended learning community college course?

RQ3: What are the teacher's perceptions of assignments that least effectively fostered social presence and promoted language learning in an English as a Second Language blended learning community college course?

# **Significance**

In a 2010 EDUCAUSE report titled "Blended Learning: A Report on the ELI (EDUCAUSE Learning Initiative) Focus Session," authors Diaz and Brown suggest that "over the past 10 years, blended learning has matured, evolved, and become more widely adopted by institutions of all types" (p. 2). The community college's diverse setting presents an ideal opportunity for blended teaching and learning to serve ESL students in an impactful way. As the Diaz and Brown report goes on to say, blending online and face-to-face learning allows institutions to "address learners' specific needs and customize the learning environment rather than rely on a one-size-fits-all approach" (p. 2).

While there is an abundant amount of research on the integration of technology in language learning, focus on the effects of a blended approach on ESL students at the community college level is minimal. Research that specifically addresses ESL students tends to focus primarily on online assignments in place of the F2F, rather than a hybrid of both instructional types (Al-Jarf, 2004; Campbell, 2007). A focus on quality blended instructional deliveries is imperative in order to implement BL courses into community colleges across the United States to fully promote language learning achievement.

The unique feature of blended classes is that they offer two instructional delivery methods. Educators who advocate online-only instruction often cite the opportunity for students who might normally be shy or feel intimidated in a face-to-face classroom to find their voice through online assignments. This, they claim, allows marginalized students to establish themselves as part of the classroom community (Palmer, Holt, & Bray, 2008). Other researchers argue, however, that online communicative assignments, such as discussion boards, negatively impact students who feel self-conscious and will cause them to limit their participation (Zhao & McDougall, 2008). Previous research suggests that an LMS, such as Blackboard, is structured to facilitate formal and monitored communication, and is therefore unable to provide adequate and meaningful social presence for participants if used in isolation (Brazington, 2012).

Students want to feel connected with their teachers and other students in an online class (Dikkers, Whiteside, & Lewis, 2013). Social presence is a strategy that faculty can use to increase student involvement and engagement in the blended classroom when collaboration occurs online that may have a positive effect on student success, such as

persistence and academic achievement. Social presence, concerning ESL learners, is found to be more critical than the other two constructs in the CoI framework (Estrada, Doupoux, & Wolman, 2005) because language learners tend to have more of a need to establish a connection in the community in order to interact with peers (Becerra, 2012). BL could potentially solidify the classroom community by providing students differing ways to communicate, allowing them to choose the best and most comfortable ways to express themselves. Research in BL needs to begin to aid educators in gaining insight in how to foster a learner's sense of connectedness in order to promote student language learning in the community college setting.

# **Organization of Study**

This dissertation research will be presented in five chapters. Chapter Two provides a review of the literature, including the impact of social presence in BL environments and the challenges instructors and students face in those courses. Chapter Three presents the methodology of the study, including participants and setting, measurement and research design, procedures, data collection, and analysis. Chapter Four presents the study's findings and results of data analysis for the faculty pre/post interviews, document analysis, and classroom observations. Chapter Five provides a discussion of the findings, implications of the findings for theory and practice, recommendations for future research, and conclusions.

## **Definition of Terms**

For clarification and to assist the reader of this study, definitions and explanations are provided of important terms used in the research.

#### Asynchronous

Learning happens when a time lag exists between the presentation of instructional material and student responses to that material (Means, Toyama, Murphy, Bakia, & Jones, 2009).

# Blended Learning

Although multiple definitions exist, and the term 'hybrid' is also used synonymously in research, this study uses the meaning by Rovai and Jordan (2004) where BL is "a mix of classroom and online learning that includes some of the conveniences of online courses without the complete loss of face-to-face contact" (p. 1). *Collaboration* 

In order to narrow the focus of collaboration to its value in education, the study employs the definition by Roschelle and Teasley (1995) where they state that collaboration is "The mutual engagement of participants in a coordinated effort to solve a problem together" (p. 70). The academic work to be examined in this study will be taken from both F2F and virtual tasks.

# Community of Inquiry Model

The Community of Inquiry Model was developed through an emphasis on three elements of presence: teaching, cognitive, and social presence. The research in this study will focus on the area of social presence exclusively. This model represents the space where interaction occurs online between learners, teachers, and the learning material (Garrison, Anderson & Archer, 2001, p. 88). The social presence component will also be observed in the F2F environment for this study.

#### Computer-Assisted Language Learning

Computer-Assisted Language Learning, according to Beatty (2003) is "Language learning in which a learner uses a computer and, as a result, improves his or her language" (p. 7). While BL is a type of instructional delivery, CALL applications serve as avenues to deliver the type of online tasks in which instructors would like to have students engage virtually.

#### ELL

English Language Learners, or non-native English speakers learning English (Bifuh-Ambe, 2009).

#### EFL Learner

English as a Foreign Language, a learner taught English in a country where it is not a first language (Shawer, 2010; Wu & Alrabah, 2009).

# ESL Learner

English as a Second Language learner. The learning is usually taking place in a country that predominantly speaks English (Warren & Miller, 2015). The other names used in research are English Language Learner and English as a Foreign Language Learner (EFL). ESL will be used in this study.

# Social Presence

Social presence is "the ability of participants to identify with the community, communicate purposefully in a trusting environment, and develop interpersonal relationships by way of projecting their individual personalities" (Garrison, 2009, p. 352). For the purpose of this study, the community refers to the BL course in which the

participant is enrolled. Social presence subset categories included personal/affective, open communication, and group cohesion.

# Synchronous

Palloff and Pratt identify communication as synchronous when "those communicating do so at the same time" (1990, p.189). Synchronous communication can be done in a virtual or F2F setting.

#### CHAPTER TWO: LITERATURE REVIEW

#### Introduction

This chapter presents the rationale for conducting research on the effect of blended learning instruction in English as a Second Language (ESL) classrooms at the community college level. The review of the literature includes studies that focus on the use of social presence in blended ESL environments in addition to strictly online environments with traditional college students. Studies in this literature review also focus on pedagogy that promotes language learning in a blended learning environment. The studies have contributed to the research on social presence and language learning because they provided direction for methodology, extensive literature review, and focus for the purpose and significance of this study.

Additionally, to support the review on fostering social presence in BL, the literature includes studies on teacher perceptions of social presence and language learning in BL and the benefits and challenges to BL that might influence those feelings and learning outcomes. Literature on perceptions are included because perceived social presence by students has been positively associated with student learning outcomes (Hostetter & Busch, 2013; LaPointe & Gunawardena, 2004; Liu et al., 2009; Swan et al., 2008). However, research is limited on teacher perceptions of social presence and language learning when designing a course which therefore guides the research questions for this study. The body of literature selected is based on relevance to the topics of social presence and a sense of community in blended English language learning classrooms.

The literature review in this study discloses various perceptions of social presence and language learning and the value of social presence in learning communities while delivering quality blended instruction.

# The Flipped Model of Blended Learning

A BL environment could look different in every classroom. A BL model can be implemented at an activity level, course level, program level, or institutional level (Graham, 2006; Osguthorpe & Graham, 2003). This research focused on the course level assignments, and the impact of those assignments on a teacher's perception of social presence and quality language learning in a blended learning environment. Lee and Wallace (2017) investigated language learning in a blended classroom and reported that flipped learning allows students autonomy in their own learning and reinforces their comprehension, performance, and confidence through a repeated process. While most literature on the impact of beliefs on language learning is situated within the traditional classroom, investigations on how teacher's beliefs influence language learning in a flipped classroom could expand the literature by enacting the flipped classroom approach in an English language classroom.

In 2011, Staker and Horn profiled forty organizations that offered a blended learning environment within brick-and-mortar classrooms and identified six models of blended learning. In 2012, they further narrowed the models down from six to four structures: station, lab, flipped, and individual (Staker & Horn, 2012). Of these structures, the flipped-classroom model has been seen most frequently in college BL courses (Jensen, Kummer, & Godoy, 2015) and will be focused on in this literature review. In a

flipped model, students rotate on a fixed schedule between face-to-face guided practice in a classroom and online content and instruction asynchronously from an off-campus, alternative location. Students spend their time in the classroom participating in collaborative activities with classmates, in a lab with online tools, and at an off-campus location learning course content (Jensen et al., 2015).

The concept of flipped learning first emerged in the 1990s. The basic idea involves learners studying new concepts on their own. The classroom, then, becomes a place where students can do homework under the supervision of an instructor, who can offer one-on-one help and guidance whenever they need it (Bailey, Ellis, Schneider, & Ark, 2013, p. 6). Researchers note that flipped learning is not a substitution for traditional learning (Bauer-Ramazani, et al., 2016; Blake, 2009; Vaughan, 2014), but instead, a goal of flipped learning is to allow teachers to become guides to learning instead of being the center of the learning process (Bauer-Ramazani et al., 2016; Webb et al., 2014; Han, 2015; Kvashnina & Martynko, 2016).

Research in flipped learning has identified four requirements involved in the development for instruction (Hamdan, McKnight, McKnight, & Arfstrom, 2016) including: flexible environment (Chenoweth, Ushida, Murday, & Ushtoa, 2006), learning culture, which enhances motivation in a classroom, (Bauer-Ramazani et al., 2016; Hsieh et al., 2017), authentic content (Bauer-Ramazani et al., 2016; Kvashnina & Martynko, 2016), and guidance from an informed perspective (Bauer-Ramazani et al., 2016; Coryell & Chlup, 2007). Authentic content, from a professional educator, can be derived from project-based learning (PBL) and multiple informal assessments (Han, 2015) where the

teachers are not simply following a set curriculum designed by the university. Educators can use authentic content to maximize classroom time in order to adopt methods of student-centered, active learning strategies. The flexible environment has the potential to allow for individualizing needs of students and providing students with frequent, personal feedback from the instructor (Bauer-Ramazani et al., 2016; Webb et al., 2014).

Through flipped learning, research has shown that teachers are able to assist students during class time without sacrificing direct instruction (Bergman & Sams, 2012; Kim, Byun, & Lee, 2012) and enabling students to take ownership of their own learning in order to be autonomous, lifelong learners (Hsieh et al., 2016; Kvashina & Martynko, 2016; Pinto-Llorente, Sánchez-Gómez, García-Peñalvo, & Casillas-Martín, 2017; Yalçınkaya, 2015). Studies show that with autonomous learners immersed in engaging activities, the classroom can become a setting for meaningful discussion and collaboration (Coryell & Chlup, 2007; Egbert, Herman, & Lee, 2015; Han, 2015). Researchers claim that flipped classrooms give ESL students more opportunities to engage in higher order activities that promote participation and language learning (Evseeva & Solozhenko, 2015; Lee & Wallace, 2017). Egbert, Herman, and Chang (2014) found that through the use of Moodle and flipped strategies, students had opportunities to be involved with authentic language learning materials in the form of audio, text, video, and graphics. Students were able to discuss topics outside of class via the online discussion board, and they were able to learn and practice at times they deemed appropriate rather than being restricted by in-class time restraints (p. 8). Similarly, Hsieh et al. (2017) found that flipped instruction, when using extensive

online interaction, motivated the participants to learn English idioms and to improve their oral ability and engaged participants in the learning tasks, making them more active and competent in using the learned idioms for communicative interaction in class discussions.

One of the most important implications for English language teaching is that the flipped model can encourage discussion and critical thinking which is key to increasing students' language production and collaboration (Chen, Wang, Kinshuk, & Chen, 2014; Davies, Dean, & Bell, 2013; Kostka & Brinks Lockwood, 2015). Mehring (2016) supports flipped learning in ESL courses due to the study's findings that it creates a communicative and student-centered learning environment. Further studies are necessary in order to develop an understanding of flipped instruction in an ESL setting in regard to social presence, specifically in a community college setting. The connection that the current study observed is the student-to-student and teacher-student relationship (Gaughan, 2014; Slomanson, 2014; Van Veen, 2013).

#### **Social Presence in Blended Learning Research**

Online interactions have been reported to significantly enhance learning outcomes in online learning contexts (Ravenna, Foster, & Bishop, 2012). Meaningful interactions can facilitate peer relationships between students in a blended synchronous and asynchronous learning course. Garrison et al. (2010) further suggested that social presence occurs when learners are able to identify with a community, communicate within that community, and develop relationships by projecting their personalities. Tu et al. (2012) agreed with their findings and observed that students felt enabled to make learning more personal by projecting positive social digital identities in a community of

learners. Weinel, Bannert, Zumbach, Malzahn, and Hoppe (2011) found that while social presence does not cause collaboration, it can affect the attitude of participants towards collaborating on a particular assignment. The study findings showed that ambiguity and uncertainty in online communication can be minimized through the use of chat protocols and norms established by the teacher. More research is needed to determine "how such rules can influence the perceived quality of the communication" (p. 519).

Other researchers have also found a connection between social presence and blended learning (Jusoff & Khodabandelou, 2009; So & Brush, 2008). Jusoff and Khodabandelou (2009) found, "social presence is an individual's ability to demonstrate his/her state of being in a virtual environment and traditional environment and so signal his/her availability for interpersonal transactions" (p. 81). The study suggests that the blended mode decreases the psychological distance and also increases interaction between instructor and students. Similarly, So & Brush (2008) found that a blended format lowers the psychological distance while still maintaining a positive perception from students of social interaction. The research also highlighted the importance for instructors to design courses with meaningful online opportunities to collaborate and socialize. There is a need for additional research that explains how to accomplish these goals in a BL classroom. To work collaboratively in the virtual environment of a blended course, ESL students need carefully structured and supported assignments that facilitate communicating with their peers.

Computer Assisted Language Learning (CALL) has emerged as a tool for online communication and collaboration in blended courses for second language learners. CALL

is a language learning and teaching approach in which a computer is used as a tool for presenting, assisting students, evaluating materials, and promoting interaction (Jafarian, Soori, & Kifipour, 2012). On the use of technology for language learning, Garrett (1991) stated that the power of technology as a medium for supporting new kinds of language learning activities is multiplied by its potential for an unprecedented integration of research and teaching. A CALL lesson has the possibility to create an environment for interesting language learning assignments that could then be fitted with a program collecting data on how the learner makes use of that environment. "The data can not only feedback into improving pedagogy but can also contribute to the development of second language acquisition theory" (Garrett, 1991, p. 94).

Studies of perceptions of social presence have focused on the satisfaction level of group work and class discussions. This dissertation, however, will look at social presence as defined in the introduction. Little research to date has focused on community college faculty members' perceptions of creating digital content to enhance their own social presence with their students in a BL course. Studies have reported online university faculty members' perceptions of video as a useful tool to enhance social presence (Borup, West, & Graham, 2012; Griffiths & Graham, 2009). Another study reporting online university instructors' perceptions of social presence focused on strategies other than creating digital content (Bruce, Young, & Kennedy, 2012). These studies represent a small population of strictly online teaching faculty members and an even smaller representation of community college faculty members.

The perspective from the teacher of how the students, and they themselves, become a part of the community is a question that remains unanswered by the research. More research also needs to be done on the implementation of collaborative assignments, CALL tools, and social presence in BL. The observations in this study of social constructs will allow language teachers in the community college setting to reflect on their classroom community, how they foster social presence, and what integrated assignments achieve the best results for social and language learning. A BL course has both online and face-to-face activities, so identifying how to promote social presence and language learning in both settings needs to be taken into consideration.

# Design Guidelines of Blended Learning for Language Learning

Successful BL courses for language learning require carefully designed instruction on the foundation of the learners' needs and abilities (Neumeier, 2005), as well as their potential. Parameters suggested by Neumeier (2005) provide an exemplary rubric for evaluating a BL course design and improving implementation. The framework is composed of 6 parameters: mode(s) (computer assisted and/or face-to-face), model of integration (lead mode guides the learning process and the structure of the course), distribution of learning content and objectives (the tasks pertaining to both modes are arranged and sequenced as well as the use of each of these tasks is determined), language teaching methods, involvement of learning subjects (both students and teachers), and location (virtual and traditional) of learning. CALL as a field still lacks qualitative research on blended learning and parameters such as these offer language teachers a guide in managing online learning components present in blended programs.

For mode, there are two major forms cited for BL by Neumeier, face-to-face interaction and the CALL system, usually on an LMS. Both modes need to be specific to the learning environment and intentional, which aligns with BL research that validates meaningful interactions and assignments (Chapelle, 2009). Once the balance of modes has been decided on for a course, the adoption of a model of integration plays a vital role in the BL environment as the order of modes and their level of integration into the learning process dictates the path of a course. The modes can be organized in a rotating cycle or can be overlapping and parallel. The degree of transactional distance, meaning "the space of potential misunderstanding between the inputs of the instructor and those of the learner" (Moore, 1993, p. 22) needs to be assessed during the sequencing. The primary goal is to minimize the level of transactional distance by ensuring that students do not feel isolated or uncertain (Moore, 1993).

Stracke (2007) reported on the fact that some ESL students choose to drop second language courses with a technology component. Stracke determined that one of the main reasons behind their decision had been the failure to see a connection between face-to-face instruction and CALL modes. Based on the results, Stracke suggested language teachers provide transparent connections between the two modes of instruction (face-to-face and online). In terms of the distribution of learning content and objectives for the language classroom, educators must ascertain if the target language skills will be practiced in both modes in parallel, meaning that assignments flow between online and in-person class sessions or they are isolated in one mode or the other. For language

teaching methods, learning techniques in each of the applied modes may vary due to differing manners of interactions in the classroom and virtual environments.

The parameter, involvement of learning subjects, refers to the diverse interactional styles: individual vs. collaborative language learning, teacher and learner roles, and level of autonomy of each mode (Mendieta Aguilar, 2012; Neumeier, 2005). For example, a student's role might change from a passive one, in which the student listens and take notes on a lecture and solves or struggles with homework on their own (Shimamato, 2012), to that of an active participant in a FL classroom when in-class activities focus on discussions, problem solving, and providing meaningful feedback (Baepler, Walker, & Driessen, 2014; Hung, 2015; Šafranj, 2013; Zappe, Leicht, Messner, Litzinger, & Lee, 2009). The teacher-to-student and peer-to-peer communication and interaction may help students to elaborate on the issues being studied and may facilitate the transition to active learner (Al-Zahrani, 2015). An active role can require a greater level of autonomy from the learner due to students being faced with a much greater scope and variety of roles than if their actions were only restricted to one mode of learning (Neumeier, 2005). Lastly, location means the educator designs learning spaces, both virtual and in the classroom, to assist students in achieving the goals of the course through the selected modes. These locations can extend to home, outdoors, computer lab, and other institutional settings. According to Neumeier, "locations where learning takes place successfully are social and individual artefacts that are an integral part of a learning culture" (p. 175).

Time and consideration are required to adapt to BL environments for both professors and students (Barr, Leakey, & Ranchoux, 2005; Kvashnina & Martynko, 2016). BL has been practice-led as opposed to research-based (Neumeier, 2005). Academic practice is often sustained by an instructor's understanding of BL rather than a consistent approach across an institution (Zapata, 2004). Neumeier's pedagogical parameters provide a useful framework for teachers to consider when incorporating BL in language learning courses. Nevertheless, the results of Neumeier's study point to the likelihood that time spent in online and traditional modes may not be equally valuable for all students. There is a need for a more ethnographic approach to the research to provide details of the learning environment, such as teacher beliefs about their own roles, relationships, and interactions. Ethnographic research has the potential to capture a complete picture of the BL environment, providing an in-depth correlation of the relationship between contexts and situations of language use (Geertz, 1973).

Wen (2008) proposed the Output-driven/Input-enabled model for second language acquisition and it has been adopted as a primary framework for Flipped Learning by researchers (Hsieh et al., 2016). Wen (2008) emphasizes that the need for output drives learners to pursue input, and input enables learners to produce output. According to Wen's model, teachers are responsible for: (1) designing authentic output tasks that meet and improve the learners' proficiency level; (2) providing appropriate task-based inputs to enhance learner intake; (3) providing appropriate output assistance to improve learner ability; and (4) offering targeted feedback rather than general suggestions (Wen, 2008). Numerous studies positively align with Wen's model (Bauer-Ramazani et al.,

2016; Ferreira, Salinas, & Morales, 2014; Han, 2015; Hsieh et al., 2016; Šafranj, 2013) and indicate that if those parameters are not in place in the online environment, the learner has less of an opportunity to be successful (Compton, 2009).

Practitioners cannot assume that a natural transition in teaching skills from a face-to-face classroom into an online environment will occur. Faculty teaching blended learning courses must adopt new tools and new mindsets to increase the likelihood of positive outcomes (Leonard & DeLacey, 2002). Language teachers need to account for new learning systems, especially considering that the effort and cost of creating online materials "can be wasted without the adequate training of teachers to present and support the learning" (Hampel & Stickler, 2005, p. 312). Ehlers and Schneckenberg (2010) found that practitioners are concerned with their changing roles from information transmitters to facilitators who reinforce social interaction. Hoic-Bozic, Mornar, and Boticki (2009) emphasize that instructor roles need to be redefined from instructor-centered to learner-centered. More investigations are needed to gain insight on the attitudes, perceptions, and behaviors of faculty when constructing and implementing assignments in blended learning courses.

To aid instructors in the design process, Neumeier concentrates on the design of a course broadly. Wen brings personal feedback and individualized instruction into consideration which plays a significant role in BL success (Hsieh et al., 2016). Neumeier (2005) and Wen (2008) both provide frameworks for implementation and parameters of BL in the classroom. However, both researchers lack a comprehensive guideline for how educators can adopt BL and sustain the positive effects through different courses with

varying learner styles. Both researchers also lack specific recommendations for developing social presence with students from diverse cultural backgrounds in an ESL classroom.

The CoI framework, originally proposed by Garrison et al. (2000) can guide instructors to bring in the social presence piece in their practice in blended education. The framework is based on the "collaborative constructivist" (Garrison et al., 2001, p. 92) perspective of teaching and learning associated with the work of John Dewey (1938) and the foundations of interaction and continuity in the learning environment. Social presence has evolved since the work of Dewey (1938) and Garrison et al. (2001) and is now seen as a factor for learner engagement, knowledge construction, and peer relationships (Armellini & De Stefani, 2016). Due to the influence of social presence on teaching and cognitive presence (Kanuka, Liam and Laflamme, 2007), the other two constructs in the CoI framework (Garrison et al., 2000), this study focused solely on social presence in the blended learning environment.

## **Blended Learning in ESL Higher Education**

Language learning takes place when learners interact and construct social practices (Block, 2003). BL course design, steeped in a social framework, has potential to encourage learner participation and meaningful interaction between both instructors and their students and students and their peers. A BL college course design must be intentional to create a student-centered, authentic learning community (Tu et al., 2012). A BL approach also allows for communication necessary to help build relationships to promote a rich language learning environment that can be both synchronous and

asynchronous (Webb et al., 2014). During planning for any ESL college course, considering the educational level of the language learners is a top priority. However, with a BL language course, ESL educators also need to understand the students' familiarity with the instructional materials and technological delivery systems when designing a course (Hampel & Stickler, 2005). Once a student's competency level is established, educators are responsible for giving students guidelines on how to use tools specifically for the purpose of acquiring greater language proficiency (Garrett, 2009).

Contextualization, authenticity, and sustainability enrich student content knowledge which promotes linguistic comprehension (Han, 2015; Pinto-Llorente et al., 2017). Many academic settings fail to foster authenticity which can impact pragmatic knowledge (Nikula, 2002). In language classrooms that utilize communicative language teaching (CLT), an approach to language teaching that allows for interaction and engagement with authentic materials, student-centered pedagogy is already in place, but paralleling that with BL instruction allows the teacher to be even more accessible with the added mode of online availability (Webb et al., 2014). Higher education professionals using BL have been able to deliver instruction, including new innovations in technology, which broadened communication and meaningful interaction (Bauer-Ramazani et al., 2016; Webb et al., 2014; Han, 2015). Thorne (2003, p. 18) also points out that "blended learning represents a real opportunity to create learning experiences that can provide the right learning at the right time and in the right place for each and every individual, not just at work, but in schools, universities and even at home."

## **Benefits of Blended Learning in ESL Higher Education**

The use of a BL model in university language courses has been found to be successful, despite numerous obstacles, in creating an authentic context to support language learning. Obstacles can include the high number of students per class (Bueno-Alastuey, 2009) and the different levels of proficiency among students (Ruiz, Palmer, & Fortanet, 2005). Ananiadou and Claro (2009) suggest that collaboration is one of the most important capabilities that learners can develop for a globally connected world. BL affords ESL students the opportunity to collaborate with meaningful input and interactions (Gillies, 2006; Krashen, 2014). These collaborative interactions can take place outside of the face-to-face classroom to allow more time for genuine language use (Webb et al., 2014; Pennington, 2004).

Combined with technology, BL instruction can introduce an active learning environment with flexibility in using resources for the students and provide more time for faculty members to spend with learners in small groups or even individually (Pinto-Llorente et., 2017; Šafranj, 2013; Yalçınkaya, 2015). Chen and Jones (2007) outlined other advantages of blended learning such as contextualization and understanding of topics by using web-based resources as well as active participation of students in class. Furthermore, online learning engagement provides an interactive setting for communication among teachers and students in the classroom and may facilitate cooperative activities even beyond the classroom (Yuen, 2011).

Regarding using digital communication tools in activities, Dzakiria, Mustafa and Abu Bakar (2006) found that the interaction between students and lecturers as well as

scholarly discussion, both in synchronous video conferencing or asynchronous video-taped communication, is preferred in a blended learning application rather than F2F. Many studies have shown that BL increases student achievement compared to solely online and traditional courses (Bleffert-Schmidt, 2011; Webb et al., 2014; Ferreira et al., 2014; Kvashnina & Martynko, 2016; Šafranj, 2013; Scida & Saury, 2006). In these experimental studies, the investigator actively manipulates which groups receive the agent or exposure under study (Aschengrau & Seage, 2013, p. 137).

Blended learning can also bridge the gap between the traditional sense of community and the loss of it in an online course. Rovai and Jordan (2004) applied a causal-comparative design to investigate the relationship of sense of community in fully online, traditional classrooms, and blended higher education learning environments. They found that blended courses create a stronger sense of community among learners than either traditional or fully online courses. BL provides more productive engagement and active participation among students in the online environment and in course content as well which can lead to improved academic performance (Hsieh et al., 2016; Kvashinina & Martynko, 2016; Tayebinik, & Puteh, 2012).

Finally, BL can increase the amount of input and practice that language learners receive inside and outside of the classroom (Ferreira et al., 2014). In Han's study (2015), the goal of the research was not only fluency development in speaking, but also, because of limited time, autonomous language learner training. The aim was to "train students to know how to learn English effectively and independently" (p. 99). Autonomy, a popular theme in BL (Kvashnina & Martynko, 2016; Pinto-Llorente et al., 2017; Šafranj, 2013;

Yalçınkaya, 2015), is critical for English learners to actively engage in their learning in order to extend and sustain their knowledge. Bender stated (2003, p.6) "research shows that through BL, students become far more active in their own learning, feeling more technologically empowered and able to learn anywhere and anytime in the manner that best suits their lifestyle." Learner autonomy in the ESL setting promotes self-sufficient learners and citizens capable of evaluating problems and inconsistencies in their education and in real world contexts. The current study focused on teacher perceptions of social presence in the BL environment, their impact on student achievement, and the advantageous tasks that drive successful language learning.

# **Challenges of Blended Learning in ESL Higher Education**

Difficulties in language teaching and learning have also been documented in the literature. Blended learning in ESL courses requires different teaching approaches than those in face-to-face classrooms and those in other blended learning subject areas (Compton, 2009). Furthermore, language teachers are burdened with the same challenges as traditional higher education teachers. ESL teachers face limited class time (Han, 2015). Limited practice time hinders the development of linguistic skills (Ferreira et al., 2014). BL can supply ESL teachers with more time inside and outside of the classroom environment for remediation and differentiation (Bauer-Ramazani et al., 2016; Webb et al., 2014), but BL also requires an additional time commitment for educators. Finding the time for training, both for the educator and the student, on technology tools was an obstacle found in numerous studies (Al-Fudail & Mellar, 2008; Coryell & Chlup, 2007; Han, 2015). In their 2015 study, Mirriahi, Alonzo, and Fox found technology competence

among faculty remains low and technology is used as a management system more than a way to promote student-centered learning.

Researchers found that educators had to find extra time for professional and course development (Hampel & Stickler, 2005; Hubbard, 2008; Mendieta Aguilar, 2012), some of which were lacking in useful outcomes for implementation (Hubbard, 2008). Providing individualized feedback and designing rigorous tasks, while extremely valuable, was noted to be daunting and time consuming, both in online and face-to-face settings (Bauer-Ramazani et al., 2016; Webb et al., 2014; Grgurović, 2014; Kvashnina & Martynko, 2016). Another challenge in the technology implementation was getting the students on board with BL synchronous and asynchronous tools (Webb et al., 2014). Benson, Anderson, and Ooms (2011) found that teachers perceived of the work required to develop BL materials as time-consuming and that technology resources were prone to failure.

Challenges to learning can be met and overcome using a variety of methods, but challenges posed by a learning interaction between teacher and student or student and student are difficult to vanquish if adjustments are made by one person alone (Draffan, Rainger, & Ltd, 2006). Meeting the needs of a learner's skills, abilities, and preferences in order to facilitate a particular learning interaction requires the cooperation and involvement of both the educational faculty (represented by the teacher perspective) and of the student (represented by the learner perspective) (Draffan, et al., 2006). Social presence research indicates that establishing environments where instructors and students can project themselves effectively is imperative to their accreditation of a learning setting

(Garrison et al., 2000; Whiteman, 2002). In contrast, when the environment is characterized by low social presence, participants perceive it as impersonal and the amount of information shared decreases (Leh, 2001).

## **Blended Learning in ESL Community College Higher Education**

Community colleges provide open access and affordable options for higher education to a growing population of adult learners in the United States. According to 2015 enrollment data from the American Association of Community Colleges, about 19% of community college students at the national level are Hispanic, 13% are African American, and 13% are Asian (Community college trends and statistics, 2015). About 40% of Hispanic students enrolling in two-year colleges have needed English remediation in Virginia, and 41% nationally (Complete College America, 2016). As Hispanic student enrollment increases, colleges should be intentional with their plans to serve these students.

In light of the changing demographics, community colleges have been looking for innovative ways to meet the academic needs of their non-traditional student body on a diminishing budget. Community colleges tend to have small budgets because they are generally under more pressure than universities to keep tuition and fees low which affects the amount and types of advanced technology that they can afford to purchase (Crookston & Hooks, 2012; Roe, 2009). Community colleges in many parts of the United States cater to lower-income populations who either cannot afford university tuition or are unable to gain admission because of their educational background (McKinney & Backscheider-Burridge, 2015; Roe, 2009). Furthermore, many ESL students at the community college

level tend to be working adults who are struggling to balance work, academic expectations, and familial obligations (Crawford et al., 2014; Roe, 2009). Low socio-economic levels may also mean that these students have little or no experience using, or access to, technology until their first community college class (Crawford et al., 2014; Mate-Martinsen, 2009; McGee & Reis, 2012). On the other hand, there may also be ESL students coming from public education who are knowledgeable in technology due to access in K-12 schools. ESL learners at community colleges are a diverse group. Research on blended learning is minimal and in ESL community college education, the body of research is especially scarce.

A majority of studies that address online learning in higher education focus on well-prepared university students and ignore questions of effectiveness of online learning in community colleges (Xu & Jaggars, 2011). Researchers suggest that the lack of success among students in online courses may be associated with barriers including technical difficulties, a sense of isolation, a relative lack of structure, and a general lack of support (Jaggars, 2011). Additionally, collaborative learning and social presence are not widely researched at the community college level. The findings suggest that more research needs to be done in the community college setting to promote active learning, collaboration, and social presence in the ESL education setting. However, educators must understand that learning is more than gaining knowledge about certain content and that meaningful learning occurs when students effectively collaborate, communicate, and engage in the process (Tu et al., 2012). Therefore, continued research is essential to explore how educators who use blended learning perceive the ways in which it influences

their teaching practices and assists students in the language learning process while establishing a social presence.

## **Teacher Perceptions of Language Learning and Blended Learning**

Riel, Lawless, and Brown (2016) acknowledge that in order for BL to be successfully implemented, it is valuable to understand teachers' perspectives. The authors note that teachers are the ultimate representatives and facilitators of curricula and adhering to their perspective during implementation is paramount for a blended classroom. Comas-Quinn (2011) explored teachers' experiences using blended learning. This mixed-methods study involved both participant observations and a survey followed by interviews. Comas-Quinn identified three emerging themes impacting teachers' abilities to effectively integrate technologies into the curriculum: technical issues, the lack of online tools to integrate course activities or assessments, and shortage of time. These three factors, supported by earlier studies, are among the top reasons for negative teacher perceptions about BL in the classroom (Barr et al., 2005; Lin, Wang, & Lin, 2012). However, the studies in the literature focus predominantly on experiences and strategies for overcoming challenges instead of the direct perceptions and attitudes themselves. Lin et al. (2012) reported that teachers' technology integration was affected by many factors including Instructional Computer Technology (ICT) equipment, support, curriculum, culture, teaching load, leadership, and most importantly, teacher buy-in (Lin et al., 2012). They suggested further research to corroborate the evidence and to understand teachers' personal attributes of technology integration.

In one of the few studies detailing teachers' specific perceptions and concerns of blended learning, Wanner and Palmer (2015) interviewed 47 educators about flexible learning. Teachers overall believed that flexible learning and flipped classrooms specifically required greater amounts of work. Approximately half of all teachers indicated that they had a low level of commitment to the flipped classroom but felt under high levels of pressure to include a flipped component in their courses. Many were engaged in flipping their classes while some teachers had used a flipped classroom "by default." Other participants were "experimenting" with it as they were currently providing a blended learning environment through some online content. Many were open to flipping their classrooms because, as one teacher put it, "our current model of teaching is not sustainable" in a context of budget constraints and more time demands for teachers (p. 361). A concern shared by all teachers was the time commitment and workload required to set up, implement, and manage a flipped classroom. Analysis showed that the time required to implement a flipped classroom was the most prevalent concern and the reasoning behind faculty not engaging in BL instruction, which aligns with the aforementioned studies. In addition, higher education teachers, with the exceptions of innovators who are experimenting with new ideas, are reluctant to change traditional teaching methods (Partridge, Ponting, & McCay, 2011).

Cuban (1986) emphasizes the importance of the educators' role on the success or failure of BL as a pedagogical tool. He describes teachers as "having the password that will unlock the classroom door," and advises community college administration to "understand what questions teachers ask and what criteria they apply as essential to

unlocking the door" (p. 71). Lacking in this body of literature thus far is the faculty's perceptions of the role of social presence within an ESL blended learning environment and techniques they may have adapted to compensate for distance. Without an understanding of their perceptions, it is difficult to create and sustain an effective BL curriculum to motivate students and encourage their own positive perceptions of communicative language learning.

Factors that drive student motivation in BL courses have been researched for a number of years. Educators implementing BL courses cite student motivation as the most important factor in educational success in general (Dörnyei, 2001). Gardner (1985) hypothesized that ESL learners with positive attitudes toward the target culture and people will learn the target language more effectively than those who do not have such positive attitudes. To extend his work, Gardner, along with MacIntyre, (1993) drew together findings from many studies over several decades and developed Gardner's socio-educational model of Second Language Acquisition (SLA). The researchers claim that individual-difference variables (e.g., cognitive variables and affective variables), influenced by antecedent factors (e.g., biological factors such as age and experiential factors such as previous language training experience), interact with both formal and informal language acquisition contexts and influence both linguistic and nonlinguistic outcomes (e.g., students' reactions to the learning experience). Gardner and MacIntyre (1993, p. 9) argue that this model shows the importance of what takes place in the learning contexts: "Teachers, instructional aids, curricula, and the like clearly have an effect on what is learned and how students react to the experience." The model also

predicts that students' degree of success (linguistic outcomes) affects their feelings (nonlinguistic outcomes) and that both types of outcomes will have an influence on individual-difference variables including language attitudes and motivation.

Gardner's model coincides with student motivation research that yielded positive attitudes and academic success in BL courses (Han, 2015; Hsieh et al., 2016; Kvashnina & Martynko; Šafranj, 2013). In Hsieh et al.'s (2016) research student perceptions were evaluated in four dimensions: motivation, time commitment required, nervousness in using English, and outcomes. In the study, teachers eased student anxiety by pointing out that the more time and energy students put into their workload, the more successful the students became with their learning. The results also indicated that educators should show students the formative benefits and beneficial outcomes of putting extra effort into a BL course and this should be done early on in the course.

# **Teacher Perceptions of Social Presence**

Gunawardena and Zittle (1997) examined social presence and determined it was "a predictor of satisfaction within a computer-mediated conferencing environment." They continued, "Instructors who are accustomed to relying on nonverbal cues to provide feedback and who have a lesser-developed ability to project their personality will need to learn to adapt to the online medium by developing skills that create a sense of social presence" (p. 23). Social presence research involving practitioners suggests that there are many possible roles and associated behaviors or actions that define online teaching, and these ultimately have an effect on student perceptions and learning. The gap in the research is that it lacks insight into the decision-making processes that instructors engage

in and the reasons for such decisions. Defined as "decisions made during teaching," Tsang (2004) observed the kinds of interactions three ESL teachers made in their teaching of a lesson and the basis for these interactive decisions as it related to their personal practical knowledge. Although Tsang's study focused on traditional classroom teaching, it illustrated how the development of various contextual constraints such as lack of time, equipment breakdowns, and misinterpretations impedes the ability of the instructors to make decisions that allow for building a sense of community.

A greater investment is needed to understand characteristics that may distinguish faculty perspectives about blended learning and social presence, and how they influence student outcomes. Vesely, Bloom, and Sherlock (2007) found that students prioritize instructor modeling, in which an instructor conducts an exhibition of proper skill performance, over interaction and dialogue. They also found that the reverse is true for instructors. A more current view of instructor perspectives in the community college setting is imperative to bridge the gap in the classroom community. The current study will draw on teacher pre- and post-course interviews, document analysis from the LMS, and in-class materials, and classroom observations. DeNoyelles, Zydney, and Chen (2014) suggested two strategies that instructors can use to promote social presence. First, instructors should model good social presence in asynchronous online discussions (DeNoyelles et al., 2014). Second, instructors should require that students participate in the discussions (DeNoyelles et al., 2014). The current research will examine how the instructor implements activities to reinforce those strategies.

The CoI framework construct of social presence can provide practitioners with a model for online community development. Combining classroom and online activities is only a small step in blended learning. Instructors and instructional designers should invest more time and effort on the analysis of learners. Therefore, this study aimed to examine in-depth one instructor's activities and the impact on students' language learning and social presence.

## Relationship of the Literature to the Study

Blended learning course offerings have evolved from what began as e-learning courses. Through these offerings, higher education is accessible for students who cannot attend traditional courses. BL courses are especially appealing to community college students who do not reside on campus, are of varying ages, or have work and family responsibilities. While demand for convenient courses is increasing, the BL model has both advantages and disadvantages, as discussed in this chapter. As a result of the prospective advantages to language learning, collaboration, learner autonomy, and student satisfaction are among the major components that have been studied in regard to blended learning.

In this chapter, social presence was discussed through the lens of the CoI theory. The CoI framework provides a comprehensive theoretical model to inform both research on BL and the practice of flipped instruction. The framework suggests that student satisfaction with online learning requires the development of a community that supports meaningful inquiry and deep learning (Anderson, Rourke, Garrison, & Archer, 2001). Since strong development of social presence provides a platform for sharing ideas

and supporting learning, educators interested in the effectiveness of online instruction must engage in the development of social presence by fostering community (Palloff & Pratt, 2007). For language learners, this is particularly imperative because relationship-building facilitates authentic writing and communication in English (Vanek, King, & Bigelow, 2018).

Several researchers have studied presence in distance and hybrid courses using specific tools to facilitate authentic communication. Limited research exists, though, on community college teacher perceptions of ESL blended courses in relationship to social presence and language learning. This study aimed to fill a gap in the research by examining a teacher's perceptions of student language learning with the use of blended learning tools and the cultivation of social presence in an ESL course at a Virginia community college campus. Chapter three will detail the methodology, data collection, and analysis for this study.

#### **CHAPTER THREE: METHODOLOGY**

### Introduction

This chapter describes the case study methodology undertaken in this dissertation. It provides the rationale for the descriptive, single-case design. This dissertation employed qualitative data sources that were obtained from an instructor of an ESL blended learning course at a community college campus at Northern Virginia Community College (NVCC). The primary goal of this study was to explore teacher perceptions of effective instruction to foster social presence and promote language learning in an ESL BL course and how those perceptions may affect learning outcomes. The chapter is organized into four sections: (a) research method and design; (b) population and setting; (c) procedures and data collection; and (d) data analysis.

### **Research Method and Design**

Faculty play a significant role in the success of a university's BL implementation efforts (Christo-Baker, 2004; Graham & Robison, 2007). Despite their involvement, "little has been published regarding faculty application of hybrid teaching" (Porter, Graham, Bodily, & Sandberg, 2016, p.17). As addressed in chapters 1 and 2, there is limited research on how community college ESL instructors can blend F2F and online practices successfully in a classroom in order to promote language learning (Ingerson, 2011). The purpose of this qualitative case study was to explore a teacher's perception of

a blended learning design that promoted rich social, linguistic, cultural, and academic potential in ESL learners when the instructor adapts curriculum and assignments to encourage those capabilities.

The qualitative case study occurred in the Northern Virginia Community College (NVCC) system at a campus in Northern Virginia. I conducted semi-structured interviews with the instructor, utilized documentary analysis, and observed class sessions. To answer the questions posed in the introduction, a phenomenological case study was used to guide this research. This methodology was employed to give voice to the ESL instructor's experiential claims and concerns in the context of their shared environment with students in the class. With this method, I sought to understand the perceptions of a teacher on how they design and implement course assignments to foster social presence and a sense of belonging to the community within the context of a blended learning English language community college course. Creswell (2007) deems this inquiry best for "capturing the detailed stories or life experiences of a single life or the lives of a small number of individuals" (p. 55). A qualitative case study was an appropriate methodology for this research in that it can reveal relevant data about human experiences, expectations, contextual meanings, and causal explanations (Denzin & Lincoln, 2005; Yin, 2013). Each component of this chapter has been geared to following a qualitative research methodology especially designed for a case study in order to explore a teacher's perceptions of language learning and social presence in the blended learning ESL classroom.

Qualitative research was utilized to provide an in-depth understanding of a phenomenon from the viewpoint of the participant (Bloomberg & Volpe, 2015). Forming a connection with the participant led to my responsibility to interpret the participant's understanding and knowledge of the phenomenon (Creswell, 2014). The research was exploratory; the context and experiences of the participant increased the practical relevance of the findings (Tetnowski, 2015). Case study research has been used in various areas of applied linguistics for the last 30 years (Duff, 2010) to study a wide range of phenomena ranging from bilingualism, language fossilization, and loss to the effects of individual learner differences, identity, and gender on language learning. Although they are limited, exploratory qualitative case studies can also be found focused on BL in the community college setting (Futch, deNoyelles, Howard, & Thompson, 2016; Hill, 2016; Zielinski, 2017).

While some do employ constructivist methods (Zielinski, 2017), most are multicase or comparison and none of the current studies involve language learning. The participant in this study shared their perception of social presence and language learning pedagogy in blended learning through recorded interviews. Since the study was geared toward the community college ESL blended learning environment, the I selected a campus with courses designed to be a hybrid of F2F and online for the ESL population. A one-hour pre-interview was conducted with the instructor to establish how assignments were designed in the course to foster social presence and promote language learning. The interview questions also highlighted the planning and implementation process of course materials. At the end of the semester, the instructor participated in a one-hour post-

interview to express perceptions of how assignments related to learning outcomes and if the assignments contributed to students forming a sense of community. The interview questions were formed with the guidance of the CoI framework (Garrison et al., 2000) and the CoI survey (Arbaugh et al., 2008).

CoI framework was an appropriate theoretical lens for this study because it provided a procedure for creating and maintaining the socially constructed nature of traditional classroom learning in virtual environments. Researchers using the framework, and the framers themselves, believe a worthwhile educational experience is based on a process of reflective inquiry (Bangert, 2008; Garrison, 2011; Garrison et al., 2000; Ice, Gibson, Boston, & Becher, 2011). In their CoI survey instrument, adapted from the CoI framework, Arbaugh et al. (2008) developed 34 items to determine a strong educational experience based on three overlapping elements. From a methodological perspective, the three categories of social presence (open communication, group cohesion and personal/affective projection) are used to operationalize the concept (Arbaugh et al., 2008). Based on the importance of social presence in the framework, the current study concentrated on that construct with its three subsets from the model and survey to create open-ended questions for a semi-structured interview in order to delve into the participant's experience as an ESL community college BL instructor.

This single case study explored social presence in an adult ESL blended learning course at a community college. The study met the criteria of a phenomenon and exploratory case because the researcher investigated the phenomenon of a teacher's perception of the effectiveness of blended learning and social presence on language

achievement in an ESL community college course (Yin, 2013). This case study examined the blended learning model of the course and perceived attitudes towards the effectiveness of the assignments by the teacher. The comparative methodology in the past literature did not allow for a sufficient description of the detailed experiences in an environment of a blended learning class and in its context. Moreover, numerous authors of blended learning comparison studies (BL and traditional) in higher education have uncovered mixed results or insufficient evidence for either learning style (Amaral & Shank, 2010; Bowen, Nygren, Lack, & Chingos, 2013; Jones & Lau, 2010; Tsai, 2010) and therefore lack a rich description of a BL learning environment that produces an impact on learners.

The review of the preceding literature indicated the lack of information about technology tools, course design, collaboration, and participant roles, which are all important features for successful design and implementation of blended models (Neumeier, 2005). Without knowing these elements, researchers have difficulty accounting for why learners performed the way they did or if a strong social presence in the course was formed. Studies have tracked the impact of online and hybrid courses on student achievement and retention as colleges and universities expand their online course catalogs (Palmer et al., 2014; Shea, & Bidjerano, 2014; Xu & Jaggars, 2011; 2013). However, rather than solely focusing on how students achieve success, the current study aimed to focus efforts on identifying practices that foster social presence and exploring the teacher's perception of how students achieve language success in a collaborative, blended learning environment.

#### **Population and Setting**

NVCC has five campuses in Northern Virginia. The population was selected because the number of students enrolled in ESL courses has increased in the specific campus of the research site since 2013. The college was among the top five in a ranking of digital community colleges by the Center for Digital Education (2015). The ranking recognizes technology innovations at community colleges. Therefore, this population was appropriate for the study design. I selected an educator from the research site who demonstrated extensive knowledge and experience in both ESL and blended learning. The selection of the instructor was critical to this research in providing valuable data and insight into future development of courses and activities needed to address teachers' pedagogical practices for a blended learning ESL environment. Once the instructor was selected based on expertise in ESL education while teaching in a blended learning environment, a reading course was chosen due to the fact that the understanding of texts requires cultural and social value knowledge (Rance-Roney, 2010). ESL courses at this institution range in levels and the courses included in the requirements are reading and vocabulary, written communication, oral communication, and reading. Researchers have defined reading comprehension as "a basic piece of the diverse exchange of systems engaged with L2 reading" (Brantmeier, 2003, p. 52).

Creswell (2007) stated, "we intentionally sample a group of people that can best inform the researcher about the research problems under examination" (p. 118).

Therefore, critical case sampling, a type of purposive sampling technique, particularly useful in exploratory, qualitative research (Patton, 1990) was used. ESL community

college students exhibit a range of demographic traits, personal attributes, and cultural awareness. The aim was to view the phenomenon of blended learning through the ESL instructor's perspective and observe their students in the blended learning setting that they had created. As an interpretive form of research, case studies explore the details and meanings of an experience. The richness of case studies is related to the amount of detail and contextualization that is possible when only one or a small number of cases are analyzed (Yin, 2003).

## **Region Characteristics**

Virginia has 11 Metropolitan Statistical Areas and of these, Northern Virginia is among the three most populous. Northern Virginia is most frequently defined to include the counties of Arlington, Fairfax, Loudoun, and Prince William and the independent cities of Alexandria, Falls Church, Fairfax, Manassas, and Manassas Park (Northern Virginia Regional Commission, n.d.). It is the most populated region in Virginia and the Washington metropolitan area with more than 2 million residents (Northern Virginia Regional Commission, n.d.). The most populated counties, both in Northern Virginia, are Fairfax and Prince William, with respective populations of 1,150,795 and 468,011 (U.S. Census Bureau, 2018).

There are more than one million immigrants in the state of Virginia (U.S. Census Bureau, American Community Survey, 2017). People who were born in El Salvador make up 11% of the population in Virginia, with those born in India (9%) and Korea (6%) comprising the top three immigrant groups throughout the state (U.S. Census Bureau, American Community Survey, 2017). Due to the proximity to Washington, D.C.,

Northern Virginia attracts a wide range of cultures and backgrounds from all over the word and most immigrants who come to Virginia reside in Northern Virginia. Over two-thirds of the state's immigrant population lives in this region, making up almost 30% of the total population of Northern Virginia (U.S. Census Bureau, American Community Survey, 2017). Immigrants in the state of Virginia who are over the age of five and speak English fluently, represent 61% of the immigrant population in this area (U.S. Census Bureau, American Community Survey, 2017). In the county of the case study, immigrants comprise more than 20% of the population and among this group, more than 30% speak languages other than English at home.

### **NVCC Characteristics**

Founded in 1964, Northern Virginia Community College (NVCC) is a public, 2-year community college that advertises a convenient, affordable, quality education. According to the NVCC website, it is the largest educational institution in Virginia and the second largest community college in the United States with more than 75,000 students and 2,600 faculty and staff members. NVCC is also one of the most internationally diverse 2-year community colleges in the United States, with a student body consisting of individuals from more than 180 countries. NVCC offers more than 160 degrees at the associate's level, including certificate programs. The six main campuses of the NVCC system are located near Washington, D.C., and include: Alexandria, Annandale, Loudoun, Manassas, Springfield, and Woodbridge. In addition, NVCC has major locations in the northern Virginia area that offer distance learning through NOVA Online and Workforce Development courses. Of the 75,000 students

enrolled in 2018-2019, 18,000 students registered for noncredit courses throughout the campuses and online.

Students coming to the NVCC system can be first-time, returning, and transfer students. The international population represents 20% of the entire student body. The number of students enrolled in ESL courses has fluctuated among the campuses somewhat between fall 2014 and fall 2016 with an increase in Loudoun and Manassas, stable enrollment in Annandale, and declining enrollment in Alexandria and Woodbridge. According to the NVCC Office of Institutional Research Factbook (2012-2017), the college ESL student body, on average, is 57% female and 43% male. Data is consistent with the student gender trends in Alexandria, Annandale, and Loudoun, with Manassas and Woodbridge showing a somewhat larger female population (approximately 62%) and smaller male population (about 38%). In terms of linguistic backgrounds, the largest linguistic group is represented by speakers of Farsi (14.52%), followed by Spanish (14.31%), Arabic (11.04%), Amharic (8.18%), and Vietnamese (7.87%). Interestingly, in the 2007 ESL Discipline Review report, the four largest linguistic groups were comprised of speakers of Vietnamese (almost 23%), Russian (over 17%), Japanese (almost 13%), and Spanish (over 10%). These differences represent demographic changes and recent immigration trends.

The OIR Factbook also reported that the ESL student population differs significantly from the non-ESL student cohorts at NOVA, and ESL students come with their own unique sets of strengths, challenges, and needs. One challenge is the commitments that ESL students have outside of their college life. Additionally, the OIR

students are employed, with 46% working full-time (35 hours per week or more), and 18.35% working 20 to 34 hours a week. In addition, 43% of ESL students are married, with 28% also raising children. At the same time, 21% of students live in the United States by themselves, without their families, and 5% live with extended family. All of these factors affect the time that the students can devote to their studies, in addition to affecting them psychologically (OIR ESL Enrollment Profile, 2014-2016).

These factors need to be taken into consideration in relationship with student recruitment and retention. In regard to prior education, 55% of ESL students graduated from high school in their native country and 20% earned a bachelor's degree in their native country. The reasons for attending an NVCC campus varied, however 43% listed the goal of transferring to a 4-year institution and 20% aimed to acquire a better job or a promotion. It is important to note that only 6% of the ESL population from all campuses combined responded to the survey (OIR ESL Enrollment Profile, 2014-2016).

## **Campus and ESL Program Characteristics**

The campus site of this study was one of the first permanent campuses built for NVCC. The population in the city where the campus is located is about 40,000. The campus is smaller in size and maintains smaller than average classroom sizes compared to more populated campuses in Northern Virginia. The ESL program at the campus offers classes to international students, ESL students from the surrounding area, and students attending college on student visas from around the world.

The college ESL program offers four levels of instruction from Level 2 (lowintermediate) to level five (advanced). At all four levels, students take courses in reading and writing. At levels two and three, they also take a course in oral communication. Students in levels four and five can begin taking introductory general education courses for credit along with their ESL reading and writing courses. Prior to that, all courses in the ESL program are non-credit courses. Students are placed into an instructional level prior to enrolling at the college. In order to qualify to enroll, a student must score at least 225 on the placement test. If a student scores 225 or higher, they can enroll in ESL courses, or may be eligible to begin taking credit-bearing courses in English. If a student scores below 225, the campus also offers courses in the American Culture and Language Institute (ACLI) ESL Program, including a focus on the work environment and areas of cross-cultural communication, multiculturalism, and diversity awareness. The ESL program at the campus currently has classes in reading and vocabulary, reading, oral and written communication, and composition. This study focused on a Level 4 reading course. For a student who started the program in Level 2, this would be the third level of classes. Other students might begin at Level 3 or test directly into Level 4. The campus ESL program currently serves more than 300 students.

## **Participant Profile**

In order to provide a context for understanding the results and to develop a strong narrative about these findings, a brief description on the teacher participant will be presented. A pseudonym has been used to maintain the anonymity of the participant. The instructor who consented to be part of the study teaches hybrid and F2F classes at the

community college that was the research site for this study. This research will refer to the participant as Linda.

## **Background**

Linda has been teaching ESL for over 25 years in the United States. She started at NVCC as an adjunct professor and acquired a full-time position seven years later. NVCC does not offer tenure to instructors. The school offers a one-year contract first, then multi-year contracts after that time period. Instructors can be on a 3-year or 5-year contract. Linda remarked that she is unsure of how instructors progress through the contracts. She is currently on a 5-year contract. At the same time, she has been teaching English at a four-year institution in Virginia. In the past, she has also taught at a large state research university and in programs for refugees. The ESL program at her campus introduced hybrid courses twelve years ago and she has been a part of the initiative from the beginning. Linda has been teaching the level 4 reading hybrid course for ten years.

Linda lives in the Northern Virginia area about 30 minutes from the campus. She enjoys her schedule because a majority of her in-person classes are in the morning which she prefers to evening classes. She also appreciates the flexibility of working remotely and having her work centralized in one device. Linda has a positive attitude about the hybrid course. In general, she enjoys using the computer and she likes the challenge of finding ways to interact with students online.

Linda is dedicated to helping students develop not only a deep understanding of the objectives in the course, but also a comprehension of hybrid learning in general. A large component of that understanding is that student-teacher communication is critical, especially outside of the face-to-face time. She continually works to achieve a level of comfort amongst her students so that they will reach out to her beyond the classroom walls. With laughter she states, "I had several students say that I would be a good grandma. I want my classroom to feel welcome and comfortable." In the four classroom observations, Linda's presence appeared to be that of a calm and gentle facilitator who provided clear and detailed language instruction to the students.

#### **Procedures and Data Collection**

This study followed the extensive review of literature on BL by Bluic (2007) and incorporated the belief that the current research should be holistic in nature. Sharpe, Benfield, Roberts, and Francis (2006) state that holistic studies on blended learning "shed light on the complex interplay of the virtual and the physical and the choices that learners make in finding pathways to successful outcomes" (p. 54). In order to take on this holistic approach, I met with the ESL department chair of the research site to gain insight into the blended learning courses offered and the ESL demographics of the college as a whole. After receiving Internal Review Board (IRB) approval from both participating colleges, interview questions and a consent letter, with an introductory email, were sent to the professor of a hybrid ESL reading course being offered in the spring of 2019 at the research site.

The instruments used for this case study were semi-structured interviews, document analysis, and observations. Interview questions provided information on the instructor's (Appendix A) perception of the blended learning experience as outlined by the CoI survey instrument (Appendix B), used as a reference for this research study.

Interviews were conducted one-on-one using open-ended questions as a guide to obtain teachers' perceptions on social presence and instructional methods to promote language learning in an ESL blended learning environment. The interviews were conducted inperson and recorded using an audio digital recorder. Documentary analysis provided another source of data. A review of collaborative virtual and F2F assignments from the instructor, as well as course resources and materials, were collected. Documents on the LMS of the course were of importance to the data collection because hard copy or electronic materials were able to be closely examined beyond the initial distribution (Miller & Alvarado, 2005). Document analysis has been used to examine the content of written materials for insight into the importance of the social activities described (Bloor & Wood, 2006). The analysis process explores data contained within a document to provide insight into a social phenomenon (Bowen, 2009). Documents are "non-reactive" sources of data, meaning that they can be read and reviewed multiple times and remain unchanged by the researcher's influence or research process (Bowen, 2009, p. 31).

After the initial in-person class session, data collection also included four classroom observations. Marshall and Rossman (1989) define observation as "the systematic description of events, behaviors, and artifacts in the social setting chosen for study" (p.79). Observational data collection methods contribute by their focus on "natural" settings which allow the explanation of social processes and phenomena. Specifically, they can create an understanding of what people do and how this can change situationally and over time (Walshe, Ewing, & Griffiths, 2012). Observations occurred four times, evenly spaced, during the semester and each time the researcher stayed for the

entire class session (155 minutes). Notes for the observations were completed through the use of Google Keep. The tool is primarily designed to take and organize notes, record audio, and keep track of media and fit the design for this research, especially for taking field notes. The data collection is stored on a password protected Google Drive where Google Keep is located. In accordance with George Mason University's IRB guidelines, the data collected will be retained for five years before being disposed of by the researcher.

## **Data Analysis**

The process for analyzing, interpreting, and determining the validity of data was conducted through coding. First level coding was used in the initial process to create descriptive, low inference codes that were important in categorizing segments of data and provided a structure for a second level of coding (Punch, 2014). The second round of coding focused on making patterns with the codes to minimize and focus on major themes. "A pattern code is," Punch (2014) wrote, "a more abstract concept that brings together less abstract, more descriptive codes." After the establishment of the coding system, data were divided into themes and categories to compare and contrast results to reduce the amount of data being analyzed (Creswell, 2015; Richards, 2015; Saldaña, 2016).

The process for data analysis used to accomplish the goal of documenting the ways in which ESL faculty perceive social presence and language learning in a BL community college learning experience will be discussed below according to research questions (RQ).

RQ1: What is the approach of an experienced community college English as a Second Language teacher to structuring a blended learning course to promote language learning?

RQ2: What are the teacher's perceptions of assignments that most effectively fostered social presence and promoted language learning in an English as a Second Language blended learning community college course?

RQ3: What are the teacher's perceptions of assignments that least effectively fostered social presence and promoted language learning in an English as a Second Language blended learning community college course?

To ensure accurate data collection, the interviews were recorded. Seidman (1991) recommends that interviews be audio recorded in order to preserve the words of the participants, and to ensure that interviewers can return to the transcripts to check for accuracy. The instructor as informed of the recording and signed informed consent forms. The following steps were taken to analyze the interview data. First, I transcribed the interviews from the digital recorder. Next, during the transcription process, preliminary categories were digitally noted and given a code. Interviews were transcribed verbatim and analyzed thematically and inductively using Dedoose qualitative software. I then read through transcripts, making notes and placing codes in memos with Dedoose.

Observation notes were kept in a password protected Google Drive in Google Keep.

The participant was identified on the typed transcript by the pseudonym "Linda." The resulting document facilitated further analysis and established a permanent written record that could be shared with interested parties. Added to the transcript were notes on

observations that I took during the observed class sessions. The transcript, notes, and academic F2F and online activities were combined and coded manually according to themes. The instructor role in the fostering of social presence and language learning for the student learning experience was examined. Therefore, the instructor interview questions were developed to determine Linda's perspective on her methods of creating social presence in the BL model, as well as her implementation and planning of a BL course for ESL students' language learning. The purpose of this research was to suggest ways to improve student language achievement and the quality of the blended language course. Secondly, the propositions of this research are the effective factors for a blended language course based on the analysis of a teacher's perception.

Transferability refers to the ability of results of the qualitative study to be transferred to other contexts or settings. To engage thoroughly with the data and get a full view of the subject of study in context, Geertz (1973) recommends "thick description." Thick description goes beyond surface data. It establishes the significance of an experience, or the sequence of events for the person or persons in question. In thick description, the voices, feelings, actions, and meanings of interacting individuals are heard (Denzin & Lincoln, 2005, p. 83). Thick description affords the reader the opportunity to determine whether findings may be appropriate in another setting; this is not possible without a rich description by the researcher of the studied environment.

### **Researcher as Instrument Statement**

Because the researcher is the instrument in semi-structured or unstructured qualitative interviews (Pezalla, Pettigrew, Miller-Day, 2012), it is important to reflect on

my attributes as a researcher and the professional development that has informed my practice. In my career and current work as a doctoral candidate in the higher education program at George Mason University, I have been guided to expand my thinking about the nature of evidence and the concept of researcher "as the primary instrument or medium through which the research is conducted" (Lofland, Snow, Anderson, & Lofland, 2006, p. 3).

First, I received a Teaching English as a Foreign Language (TEFL) certificate in Rome, Italy, during my undergraduate studies. The training provided me with introductory, intensive coursework in linguistics and practicum experience teaching adult ESL students ranging in levels of English proficiency. With the certificate, I received a teaching position in Naples, Italy, and taught English to elementary students and adults. The varying range of abilities among the students allowed me to differentiate instruction while still maintaining best practices for language learning. The institution in Naples did not have access to technology and therefore the materials used were print textbooks and journals. Second, I hold a master's degree in special education and a bachelor's degree in elementary education. I have been licensed in Virginia to teach K-12 special education subjects, as well as K-6 elementary education curriculum. My experiences as a general education teacher at the elementary level for the past nine years have enabled me to work closely with English as a Second Language students and their families. Through Fairfax County Public Schools, I have attended blended learning trainings, served on the technology committee of my school, acted as the school's equity leader, and adapted and created curriculum to fit the ESL population's needs using a flipped model approach.

Third, my current experience in the doctoral program has provided opportunities for internships in instructing ESL students in the community college and 4-year environments and assisting with qualitative research. Through INTO Mason, I was able to conduct workshops for ESL professors in blended learning as well as to tutor international ESL students related to their coursework. The experience provided practice in editing academic work and offering feedback that will promote student understanding of the English language. At Northern Virginia Community College in Woodbridge and Loudoun, I was able to encounter a hybrid course and understand the technology options available to instructors from the institution. I assisted faculty with lessons and introduced them to a blended model. During my opportunity at NVCC, I was able to use my blended learning training to move beyond the textbook and paper and pencil exercises that might not appeal to all learning styles in the classroom. The time of the practicum was limited but gave me an overview of the ESL program at the community college.

Finally, my program of study included coursework in linguistic competencies, community college trends, and, as stated, work on qualitative research. The main duty of the research consisted of conducting semi-structured interviews, transcribing, coding, and writing about the themes that emerged. Specifically, I used a phenomenological approach to gain insight into perspectives of faculty involved in self-study and the inclusion of digital andvisually rich media in the classroom. For the data analysis, the structure and meaning were relied upon to draw from the participants' experience with reflection and technology.

### **Summary**

In order to accurately describe the BL experience of an ESL CC course, a qualitative, exploratory single-case design was used. Case study research is recommended in instances where the use of qualitative data is sufficient to fully describe the phenomenon from the participants' perspective. As the study is holistic in nature, it was determined that this was the best methodology for the task at hand. The research consisted of qualitative analysis of semi-structured interviews, virtual and physical documents, and class observations from an ESL blended learning faculty member in a reading course on the campus of a community college.

The goal of the research was to richly describe the BL experience of an ESL community college instructor and her perception of social presence and effective language learning pedagogy in the classroom community. The learning experience elements, such as teacher perceptions, engagement, motivation, successful acquisition of knowledge, and course delivery method, were examined under the umbrella of the social presence construct from the CoI framework and guided the categorizing data collection and analysis. The ultimate result of the findings will allow faculty at similar community colleges to offer BL in a delivery method that is grounded in research to assist ESL students in their language learning while building a strong sense of community for learners.

#### **CHAPTER FOUR: FINDINGS**

This dissertation explored the experience of an ESL instructor in a blended learning classroom at a community college campus in Northern Virginia. The researcher sought to understand how instructors develop blended courses to promote language learning and social presence in both face-to-face and online modes. The specific assignments that were perceived by the instructor in the hybrid environment to be effective or ineffective were also explored. The researcher employed an exploratory single case study approach to generate new ideas and examine practices for blended learning language teachers. This chapter presents findings from the study that answer the following research questions:

RQ1: What is the approach of an experienced community college English as a Second Language teacher to structuring a blended learning course to promote language learning?

RQ2: What are the teacher's perceptions of assignments that most effectively fostered social presence and promoted language learning in an English as a Second Language blended learning community college course?

RQ3: What are the teacher's perceptions of assignments that least effectively fostered social presence and promoted language learning in an English as a Second Language blended learning community college course?

This chapter begins with a descriptive profile of the learning environment, classroom structure, and Linda's perception of the nature of a hybrid course. The chapter then highlights four broad areas: F2F format, LMS and online activities, communication, and challenges to detail the instructor's approach and develop an understanding of more effective and less effective assignments in the blended environment. The chapter concludes with a summary of findings. The data collection consisted of an extensive preand post- interview with the course instructor, document analysis from the LMS and F2F learning environments, and four classroom observations. Data was collected for 12 weeks during the spring semester of 2019. Data focused on how the teacher perceived the establishment of social presence, both F2F and online, and how the design and implementation of a blended learning approach influenced language learning.

# **Learning Environment**

During the semester of data collection, spring 2019, Linda taught four courses. Three were F2F. The course observed for this research, level four ESL reading, was the only hybrid course she taught at the time. This specific campus only offers hybrid courses at the upper language levels because faculty feel that lower levels benefit more from increased in-person interaction. Other campuses in the NVCC system offer hybrid for all ESL levels. The class met once a week in person for two hours and 35 minutes with an average of five to six hours of work to be completed online, outside of the classroom. The class was at capacity at 24 students and Linda remarked that the hybrid class typically filled due to its popularity. However, due to the increasing number of ESL students at the

campus, the traditional F2F classes filled to capacity during the spring semester as well. When asked why the hybrid course is popular, Linda explained:

I think for a lot of them, it's schedule reasons. Most of them live and work here in the community. They have family obligations, work obligations, and it just makes it easier for them to get their work done and attend class.

In the course observed, Linda states that the student ages ranged from 17 to 77, although she recalls that most of the students were 18 to 22. The students came from diverse linguistic backgrounds. The location of the classroom was a computer lab, although the students generally did not use computers during the F2F class time. They completed a significant amount of work via computer outside of class meetings and Linda prefers the class to focus on face-to-face time when they are together in person. The lab was set up with long rows of tables on each side of an aisle with 3 seats per table, leading up to a teacher's desk in the upper corner with a computer that can project on the whiteboard in the front of the classroom. The students tended to remain in the same seats throughout the semester and therefore had discussion with the same group of people during discussion tasks. In order to get tables of students talking with one another, Linda would often facilitate discussions and rigorous tasks among tables involving short stories. She would choose a character and have students identify a problem the person was experiencing. Through discussion, debate, and time to develop their ideas, they would agree on a solution which then they would share with the whole group.

During F2F time, all of the students consistently came to class and even arrived 20 to 30 minutes early to work on assignments or go over material from other courses.

During the four observations, only two students, at separate times, were absent for a class period. Linda observed that everyone participated in the online posts and homework assignments. Some students turned in homework after the due date, but it was not enough to label this as a challenge in the course. Linda reported that two students did not complete the course; in one case that was due to a family emergency.

## **Classroom Structure and Assignments**

The research took place during the spring 2019 semester in the reading course which ran for 12 weeks from the end of January until the first week of May. Linda created the course in a blended learning model, using the classroom time for activities that benefit the most from direct, in-person interaction. The F2F space was dedicated to structured exercises that applied the topics and themes from the at-home, online practice. The students regularly discussed main ideas and symbolism together and worked through specific learning tasks. During the semester, students completed a significant amount of work online that informed their in-person classroom discussions. Linda purposely used discussion boards to cultivate connections outside of the traditional classroom walls.

Linda held in-person office hours on campus on Monday and Wednesday from 11:30 a.m. - 12:30 p.m. and on Thursday from 12:30 to 3:30 p.m. In addition, she made herself readily available to her students virtually by phone or email and scheduled additional F2F meetings with students when requested. Linda designed the course to help students improve their reading comprehension and vocabulary in order to move to, and succeed in, ESL level five and beyond that, to credit-bearing college courses once students complete the ESL program. The online class assignments emphasized reading

and responding to readings in written form. The other course objectives from the syllabus included: integrating reading, writing, speaking, and listening skills to promote thinking and learning; becoming active, independent learners able to apply integrated strategies appropriate to reading-related tasks; increasing self-confidence in one's ability to be an effective reader; and understanding the role of reading in academic, professional, and personal life.

The course assignments included tasks of varying degrees of difficulty that students, for the most part, completed during the asynchronous time. Reading homework, relating to assigned articles and short stories, consisted of questions about the setting, symbolism, and comprehension and instructions usually required answers to be supported by evidence from the text. The reading posts were done on the discussion board and equaled 15% of the final grade. In connection with the reading homework, summaries of readings, that were typed at home and handed in at the beginning of F2F class sessions, made up another 15% of the student grades. Additional homework assignments, related to the short stories, comprised another 15% of the total grade. For example, students wrote journal entries from the character's point of view with guiding questions to help them develop a post with detail.

The vocabulary portion, worth another 15% of the final grade, was broken down into three parts over a three-week cycle. The first week with a vocabulary list had students finding definitions of words that were tied to articles and short stories. Students chose from the list of words and typed the original sentence where it appeared in the reading, the part of speech, and a clear definition of the word. The second week, students

worked to apply the vocabulary word and developed original sentences of their own while also completing a matching practice. Finally, in the third week, students completed vocabulary quizzes on the list of words. The students had 15 minutes to complete a matching quiz online. The quizzes could not be made-up if missed and Linda used the exact definitions from the practice matching from week two. The cycle continued with a new list of vocabulary words throughout the semester. Included with the vocabulary work and grade was the program *ReadTheory*, an online reading comprehension assessment tool. Linda had students sign-up for the program which is free of charge for students and instructors.

The remaining portion of the class assignment grade consisted of an advertisement PowerPoint project that compromised 10% of the grade and contributed to many in-class discussions. The first exam at the beginning of the semester compromised 10% of the grade and the final exam included three separate portions, reading, vocabulary, and summarizing, completed the last 20% of the final grade. The students needed a 75% or higher in the class to pass for the semester.

The class utilized Open Educational Resources (OERs) and students were not required to purchase any additional books. Linda posted all materials on Blackboard and provided additional paper copies in class as requested by students. The course did require that the students have access to a reliable computer at home or on campus for weekly hybrid work. In addition, students could use the classroom itself as a computer lab for their academic work when class did not meet in person. The course requirements specified that students have the technical skills of turning on a computer, accessing the

internet, sending an email, and typing well enough to produce a one-page document during a class period. The online responsibilities of the student consisted of logging in to the LMS at least twice a week to complete assignments at home or an alternate location and contacting the instructor via email if they experienced technical difficulties with an assignment.

Linda outlined her responsibilities as an instructor of a hybrid course on the LMS in the class syllabus. The syllabus stated that it is the responsibility of the instructor to respond to student email and address student problems within 24 hours of receiving the email on a weekday and within 48 hours of receiving an email on the weekend or during breaks. During the interviews, Linda shared that she tries to respond to emails as quickly as possible regardless of the day because student-teacher connection is important to her.

# **Nature of Hybrid Courses**

Linda believed that blended learning allows students the flexibility to complete their online work and attend class while still meeting all of their family, work, and other outside obligations. Success in a hybrid course, however, requires an understanding of the nature of the course design and a commitment to the course requirements. Students as well as teachers need to understand that a blended hybrid is not half a course. Linda stated, "I think people don't understand what hybrids are." She continued, noting that students often find the course in the catalog and say "Oh, it's a 5-credit course but I only have to go once a week. I'm going to take that one." She provided guidance to students by explaining that the course requires as much work as a F2F class. She reminded students that, "you need to carve out your own time for yourself," to do well in this class.

She recommends that students clearly schedule time during each week, saying to themselves "this is my school time." If they are unable to do that at home, she recommends that they "need to leave the house" to create dedicated time to complete class assignments. In addition, Linda shared her experiences with colleagues and adjuncts and will often help them build their LMS and provide feedback on what works well in a hybrid setting.

The remainder of this chapter will explore the findings within the context of the research questions. The questions seek to determine how the instructor in this study experienced designing, and teaching in, a BL environment. A detailed analysis of the preand post-interviews, class documents, and the researcher's classroom observations provided the data. The qualitative data findings are presented in categories to highlight effective teaching in the ESL blended environment and assignments that lead to both language learning and to building a sense of community. The categories include: organization of the F2F format, the use of the LMS and online activities, communication between students and communication between students and the teacher, and challenges in both modes that could contribute to assignments being less effective for language learning and establishing community.

### **Organization of the Face-to-Face Format**

The category of organization of the F2F format centers on the teacher's approach during classroom sessions and her instructional practice to guide students' language learning. In her approach, Linda encouraged students and provided explicit directions and expectations on assignments. During her lecture period, the students seemed engaged and

actively listened while taking notes. Linda permitted students to call out questions and ideas and did not interrupt them until they were finished. She did not use rephrasing with student ideas, allowing the student's original statement to remain intact, and she consistently related student comments to the main objectives of the day. Her tone and manner were calm in class sessions. She started most class periods by introducing the agenda and then focused on addressing any misconceptions from the online forum or homework. Next, she introduced the topic for the class session and gave a lecture-style lesson on that topic. She also allowed 20 to 30 minutes during class to explain directions for upcoming assignments and projects.

# **Explicit Instructions**

One of Linda's strategies is to use class time to discuss instructions for projects and upcoming assignments. She does this to overtly demonstrate how to complete an out-of-class task which can be used to teach basic and higher-order reading skills. She set routines and used consistent language, so the students learned the new skill in the task but also the language associated with it. During one class in February, Linda explained a major project that was to be completed at home. She spent 30 minutes of classroom time stating expectations, guiding students to annotate the directions to ensure completion of all elements of the project, and reviewing the components of the grading rubric (Appendix D). In her explanation of the project, she stated, "Reading is about reading and making connections. Take your pen and annotate a little bit, because there are important things that you need to know to do well on this assignment. We will go through the project packet." She opened the discussion with the due date and stated that the

assignment would be submitted via email. She highlighted this information on the overhead where she also projected the project summary.

The project asked students to develop a PowerPoint presentation displaying an advertisement on each slide. The students needed to choose a total of five advertisements to demonstrate their knowledge of at least five of the commonly used advertising techniques that they had discussed during in-class sessions. The students needed to state which advertising technique they thought the designer or artist meant to use to draw consumers into the picture. To prepare for the assignment, Linda posted more than 70 advertisements in a Blackboard folder. She stated with emphasis, "You will not print. It is going to be digital, you are in a hybrid class" She annotated notes for each section of the project on the packet and told the students to do the same for what should be on each slide for each individual advertisement. The students expressed confusion and it took time for them to understand the project. Almost every student took the time to annotate to make sure they understood. Linda showed one slide for an advertising technique of "direct gaze" as an example. Her tone during directions was even as she spoke slowly and kept her voice neutral when answering numerous questions. The class seemed confident in following the parameters of the project by the end of class. Linda was pleased with the final results of the advertising assignment and she noted that most everyone followed the directions.

The project took the place of a test and that made her "hard on the deadline." At the end of the explanation, Linda drew attention to the bottom of the rubric (Appendix D) and reiterated that this was a big project. She stated, "You will lose 10 points if you don't

spell check and 10 points every day that it is late." There was nervous laughter in the room among the students about the amount of points someone could lose. Many students commented on the importance of turning the assignment in on time. Linda deliberately scheduled the due date after spring break to allow students additional time to work on the project. At the end of the discussion, before moving on to the day's topic, one student asked, "Where do we put our names?" The class worked together to add an extra slide to the directions to include student name, project name, and the date. "I assumed," she said to the class "and I didn't count that."

Linda would often spend large amounts of class time on the expectations of projects, assignments, and exams to ensure that the students comprehended the directions. She commented that detailed directions are especially important in an ESL classroom where the students are navigating vocabulary for which they sometimes do not have background knowledge. The students needed accountability measures with assignments and continued to need explicit instructions. Linda used F2F time for explicit instruction for the success on assignments, but also to help students develop background knowledge on a particular topic. Linda found it beneficial to spend time reviewing previously learned concepts, checking whether students had understood concepts, and correcting misconceptions during the lesson.

### The Use of Advertisements

Linda used a diverse range of instructional materials to provide language instruction in the F2F format. She incorporated visuals, short stories, and songs to engage students in the learning environment. The advertising PowerPoint project provided

individualization and academic choice and contributed to language learning as well as social presence through partner discussions about the advertisements prior to the independent task. With the assignment, Linda also had students break down the symbolism in the visuals and portray their carefully considered assessment of what the ad was trying to achieve. The last slide on the PowerPoint provided a space for the student's analysis about why they thought the advertisement was using particular techniques (color, repetition, direct gaze, association, body language, composition, rule of thirds, vectors, focal point, symbolism).

Students using the same advertisement could have differing opinions about which techniques were being used. They received full credit if they were able to explain themselves and justify their answers. Linda reported that she was "happy with the results" and mentioned the improvement in the use of vocabulary by the students, which had been a struggle during the semester. Visual materials worked as a powerful tool in the project because they gave Linda the opportunity to show the culture of the target language in a unique way. The visual impact of images has proved to be superior to the one of texts (Clark and Lyons, 2004), and that is why visual aids can be very effective to help students in memorizing new vocabulary and structures.

Linda perceived the in-class discussion of the project, used to build background knowledge on the advertisement techniques, and the PowerPoint as a whole, to be the students' preferred activities. The researcher observed evidence of social presence during class sessions used to prepare for the assignment. For example, students participated in active listening with each other and worked to weave their own comments about the

visuals into the larger conversation at their table. Students listened to each other and prompted further discussion when the speaker of the group did not say enough or was stuck on an idea and could not find the words to express themselves. The students seemed to listen not only to understand one another, but also to add to the conversation and participate in the group. Students consistently added on to ideas and comments of their peers during small group tasks.

In addition, the students made connections between each other's comments and used specific evidence from the advertisements to foster conversation. The advertisements provided a platform to share opinions without having a "correct" answer. This allowed students to learn from each other and experience multiple perspectives about the same visual. The exchange of ideas allowed the students to see that they can learn from other members of the learning community and not rely solely on the instructor for insight and information. Linda believes that the visual aspect of the assignment fostered engagement from the students. Interpreting pictures together are things that they do socially online already and for many students, sharing photos is a daily feature in their lives. Linda points out that the assignment also helped students make connections between the symbolism in stories and articles. She selected pictures that were purposely dynamic and interesting, and the students created a rich discussion about the meaning behind the imagery.

Key affective components of social presence are humor, emotions, and selfdisclosure (Leafman & Mathieson, 2014). The researcher observed students in the course engage the areas of social presence during collaborative discourse of the in-class discussion surrounding the advertisements and varying techniques. During one occasion, advertisements were used from a homework folder titled, *Cigarette and Other Ads*.

Students who appeared disengaged during the whole class discussion, participated fully and in English. One student's group members listened to him purposefully as he discussed being a former smoker and they all laughed about the company's approach of using attractive women to lure men into buying cigarettes. Another pair of students had an in-depth discussion about an ad with the caption, "Forests are the Lungs of Our Land." One of the students clarified for another student why the ad used the word "lungs" to represent forests. They discussed the issue further and agreed what advertising techniques worked for the ad that the class had already reviewed together. Based on the researcher's classroom observations, the advertising project promoted students to bring forward their own interpretations about the advertising techniques that companies use to influence consumers, such as themselves, in their daily lives.

## **Literacy Instruction with Short Stories**

During in-person class time, Linda integrated short stories that the students had read at home. The objective of using the story was finding the main idea and summarizing, but also analyzing and discussing the symbolism in the story. Linda used literary symbolism, where an author uses figures of speech, such as metaphors, similes, and allegory as tools, to help students to understand the abstract concepts more concisely and deeply comprehend the character and their actions. Through the use of symbolism in the short stories, students expressed themselves freely in the English language without feeling that they had to choose a side in a debate to please the teacher. They focused

instead on conveying their opinion on the characters' actions and the elements in the story. During the interviews, Linda explained:

I think maybe it's just less pressure. Finding main ideas is a skill that they're not used to and so they're hesitant. They don't want to look dumb in front of their peers. But with symbolism, pretty much everyone doesn't know what's happening and so I think they all felt like they were on a more even playing field. It seems like they always talk more, not just this semester but every semester, during that time. And it's an opinion rather than an answer.

Students in the reading course were also able to connect and engage with the short story material because of the planning and implementation that Linda provided before the course. Stories can engage the learner at "more than a cognitive level; they engage our spirit, our imagination" (Clark & Rossiter, 2008, p. 65) because they evoke the learner's prior experiences. In class, students contributed to thoughtful discourse about the stories and listened to each other's opinions and agreed or respectfully disagreed, often with laughter and positive responses. Short stories are appropriate for language learners because they often give multiple viewpoints and styles. Linda's selection often told a narrative about a culture as well as a person and allowed students to connect with the character while understanding different perspectives. *The Somebody*, by Danny Santiago, is a short story that narrates a day in the life of Chato de Shamrock, a teenager who feels neglected and wants to be known by everyone. The main character is of Mexican heritage. The main reason Danny Santiago wrote this short story was to show how somebody can change their identity because of certain things they are experiencing in

their life. Students in the class had a rich discussion while discussing elements in the story. Students discussed that the language of the story was simple. They acknowledged an understanding of what the author was saying, what the character was going through, and how the character was feeling. Students also related to the character by establishing a connection that many of their family members were far away and it could feel isolating.

Linda set a goal with short stories to go beyond a basic comprehension level because of the level of the course. For example, she went beyond discussing the conflict in the story and instead had students analyze the ways in which a character's actions impacted the conflict of the story. When Linda asked students to respond to one another in small groups, she walked around and listened to the differing opinions in order to reflect with the entire group at the end of the discussion. The use of short stories also helped the ESL students with summarizing because there are limited subplots to confuse students unlike a novel. The clear, concise plots guided students toward the intended learning objectives without overwhelming them with language and text.

## **Language Instruction Around Song**

Linda chose material for her course to maintain learner's interest throughout lessons. Her goal in selecting music was to provide universal appeal that connected cultures. Classroom observations showed that students actively engaged in the English language while analyzing the song, "Cats in The Cradle" (Chapin, 1974). The discussion observed in class on the song provided a motivating resource for all ages and backgrounds and encouraged language practice. The students came to class prepared to engage in discourse on the song due to the assignment of analyzing the lyrics prior to

class. When starting to discuss the song F2F, students commented on how much they enjoyed the song. During discussion, the students agreed it was a sad song. They shared similar opinions that the father did not have time for his song, and although working was the right thing to do to provide for the family, it was unfortunate.

The students continued to make connections to the song around the room, sharing personal anecdotes with their peers about their own lives. Their family obligations and their own children provided a way to share how the song made them feel. During these times of connection through visuals, stories, and song in the F2F format, the students demonstrated comprehension of the English language through authentic interactions. In the F2F classroom setting, Linda used instructional strategies and materials that appealed to the students who represented a range of ages and cultures. Though a large amount of classroom time was spent providing explicit instruction on tasks, the instruction went beyond lecturing due to high levels of questioning from both the students and the instructor.

Through Linda's modeling, the students utilized annotation and organization skills in their independent assignments. In addition, Linda used resources that captured the students' attention. Linda introduced advertisements to guide students in critically reading text and interpreting imagery. The students focused their attention on how the words and images worked together to communicate a message. The short stories chosen helped students demonstrate their language skills, critical thinking skills, cultural awareness, and creativity. Lastly, Linda introduced songs to engage students in discussions about symbolism and interpretation. The students selected metaphors in the

song to represent their viewpoint and debate the meaning. The researcher observed students using these materials and guidelines as they consistently immersed themselves in English language learning.

#### LMS Utilization and Online Activities

The LMS for the blended class served as an engine to run Linda's online classroom. The LMS that the campus used at the time was Blackboard. Linda's longtime experience with hybrid courses has led her to find engaging resources that the students can interact with on their own. As previously mentioned, she assigned readings of short stories and articles for homework and had the students complete a variety of tasks asynchronously based on the stories. The students had to post weekly about varying questions related to the readings and to respond to at least two other members of the classroom community. The vocabulary work was also done online on the discussion board. Linda used technology tools to increase feedback and connection with her students by an online delivery of content through numerous videos for guidance, self-check quizzes with the use of the online reading comprehension tool *ReadTheory*, and interaction through discussion boards.

### **Discussion Board**

The ongoing assignment of the class that Linda perceived to bring social presence, collaboration, and discourse were the weekly discussion board posts, which she also perceived to promote language learning. Discussion boards used in the course provided many opportunities for students and the teacher to engage with one another during the semester. Linda noted that she felt students liked the self-paced and self-

regulated asynchronous discussions and that participation was very high. Linda provided time for the students to reflect before replying to fellow classmates which enabled less vocal or more introverted students to make their voices heard virtually, allowing them to become a part of the community. During online work, Linda felt it was helpful for students to observe other responses first before attempting their own posts. Student posts were guided, analyzed, complimented, and supported by Linda and peers routinely. She did not have students who did not participate or made no effort to post during the semester. This system allowed students to remain a member of the learning community beyond F2F time.

The interactive element of social presence includes acknowledgement, appreciation, and return of communication among learners (Garrison et al., 2010; Rourke et al., 2001). The students demonstrated those elements throughout their posts. Some students selected their first introductory post to a partner who was from the same cultural background as themselves and acknowledged their commonality in their responses to one another. In their work, Biocca, Harms, and Burgoon (2003) identify social presence in terms of special presence, the sense of being present, and physical presence, the sense of being together with others. The conversations established from the LMS discussion board enabled students to feel comfortable in the virtual environment in expressing themselves to one another and finding commonalities.

Linda implemented online discussion board posts as a way to offer ESL students a tool to actively utilize the English language without feeling anxiety about their response.

Conversations often worked more fluidly online than in the classroom where many topics

were covered in a session and time was limited for extended conversation. As students reflected about their posts, the tasks frequently required them to integrate class readings and dig more deeply into texts. The use of discussion boards as an active method of language learning enabled students to retain and apply reading strategies.

#### **Short Stories**

Through the discussion boards, Linda used the short story assignments to aid students in interaction with the English language online. Similar to the use of short stories in the F2F class, Linda had students work online on comprehension, vocabulary, and critical thinking tasks by posting responses based on the literature. Linda required students to reply to at least two other students for each post. Linda preferred short stories that had a universal theme where the students could reflect on the plot points and relate them often to their own experiences. The online posts would be directed towards the symbolism in the title and in the story, as well as literal comprehension questions such as the setting. Linda also asked students to analyze decisions by characters and put themselves in the mind of those characters. These tasks allowed the students to share in their responses about their background knowledge and experiences in relation to the text.

When the students were asked inference questions or questions related to their opinion, Linda emphasized using evidence from the text in their posts. She established guidelines for how to properly quote text on the LMS to guide students. In replies to one another, students agreed with others about their opinions or used it as a platform to respectfully disagree or share an additional point. The students responded positively to

each other and it appeared to the researcher that the stories chosen elicited student involvement in rich discussion, analysis, and responses.

### **Communication Across Modes**

In designing effective hybrid course assignments, Linda considered that students would be coming into her classroom with individual learning styles. She selected multiple course delivery modes to appeal to a diverse student population. The decision to connect the F2F class agenda to the online portion of the class came, in part, from a desire to motivate students to stay engaged due to the factor of being prepared for class. Linda found it important to create a sense for students that they had to be equipped for the F2F classroom with the hope that it would motivate them to finish online assignments by the due date. The topic of communication covers two aspects of discourse that happened in both the F2F and online formats. The first centers on student-to-student interaction. The second focuses on student-teacher interaction. Throughout the course, students developed their communication skills through the frequent and direct contact that they had with the online communication tools, thus increasing their engagement in the course whether it was F2F or online.

#### **Student-to-Student**

Through her online forum, Linda monitored the growth of her students' language learning by setting tasks that were based on a range of cultural and social interests.

Students demonstrated high levels of interaction which translated to success in group discussions in the classroom because they were prepared and familiar with the materials.

Referring back to the story of Chato, students made connections to the story on posts and

related back to these thoughts once in class to further the discussion. Students also often communicated using comments to offer feedback to each other on posts. During a F2F format, the researcher observed students discussing the comments made by their peers in the online setting and asked for further clarification. The students seemed comfortable asking classmates follow-up questions related to other items on the agenda that the teacher discussed during class time. This indicated student comfort in asking classmates follow-up questions related to other items on the agenda that the teacher discussed during class time.

#### **Student-Teacher**

Student-teacher communication was of the utmost importance to Linda and she enjoyed being able to connect with her students using multiple modes. The time that Linda dedicated to responding to each post on the discussion boards seemed to facilitate students efforts to remain accountable and active online. Linda purposefully connected these posts to classroom conversations making the F2F time a way to create a social community as well. Linda stated in interviews that she wanted to encourage her students to be independent and active learners where participation was required, but also wanted them to feel appreciated. If students had difficulties, it was imperative to Linda for them to ask for help, which she often spoke of during class sessions and in the LMS platform. During the F2F observations, students appeared eager to participate and accept teacher feedback. With feedback, Linda saw many improvements, especially in the discussion board. The main points of feedback came from some of the class objectives such as

organizing a summary, following directions with multiple parts, and expressing main ideas.

In a discussion post about a short story, students would be given multi-part directions and questions to answer about the elements of the story. This allowed students to practice identifying and relating main objectives of the reading course as well as following directions and developing the technology skills to post online and interact in an online forum. Some students experienced difficulty with posting when required to submit multiple parts to an answer. One student posted three different times about the same post and took advantage of the rewrite that Linda offered the class if their attempts did not meet the full requirements. On the first attempt before the due date, the student answered what the definition of the word "setting" meant but did not explain the symbolism of the setting or the symbolism of the title for this particular story. On the second attempt, the student went further into detail, but over-explained the setting of the story and copied words directly from the book. The student also gave too much information about symbolism overall instead of answering the specific question about the title. The student also omitted the answer to the fourth part listed in the directions which was an inference question about an event in the story. The rewrites that Linda allowed, with her feedback to work from, proved to have a significant impact on the student's grade for these posts. The second chance also served as motivation to work through the material in order to improve and accept constructive criticism from the instructor to allow for improvement.

Online communication between student and teacher, especially email contact, was fostered, in part due to Linda's preparation before the course. The excerpts below from

the LMS show Linda's effort to give students guidance before the course began in the online syllabus and to stress the importance of email communication during the course:

Our learning is taking place across the week, sometimes in our classroom, and sometimes while you are at home. If you have a question or cannot open an assignment, PLEASE EMAIL ME. Do not wait until we are face-to-face. As a hybrid teacher, it is my responsibility to be available to answer your questions when we are not face-to-face. There is no such thing as "too many emails" and you are NOT bothering me! I am available for extra help if you need it. 12-week courses are very quick. If you feel like you are falling behind and would like some extra help, please make an appointment with me. I am always happy to help you.

Linda emphasized that the students had a responsibility to check their email regularly for class announcements. She created an email assignment and asked the students to respond in order to make sure they were familiar with how the communication system worked. In addition, she outlined how to format a professional email on Blackboard and the necessary information that would be included. Her perception about the communication process, as conveyed during the pre-interview, is that:

In a hybrid class, you're not bound by being in the classroom with the teacher. When they're online working, I'm supposed to be available within reason to help them and respond to their questions, so they were a great class for emailing, for asking for help. And I said, 'if you were in a face to face and you were going to raise your hand five times, then I expect five emails.' That's how it works and you're not bothering me. And so that piece of it, I think it helped.

Communication was an important skill for every student to practice when learning the English language in Linda's classroom. Linda maintained her dedication to regular and timely feedback. She felt the extra effort was necessary for her students to maximize their learning. She needed to facilitate the communication for the students and make the learning environment comfortable for academic conversations.

## Challenges

During the course, Linda also had challenges regarding her hybrid classroom structure. She shared concerns involving technology skill issues, creating clear directions for students to follow, and plagiarism. Based on interviews with Linda, these issues tend to repeat themselves during each semester. However, Linda expressed that with adaptations to the course syllabus and online format, incidents in these categories were slowly decreasing. She developed an entire section in her LMS titled, "Before the Course," for students to review in order to minimize these challenges. Implementing specific precautions and accountability measures described in the sections below, Linda has seen a change in the problems that occur during her hybrid course. The challenges described, however, can still pose a barrier to student learning in the blended environment.

## **Technology Skills**

Linda noted that teachers need to consider that there are physical typing skills, but also understanding basic computer concepts that may be difficult for students. In the blended environment, students are completing assignments on the computer, often without the guidance of the teacher. Linda observes that older students and students from

countries with less access to technology tend to have more difficulties with computer skills than others. Depending on student backgrounds and experience with computers, Linda makes accommodations for students by allowing them to handwrite assignments and she regularly posts free typing tutorials on Blackboard.

Assignments such as online posts and homework are more time consuming for students who do not have basic typing and computer literacy skills. Some do not intuitively know how pages work or that you might navigate to a page by clicking an icon and these difficulties, in a blended course where technology is prevalent, can make it more challenging for a student with these lower technology skills to succeed. She also has allowed students not familiar with PowerPoint to make projects on Word document pages instead. She implements keyboarding practice on Blackboard and directs students to tools available online for free.

Linda observed that students who are slow to succeed with their online activities usually have the largest issues with technology. She reiterated that the course in this study is the first time that students have an opportunity to take a hybrid if they have been at this particular campus since the beginning of their coursework. However, the level two and level three courses do use Blackboard for announcements and communication of assignment expectations, so they likely had some exposure. She also noted that students who are entering NVCC at level four usually do not have any previous instruction with an LMS or online tools. Linda did not want the BL environment to be a barrier for students and encouraged those new to NVCC to take advantage of her office hours to go over the tools one-on one. Although strict with her assignment deadlines, she added that

she is "softer for the first few weeks, because if they didn't understand how to get access, they're behind and it's no way to start off a semester."

## **Following Directions**

Reading comprehension proved to be a challenge for students during the course in the area of written directions. The students had great difficulty following written directions, often completing one part of an assignment, but not other portions. The challenge of adhering to instructions had a significant impact on grading for these students. A large group of students could not perform activities correctly because they misread or misunderstood the instructions. Online, students would answer only one or two parts of a three-part question. In class, when taking an exam, students had difficulty completing multi-step questions as well. Linda routinely modeled the process of annotating directions to important assignments in class while the students practiced as a whole group in order to prepare them for reading and interpreting the directions on Blackboard at home and in the F2F format on exams. On Blackboard, she also posted numerous reminders that understanding directions is a critical component of being successful on an independent assignment. In the syllabus, Linda included two separate sections dedicated to reading and following directions:

Reading Directions and Spell-checking Assignments: In a hybrid course, reading directions is extremely important. Please take the time to read ALL the directions for every assignment, even if you think you know what to do. Assignments that do not meet the requirements in the online directions will not be graded and will

lose 10 points off the final score. Assignments that are not spell-checked will lose 10 points off the final score.

Reading directions is VERY IMPORTANT! Although the at-home work is based directly on what we do in class, I am not there to explain things to you. Please read the directions for each assignment carefully, especially discussion boards.

KNOW WHAT YOU NEED TO DO BEFORE YOU START! This saves both you and me a lot of time.

With the many class observations involving practice with explicit instructions, written feedback on the discussion board, and guidance on the LMS, the researcher noted that students have a difficult time comprehending written directions with multiple parts. Linda also included a large section on the ways in which to post on a discussion board which included etiquette and modeled recommendations for displaying key information.

Following directions for online posts was often a challenge and when intertwined with the difficulty that students had engaging in vocabulary lessons, it became clear to Linda that she needed to change the wording of her vocabulary instructions for the next semester. Linda gave directions in the online format to post authentically about vocabulary directly from a short story, in hopes that students would be able to connect to the reading. The expectations were to type the original sentence and the page number where the word could be found in the text, the part of speech, and a clear definition of the word. At the end of the post, Linda included an example. However, students still had problems rephrasing into their own words and following the directions that included

multiple parts. Linda experimented with different ways to present the directions in an organized fashion and said she will continue to work towards making explicit instructions in the future. Most of her online assignments on the discussion board involved multi-step directions and questions with more than one part. At the beginning of the semester, she also created a course requirement assignment which took students through the syllabus step-by-step and outlined the rules and expectations for the class. She hoped this would prepare them to read through and self-direct themselves in making sure they answered all of the questions. She shared that she will continue to adapt the way she presents her directions both online and F2F. She expressed routinely that the skill to follow multi-step, temporal, and sequential directions in reading is a crucial part of the course and in their careers beyond NVCC.

# Plagiarism

The issue of obtaining work from students of previous semesters or plagiarizing from the Internet came out as a significant barrier to successful language learning in Linda's experience in the blended learning classroom. She has rebuilt and adapted her syllabus and procedures numerous times to try to avoid the window of opportunities that a course could potentially provide to cheat on academic work. Prior to this class, Linda discovered the problem of cheating with former students giving materials to future students. Therefore, she implemented all new articles, which required her to create new vocabulary lists, and imposed accountability measures. In the future, she plans to update her materials regularly and limit the number of paper copies that she issues students

during F2F time. The redesign added hours to her pre-course workload for the current semester, but she did notice a notable decrease in plagiarism.

During in-class observations, Linda stated several times the severity and importance of academic integrity. Before the midterm exam, which was completed at home, she stated to students:

You may not use Google. If I find a cut and paste or Internet language, you will fail the test. I'm not trying to trick you in the exam, so I want to know that you understand what we've done. And it's independent, so each individual should be doing this in their own home. We are not working on this together.

Then again before the final exam, she reiterated to her class:

No Google. I will notice. I can tell when they are not your own words. If you need some help, please let me know. It has happened twice before, and I will give this speech again next week. You will fail the short story. It's your reputation and your education.

Technology tools aided her efforts in combating plagiarism. She frequently uses *SafeAssign*, a plagiarism catcher, which is included in Blackboard at the campus. This tool allows teachers to identify unoriginal content in students' papers by recognizing areas of overlap between course assignments and existing work. Linda dedicated large sections of the syllabus to discouraging students from acts of plagiarism. Part of her online feedback method is also used to stay alert to possible copycat posts. Linda responds to every post, but also carefully reads the mandatory replies from each student and uses SafeAssign to detect cheating. The risk of plagiarism hindered her from using

other online tools that may make cheating easier, such as Google Docs. She would enjoy letting the students collaborate and write inside a document on Google but is apprehensive considering that leaves the internet open to use in online collaboration and exams during F2F time. Linda noted, "some students, if they don't know what to do, they pull stuff and use it and in some cultures that's respectful."

Linda learned through experience to have private conversations with students she suspects have plagiarized on their assignments. A student in a previous class submitted the answer key that Linda had given out in a previous semester. Linda recounted that she "talked to [the student] and she said that she was scared, and she was worried, because she was scared to summarize. She was afraid she wasn't going to do it right." Research has shown that students' ability to write well in English tended to be low, so their confidence levels in writing were also low due to language barriers (Tran 2012). Linda attempted to have a meaningful conversation with her student and thought:

Everything was okay and then she also submitted someone else's vocabulary.

That's the first time I've ever referred somebody to the academic dean. I figured, with once, I feel like it's an opportunity to talk to them, but it was very disconcerting and disheartening, especially the second time. Basically her excuse was 'I just don't have time.'

On the LMS and in class, Linda established clear guidelines for what constitutes plagiarism and how to avoid it by allowing more time to practice the conventions of citation and referencing, which are important steps in maintaining academic integrity

(Tran 2012; Youmans 2011). She has also now taken steps to make sure the students cannot pull materials from the internet for at-home projects. For example, she explained:

The students had a shopping project. But again, one challenge has been that people are sharing their projects with each other and there's no way to test, I didn't keep them. It didn't occur to me, so I have no way, but I can tell I've seen this before. This semester, you know, they have to take a picture of themselves at the store and if they have this type of advertisement, they're supposed to take a picture of it. So I said, you need to have some identifying item in each picture, your face, a key ring, your daughter, you know, something. They aren't allowed to use the internet, one woman wrote her whole project, 'I went to Shopper's...and I saw this.' And all of her pictures were from Wegmans that she had pulled from the internet. And I'm like (laughter) . . . and so that's why I finally said you need to have an identifying thing in your picture because I can't tell if you took it from the internet or [not]. And I say, "most of you, I apologize because you're not going to do it." But every semester, there's enough. The cheating is the worst part. And the challenge of technology, sharing of materials, downloading PowerPoints that are answers or the discussion of something.

Linda communicated a few challenges in implementing the BL approach for language learning. She experienced technology issues involving skill level of students, an inability to follow directions, and students participating in plagiarism. In addition, Linda struggled with several assignments this semester, such as vocabulary, that lacked opportunities for interaction among students and opened the door to plagiarism. The

wocabulary work involved creating sentences with the intended word that showed meaning. Often, students would pull sentences from the internet without developing an understanding of the meaning. As a teacher, Linda found the vocabulary sentences exhausting and tedious to grade each week. She adapted the assignment for the next semester to allow for more engagement with students. She will be establishing a connection between the vocabulary and the short stories to make the assignment meaningful to the students. Linda also stated that there was not an established intent of social presence in creating assignments with collaboration in mind. In previous hybrid courses, difficulty with monitoring the participation of members and students meeting outside of class time have deterred her from group projects. She would like to try collaboration again with tools that she has available with the newly implemented LMS, Canvas. Adjusting her course to avoid challenges has proven to be a necessary and ongoing part of her teaching to continue the successful process of the blended learning model.

## **Summary**

This study aimed to capture the perception of a community college ESL instructor's BL approach for language learning and the effective areas of social presence in such a course. The participating instructor perceived that blended learning engaged students in a flexible and communicative approach to language learning. During the F2F classroom time, Linda used advertisements, short stories, and songs to elicit interactions among students in the areas of main idea, theme, and symbolism. Technology in the virtual space allowed for the design of a student-centered approach to learning with

routine and intentional feedback to enhance their learning. Linda focused on particular tools, such as discussion boards to individualize learning for students during short story and vocabulary exercises and to promote communication between students. While Linda appreciates the components of blended learning and has an overall positive outlook towards the design method, she also acknowledged that there are challenges to the blended learning approach. She mentioned that plagiarism is not a strictly hybrid challenge but accepts that the use of technology does allow for students to gain access to answers more readily.

Linda felt that students were able to engage more online and participate with the use of the discussion boards on Blackboard. She also experienced a stronger connection for students with language difficulties through email and providing immediate feedback than she does not normally have time for in her F2F classrooms. In her perception, more timid students felt more comfortable typing a well-thought out question to her than approaching her in class. A hybrid schedule also afforded the teacher and students the flexibility to work at their own pace for individualization.

These findings may serve as guidelines for teachers when implementing a blended learning approach in teaching, as they want their students to fully concentrate on the objective while also creating a community in which to use the target language with confidence. When teachers know how to keep students engaged and establish a social presence, teachers can provide a friendly and interesting atmosphere that enhances learning. In this study, it is not clear how social presence affects learning performance directly and indirectly. However, promoting effective communicative language learning

through discussion boards can lead to rich conversations in the target language creating a social presence among learners. Finding errors and revising the message before sending it allows learners to communicate with each other while accepting teacher feedback. The authenticity of the visuals, literary work, and song in both the F2F and online learning environments, gave students the opportunity to explore the richness of the language and complete critical thinking tasks. These blended implementations can serve as a way to enhance language learning in the targeted language.

#### **CHAPTER FIVE: DISCUSSION**

### Introduction

In the previous chapter, the presentation and analysis of data were detailed. This chapter consists of a summary of the study, discussion of the findings in comparison to the literature, implications for practice, recommendations for further research, and conclusions. The purpose of this chapter is to develop the concepts of this study to provide a further understanding of its possible impact on blended instruction in the community college ESL classroom, and present suggestions for additional research targeting how community college ESL instructors can foster social presence in a BL environment. Additionally, this chapter addresses challenges for community college faculty when designing and implementing the online and F2F portion of a course for language learning and for building a community.

## **Summary of Study**

This study sought to examine the design and implementation strategies of an experienced faculty member who teaches hybrid ESL courses at a Virginia community college, including how they perceived activities of blended learning and social presence in and outside the classroom to impact student success. The purpose of this study was achieved through pre- and post-interviews with the faculty member, rigorous document

analysis, and observations throughout the semester. The study included three research questions:

RQ1: What is the approach of an experienced community college English as a Second Language teacher to structuring a blended learning course to promote language learning?

RQ2: What are the teacher's perceptions of assignments that most effectively fostered social presence and promoted language learning in an English as a Second Language blended learning community college course?

RQ3: What are the teacher's perceptions of assignments that least effectively fostered social presence and promoted language learning in an English as a

Answers to the three research questions came from the findings from the interviews, document analysis from the LMS and F2F instruction, and in-class observations. The findings provided a detailed description of the instructor's experience with blended learning techniques in a collaborative setting to promote language learning.

Second Language blended learning community college course?

In regard to the teacher perceptions of a blended learning course, this study indicates that Linda promoted blended language learning. For research question one, the methods used for instruction demonstrated that the instructor implemented strategies that promote student engagement, enhance communication between all members of the learning community, and address course learning objectives in a flexible dual mode environment. The most frequently used instruction techniques included discussion forums, short stories, and advertisements. Based on observations, these methods

each other. The instructor used accountability measures with posting online and academic integrity, which supported student engagement and success. Interviews and observations demonstrated that the instructor regularly assessed the effectiveness of techniques and made adaptations when necessary. She considered student interests, focused on the achievement of learning objectives, and demonstrated willingness to connect with her students in a learning community. The findings from this study can be used as a resource for designing and implementing an ESL blended learning community college course.

### **Discussion of Findings**

Previous research depicts numerous benefits in support of a blended approach for language learning (Bauer-Ramazani et al., 2016; Hung, 2015; Kvashina & Martynko, 2016; Lee & Wallace 2017). The common themes in prior research related to positive outcomes in blended learning English language acquisition include flexibility in time and space, engagement in literature and activities in the dual modes, and communication among students and between students and the instructor. Addressing challenges for the dual mode approach to language learning, previous research (Webb et al., 2014; Han, 2015) identified instructor issues such as sustaining motivation, time required to prepare and collect class materials, and loss of personal connection during asynchronous class activities.

Additionally, the past research on social presence in the blended learning classroom revealed that students can feel less self-conscious and isolated while in the

online format in order to prepare them for F2F interactions (Wicks, Craft, Mason, Gritter, & Bolding, 2014). The work of Volchok (2018) asked students to introduce themselves and respond to their classmates' messages. The online forum fostered students' social presence and was a critical component of their overall grade. Volchok's study focused on blended learning in a community college classroom but did not highlight ESL students in particular. Further, the research provided scarce background knowledge on the thought process and perceived successful strategies on blended learning activities that impacted language learning and fostered social presence from an ESL instructor at the community college level.

For the purpose of my study, the Linda's interview responses provided insight into her thought process and design for the hybrid classroom at the community college, why she implemented specific activities for language learning, and her perception of her use of specific tasks to foster social presence. Document analysis and observations clarified instructional strategies Linda used in virtual and F2F settings and provided more in-depth information on what influenced her choice of the activities that she employed. Categories were identified from the interviews, documents, and observations, and within those, I identified strategies Linda used in the blended approach to language learning in the hybrid course to promote language learning and social presence. These strategies were: 1) organization of the face-to-face format; 2) activities with student participation and discussion in the LMS virtual space; and 3) student-to student and student-teacher communication in both modes. It is important to examine these categories in relation to

the literature on teaching in an ESL blended learning environment to gain a deeper understanding of how Linda's experience fits within a wider framework.

## Components in promoting language learning in blended learning research.

The first category identified through interviews, document analysis, and observations was organization of the F2F format. Students in the course, during F2F time, engaged in conversations about main ideas and symbolism in short stories, advertisements, and songs, which made them appear eager to learn and participate. These observations supported past research indicating that ESL students have enjoyed listening to songs and drawing meaning from them to share in discussion with the class (Şahin-Kızıl, 2014). The instructor communicated most tasks with students during F2F time using the target language, English, in order to accomplish the objective of integrating real-life scenarios and meaningful interactions (Egbert et al., 2015; Gleason, 2013). As students actively engaged in authentic tasks, with guidance from the instructor, they collectively and collaboratively built understanding (Egbert et al., 2015; Senior, 2010) and developed autonomy with the target language (Brinks-Lockwood, 2014; Han, 2015).

## F2F format.

In their study, Egbert et al. (2015) found online tasks that used engaging literature, appealed to student interests, and developed necessary skills for language learning. During F2F sessions, I similarly found that the literature used in discussion and student-led conversations engaged students by allowing them to work at their own pace, receive feedback, and help each other navigate assignments. As Clark and Rossiter

(2008) noted, stories engage learners at "more than a cognitive level; they engage our spirit, our imagination" because they draw on the learners' background knowledge. This strategy empowers students to share their ideas with confidence, giving everyone a voice without unnecessarily imposing particular opinions (Senior, 2010), which the instructor in the course for this study hoped to avoid.

During observations, the class was regularly involved in dissecting short stories together. The instructor would guide students to apply concepts and dissect the main idea in an open and welcoming F2F classroom setting. Research supports this strategy, as it shows that students are more responsive and involve themselves in learning tasks if they sense the teacher is invested in their education and success (Senior, 2010). Based on observations and interviews, Linda worked to establish rapport and trust without undermining her classroom management and expectations of her students in order to draw students into the learning. Additionally, Linda had found in the past that vocabulary exercises that were not intentionally tied to other activities and themes in the course were less successful. They did not foster student engagement and a social presence in the classroom for the students or for her. In her post-interview, Linda remarked that she would adapt the curriculum to attach vocabulary tasks in a more meaningful way to the short stories in the future in order to make them more authentic for students.

During the course of the semester, Linda incorporated visual materials to teach the concept of advertisement techniques, main idea, and symbolism. Many online courses at the community college level still consist heavily of text-based course materials and

lack auditory or visual stimuli (Jaggars et al., 2013), however, the hybrid course in this study did not use a textbook and allowed engagement with visuals in both online and F2F time. The students engaged in discussion and shared personal stories and interests while discussing the meaning and purpose of the advertisements.

### Learning management system.

The second category that promoted learning in the BL environment by the instructor was in the utilization of the LMS and online activities. Neumeier (2005) reported that giving learners academic choice and independence creates autonomy and responsibility for their learning (Šafranj, 2013; Stracke, 2007). In a blended ESL classroom, technology tools should enable the instructor to deliver a significant proportion of course content (more than 30%) through computer-based tasks, including online quizzes and synchronous-asynchronous discussion complementary to carefully planned, in-class language teaching (Allen, Seaman, & Garrett, 2007; Vernadakis, Antoniou, Giannousi, Zetou, & Kioumourtzoglou, 2011). The LMS used in this study aided in the transition from the physical to the virtual community because the instructor used Blackboard purposefully to prepare students for the course, displayed all course materials, and provided support materials for students who needed additional support. Having a flexible platform to display materials and communicate with the students is a way for teachers to individualize instruction and allow students to focus on skills outside of the classroom walls (Han, 2015). The students in this study participated in flexible and individualized learning opportunities which has the potential to reduce the feeling of

isolation during asynchronous activities. The open schedule allowed students to affirm and clarify their learning with concepts and tasks as many times as they needed to be successful. Blackboard was the LMS used for the virtual mode, however the instructor was also able to share insight into Canvas which was the new LMS the campus was switching to the following semester. The instructor's initial experience with Canvas indicated that some of the technology challenges from the observed course would be resolved in the new mode. For example, there is a calendar feature on Canvas that assists in making it clearer to students about directions and due dates for assignments.

One benefit in using the virtual space during blended learning is the opportunity to let students decide when they will engage in online activities. The intention of a blended learning course is that students are motivated to learn and engaged in the process, therefore they will learn more quickly and achieve language success (Radosavlevikj, 2015). Scott and Ryan (2009) in their study discovered that online members become more engaged in discussions and interacted effectively when they were set appropriate tasks. In addition, Batardiere (2015) researched learners' cognitive activity in an online forum with a collaborative task. He investigated the type of cognitive activity learners experienced while completing the task, and found high levels during intercultural online communication and critical reflection of different perspectives. The findings of my study similarly indicated that the online discussion forum enabled the participants to improve their critical thinking skills and linguistic and intercultural

competence. During the online portion in the course, discussion posts kept students interacting with one another and accountable in preparation for the next in-class session.

As documented in earlier research, discussion forums have long been used to facilitate, develop, and sustain social presence (Hostetter, 2013). Previous studies show that discussion boards, blogs and online collaboration tools help develop students' online communication literacies (Levy, 2017). In addition, a factor in successful engagement in F2F sessions during hybrid courses is the preparation that students do before class (Hsieh, et al., 2017; Hung, 2015). Research shows that instructors cannot fully engage students at a high level of academic achievement if proper work is not done ahead of time by the students (Webb, et al., 2014; Hsieh et al., 2017). Linda perceived visual activities as engaging to learners, both in the online and F2F environment. In observations, the interactions with materials elicited rich discussion and seemed to build confidence in students who engaged in sharing opinions about literature and various types of media. Asynchronous online discussions have gained considerable interest among educators and researchers in terms of designing and facilitating rich, authentic, and meaningful approaches to learning. A number of modes and online tools have been developed in order to analyze discussions by examining the relationships between interaction and learning (Hotstetter & Busch, 2013; Levy, 2017).

### Communication.

Student-to-student and student-teacher communication across both modes was found to be the third category in this study that contributed to language learning. Yang

(2014) proposes three dimensions of engagement during the online learning portion of a blended course: behavioral engagement, which can be observed as an action that leads to an outcome, emotional engagement which refers to students' sharing of feelings; and cognitive engagement which refers to learning strategies. Online learning engagement provides an interactive setting for communication among teachers and students in the classroom (Yuen, 2011) and this engagement can be observable and measured by the teacher to influence activity choices. Having students interact with each other and with the language itself also contributes to the sense of engagement in the learners (Meltzer & Hamman, 2004). Research shows that discussion within the social spaces provided in online communities has the potential to increase metacognition, as they allow students to construct meaning and confirm knowledge in the presence of peers. discussions (Akyol & Garrison, 2011). In addition, Chen and Chiou (2014) suggested that student success in the hybrid format as compared to the traditional format was due to convenient access to information, a learning style beneficial to the use of technology, and an enhanced motivation for learning due to their sense of belonging to a community.

My findings aligned with the work of Akyol and Garrison (2011) and Chen and Chiou (2016), as the flexible hybrid environment (Bauer-Ramazani et al., 2016; Futch et al., 2016; Hsieh et al., 2017) provided a virtual community and physical space for students to master language skills and practice independently, while still communicating with the teacher and other students. Similar to my findings, research notes the importance of asynchronous learning in giving a voice to students, as it builds their confidence by

allowing them to practice their communication skills in a comfortable and secure online forum (Bakar, Latiff, and Hamat, 2013; Futch et al., 2016). Albalawi (2014) found student-to-student interactions were perceived by students to be the most important for improving their satisfaction within the online learning environment. In my study, the discussion posts that elicited the most interaction between students were about the short stories. In particular, *Hills Like White Elephants* by Ernest Hemingway had the most posts. This story focuses on a couple having a conversation at a train station. The couple argue about an operation that the man wants the woman to have, which is implied to be an abortion. The teacher perceived that the students felt comfortable discussing the characters' dilemmas and agreeing or disagreeing respectfully about the decisions in the story.

Student-teacher interactions can also increase student engagement in language learning (Yang, 2014). The LMS, which was the major mode of communication in the course in my study, can facilitate the process of students reaching out and interacting with the teacher more effectively, which can make student-teacher communication more meaningful (Yang, 2014). Albalawi (2014) found learner-instructor interactions, along with the helpfulness and availability of the instructor, were valuable for improving the learner's satisfaction with online learning. The LMS in my study provided a way to enhance student-to-student and student-teacher interactions without the restrictions often experienced in a F2F classroom, in part due to the flexibility it provided. Linda provided her personal contact information to students in order to be available with questions on

weeknights and on weekends. Linda reported that many students took advantage of the flexible hours. Student-teacher communication in this study extended to virtual tutorials, F2F lectures, emails, one-on-one appointments, and phone calls throughout the course. Using these communication tools to provide more language input and feedback for students and accommodate the many proficiency levels allowed for individualized learning and increased learner autonomy and participation (Lee & Wallace, 2017). In the analysis of the data collected for my study, effective communication between the teacher and students improved learning outcomes for the students in the course. The instructor offered ongoing communication with students and encouraged students to communicate any concerns or questions during asynchronous and F2F time.

Along with student-to-student interaction, the instructor understood her feedback from teacher to student to be critical in helping students meet learning objectives. She felt that providing consistent and timely feedback was an important element of the communication during the course. The instructor also readily requested feedback on activities and projects from the students to enhance the quality of her course delivery methods. This practice is supported by findings from previous research, which showed that instructors recommended creating opportunities for students to provide feedback about the course to inform the instructor when an assignment was not working or a concept was not clear (Futch et al., 2016).

Research has shown that language students in a blended environment are less apprehensive about communicating with peers online due to the fact that they can

carefully consider and revise their replies before posting (Bakar, et al., 2013; Hsieh, et al., 2017; Levy, 2017). However, as explained in the previous chapter, despite the instructor's efforts and encouragement, there was apprehension on the part of students to reach out for assistance outside of class time.

E-learning and hybrid courses offer similar advantages for language learners. However, research in blended learning has revealed that learners can achieve a more significant sense of community or belonging than in a strictly online course (Garrison & Kanuka, 2004; Rovai & Jordan, 2004; Tayebinik & Puteh, 2013). McMillan and Chavis (1986) defined a sense of community as, "a feeling that members have of belonging, a feeling that members matter to one another and to the group, and a shared faith that members' needs will be met through their commitment to be together" (p. 9). The hybrid approach can establish a community of inquiry that allows learners to interact and collaborate with their peers and to learn in "a learning environment that integrates social, cognitive and teaching elements that in a way will precipitate and sustain reflection and discourse" (Garrison & Vaughan, 2008, p. 8). A community of learners can interact from various places at different times because of the benefits that CALL tools provide in order to learn from each other (Šafranj, 2013).

The instructor in my study used the virtual environment to provide an online support system where students could construct meaning and confirm knowledge in the presence of peers on the discussion board format, which aligns with previous research (Akyol, Garrison, & Ozden, 2009). Linda perceived the students to be a friendly, close

community in both the F2F and online modes. Many of the students knew each other from previous courses and she noticed a comfort in the way they turned to each other for clarification and help with assignments.

## Social presence.

Tinto (1993) suggested that the college classroom was the space where the social and the academic met and that questions about student achievement and success should begin with the influence of social presence and activities in the classroom. Liu et al. (2009) found that social presence is a significant predictor of course retention and final grade in the community college online environment. Social presence can help learners to project themselves in online discussion boards and feel a sense of belonging to the online community. Garrison (2009) suggested that social presence occurs when learners are able to become a part of a community, interact within that community, and develop relationships by revealing their personalities. Although Linda did not specifically use the term "social presence," her approach to teaching and course structure fostered aspects of social presence among the community of learners. Linda's interviews also relayed information regarding components of the course that led to a strong classroom community and some areas that did not contribute to language learning and the learning environment.

I observed all three components of social presence — open communication, group cohesion, and affective expression — during in-class sessions. Regarding open communication, the learning environment in the virtual and F2F space was observed as a

place where students expressed their opinions and regularly took part in activities and discussions. The interpersonal interaction seemed to elicit more dialogue than whole group conversations. Group cohesion, although not established outside of the classroom intentionally, was achieved during F2F time when learners collaborated with each other to achieve objectives regarding short stories or advertisements. Some students who did not participate regularly in whole group discussions were able to have extensive conversations with peers about the same topics. Providing time and opportunities for students to interact and collaborate with each other can result in a significant difference in group cohesion in an online course (Lee & Huang, 2018). Linda consistently organized students to work in small groups during F2F time in order to discuss short stories and homework assignments done outside of class. Students could use their discussions to help clarify topics for their rewrites on posts or for guidance on the next assignment.

According to Chang and Sperling (2014), students online can reinforce friendship and affinity with peers while helping each other with academic topics from the classroom. Having that support system can lead students to participate and engage more regularly. Kucuk and Sahin (2013) found that group cohesion was significantly higher in a blended mode when compared to solely online. Lastly, affective expression is related to reflecting emotions. Linda observed clear moments of togetherness and friendship, as well as self-disclosure and humor, during the online discussion posts. The students were able to get to know each other, interact frequently, and receive feedback and support when needed from each other and the instructor.

In their study of an online course, Kilgore and Lowenthal (2015) found that social presence could be established if "faculty and instructional designers take the time to design intentional learning experiences that help establish and maintain social presence using a variety of technologies while they also actively engage learners and model effective ways to communicate throughout an entire course" (p. 382). In a blended learning language class, participation can be associated with student success, while minimal online interactions, both passive and active, can be associated with lower academic achievement (Rubio, Thomas, & Li, 2018).

During my study, Linda wanted opportunities to communicate with students in the online environment that would match closely with the frequency of the F2F space. The course was designed with supports for student-to-student, student-teacher, and student-to-content interaction. This was achieved through sharing of personal stories and experiences, immediate and consistent feedback, and regular discourse in the LMS and F2F settings. At the beginning of the semester, Linda started creating her classroom community by providing opportunities for students to make connections through introductions and responses, on the discussion board. With these frequent posts, Linda also communicated expectations, reminders, and routines for posting, emailing, and corresponding in the classroom. As the instructor, she found it imperative to model for students how to communicate on the discussion board and replied to each student to maintain a connection. Linda made several posts prior to the first class of the course that showed how to properly communicate with another member of the class in the forum. She continued to model her guidelines in her comments back to students on their own

posts. In prior research, modeling and monitoring participation can allow faculty to adjust their courses to encourage more participation and could also help faculty to respond appropriately when students are struggling (Rubio et al., 2018). The implications from this discussion are discussed further in the next section.

### **Implications for Practice and Policy**

The increasing need for language teachers to create and implement instruction that will meet the needs of learners from diverse sociolinguistic backgrounds has been highlighted in the research (Hampel & Stickler, 2005; Hubbard, 2008; Liu & Kleinsasser, 2015). Blended learning with intentional fostering of social presence for language instruction has been identified as a way to reach those learners. It is important to identify instructor perceptions about the work of designing and implementing techniques used and to understand how they choose which techniques to utilize. An additional concept to address is the potential challenges that can hinder a BL environment from being successful and how instructors navigate through those obstacles to promote learning. The findings of this study have several implications for ESL faculty teaching hybrid courses with the goal of developing a classroom community and establishing social presence among their students. This study identified several promising strategies in the implementation of activities and methods as well as areas in need of improvement. The study can serve as an informative guide for community college faculty when making decisions about using materials and developing activities for both F2F and online modes. Novice and experienced faculty can use the challenges and solutions found in this study to reflect on their teaching and hybrid courses.

An implication of this study involves motivation to consistently adapt materials to overcome challenges and bring out the potential benefits of blended learning. Using engaging materials can result in motivation from the students to actively participate in their learning. Adjuncts and teachers new to hybrid courses need access to information on strategies and materials that apply to ESL courses they teach in order to identify new avenues to promote language learning. The instructor in this study indicated a willingness to incorporate new techniques to avoid barriers to student success and devoted time outside of the F2F environment to establish a connection with her students. However, the instructor also identified challenges, including technology skill level of the student, plagiarism, and student ability to interpret directions. In some cases, these are barriers to implementing new techniques. Making information on design and implementation strategies readily available to new and adjunct faculty of hybrid courses would minimize the time necessary to locate materials and aid in avoiding the barriers highlighted in the study. Community colleges could use an LMS, such as Blackboard or Canvas, to set up discussion forums for instructors to share specific methods and materials useful for ESL hybrid courses.

Another implication of this study involves the limited resources and professional development available to instructors for the establishment of a community of inquiry with social presence in the classroom. For the ESL hybrid instructors, discovering how to implement activities with social presence can create an environment where there is trust and information is successfully shared (Garrison et al., 2000). However, Linda, an experienced instructor, did not explicitly design online or F2F assignments with

collaboration and social presence specifically in mind. Additionally, she stated that group work was difficult to implement due to a lack of participation from some members of the group and schedule conflicts in meeting outside of class. She also felt that asking students to meet outside of the F2F environment could impact the flexibility of a hybrid course. To address these concerns, students could meet online using tools such as Google chat and Blackboard Collaborate and establish roles for accountability of equal contributions.

Finally, the findings in this study indicate the importance of: 1) researching specific activities that promote language learning and social presence in ESL hybrid courses; and 2) evaluating these resources and their connection to student perceptions of belonging and student success. Faculty mentor meetings and professional development opportunities that address instruction and implementation techniques to promote social presence and student engagement would provide information on each instructor's understanding of the concept and increase the likelihood that they might focus on these elements in their classes. This is especially important for community college faculty who may not use a textbook and are responsible for developing and finding their own materials. For NVCC, regular conferences and conventions on social presence in ESL blended learning could also ensure motivation for use on the part of the faculty. If professional development and research include consideration of instructional strategies with a social presence, faculty could implement and provide feedback and suggestions for improvement. Constant adaptations and growth in teaching would contribute to a strong methodology for English language learning across the NVCC campuses. Since the instructor in this study indicated that she consistently sought out ways to improve her

course to promote student learning, professional development that incorporates BL strategies for community building and engagement could be helpful to instructors and students in meeting the learning objectives of the course.

#### Limitations

This study was limited to one campus location at NVCC located in a culturally and socioeconomically diverse portion of Northern Virginia, so its findings may not be widely transferable to other campuses whose members are less diverse, come from more privileged settings, or may have greater access to technology outside of the classroom. While the researcher's observational experience with the institution consisted of online and in-class textbooks and drill exercises in ESL classrooms, the institution consistently ranks among the top colleges nationally for using technology to provide exceptional services to students and increase educator effectiveness. The study also limited its focus to one construct of the CoI framework: the perceived social presence and the impact on learning outcomes of an ESL blended course. There are additional factors influencing teacher perceptions of language learning, so this study is a preliminary, limited look at social presence in a specific learning situation.

Lastly, this research approach employed a single-case study which is susceptible to critique among researchers in the areas of external validity and generalizability. As King, Keohane, and Verba (1994) write, "in all social science research and all prediction, it is important that we be as explicit as possible about the degree of uncertainty that accompanies our prediction" (p. 212). The exploratory approach, rather than an explanatory one, aids in minimizing this limitation. For generalizability, the ESL program

demographics and economic make-up between counties of the institution are not drastically different from one another if the research were to inform instruction on other campuses of the institution. I am, however, cautious of overgeneralizing the study's findings.

### **Future Research**

Future research into this subject should include a representative sample of ESL faculty at campuses who are actively selecting course material with a social presence framework and collaborative environment in mind. Using this work as a foundation, a study of a larger representative sample of ESL hybrid faculty could identify similarities and differences across institutions. A study that compares faculty knowledge and use of activities geared towards language learning and social presence in a BL format could identify positive instructional strategies that promote student success, as well as common issues from instructors that hinder the success of language learning. These studies might include student perceptions and compare them to instructor perceptions and student achievement. Additional research in perceptions could examine the difference between perceived and observed instructional strategies. In addition, an examination of end-of-semester student achievement and student perceptions of engagement could capture how students react to activities in order to constantly improve language instruction in the blended environment.

Additional research needs to include the factors that motivate students to take advantage of instructor availability during asynchronous times in order to clarify concepts or practice their language skills beyond course requirements. Because the

observations in this study were limited to one classroom, strategies and activities observed may be more related to the particular course institution that were the focus of my study. A more thorough examination of teacher perception compared to observation in multiple settings would provide additional data on what instructors do in their BL ESL classrooms over the course of a semester.

Another area of research could focus on providing more detailed information on the resources ESL community college instructors use when preparing for BL courses and F2F class sessions. Due to the success that the students had with the in-class discussion and individual assignments, more research needs to be done on the use of visual aids and advertisements to bring students together in a community of inquiry. Little research has been done on using advertisements to promote language learning. There is currently no research on advertisements and the impact on social presence in the community college language classroom. Such research would provide data on where instructors find instructional materials that they perceive to engage learners and themselves. The instructor in this study frequently used articles posted on social media and topics which interested her as assigned readings. Research that includes questions more specific to the value the instructor places on different resources for learning language and facilitating rich discussion could identify effective mechanisms for professional development of ESL hybrid instructors at the community college level. In terms of the data collection instruments, future researchers should also consider including interview questions from the COI survey that break down the categories of social presence (open communication, group cohesion, affective expression). Specifically, more research needs to be done on

blended language courses with purposeful activities geared toward group cohesion both in the online and F2F environment. Future research into how ESL hybrid instructors navigate collaboration in blended learning would help instructors promote community-building among students outside of the classroom. This would provide a more detailed account of the teacher's perception of social presence in their course.

Lastly, qualitative data on less effective techniques identified by students could be compared with the perceptions of the instructor, such as disconnected vocabulary tasks or a lack of collaborative projects. Additional methods where vocabulary can be used as an engaging community building tool without seeming rote could be included in other studies. This would provide useful information for understanding both teaching and learning ESL in community colleges. Specifically, researchers could gather information on why students found a technique less effective, the steps for making this determination, and how their preference for task implementation could address challenges and create an easier process for instructors designing future courses. Studies could also include targeting those students where misconceptions in written language and following directions can be a challenge. This would provide additional information on strengthening language instruction based on students' needs.

### Conclusion

The findings of this study expanded the work of previous researchers in the area of blended learning and social presence in an ESL community college course and instructor perception. This investigation revealed that an experienced instructor of hybrid courses at NVCC campus focused on student learning objectives and planned courses to

utilize techniques that promote student engagement, included relevant connections and references, supported discourse and communication, and included assessments that encouraged students to demonstrate their language learning. As they do throughout higher education and specifically in community colleges, instructors play an important role in providing language teaching in an evolving field. The instructor in this study indicated an understanding of that role, a dedication to constant improvement, and a dedication to developing a welcoming classroom community built on open communication.

Additionally, this study showed the instructor utilizing a variety of resources that they collected and curated. The materials gathered required a large amount of time to prepare for students prior to the start of the course. Due to complications with academic integrity, the instructor regularly revised the course assignments and assessments to hold students accountable for their learning. During these revisions, the instructor was also advising adjunct faculty, teaching courses, and learning a new LMS for the next course.

While this study was not able to capture student perceptions of blended learning and social presence, the data presented here indicated student engagement in the course. The instructor reported that only two students in the class did not pass the course and that the student end-of-semester questionnaires were overwhelmingly positive. A more intentional approach to capture student satisfaction in a blended learning course would be beneficial in future research. The instructor worked to expand knowledge in the area of reading in an ESL course. This included efforts to meet student learning objectives while keeping content authentic for the learner, creating a dynamic classroom environment for

students in dual modes, maintaining their own passion for the subject matter, and constantly growing as an education professional in order to meet student needs. It is my hope that community colleges can use my study to serve dedicated teachers and ESL learners in technology and language education.

#### **APPENDIX A**

### **Social Presence Faculty Pre-Semester Interview Questions**

- 1. How do you convey to students how to have discussions online and face-to-face? Do you have specific conversations about agreeing or disagreeing, sensitivity to culture, etc?
- 2. How comfortable are you facilitating content discussions? In content discussions online, what is your expectation from peer-to-peer on the discussion boards?
- 3. What is your process for making sure collaboration and meaningful interactions happen outside the traditional classroom with your students? Inside?
- 4. What are your measures for holding students accountable while still keeping the learning meaningful?
- 5. What are some things that you feel you are successful at in creating a community of learning among your students online?
- 6. Describe how you provide positive feedback, both online and F2F. Constructive feedback for improvement? Group? Individual?
- 7. Are there any specific ways in which you help students get to know each other?
- 8. Do you find the online communication of your course to be useful for social interaction? In class?
- 9. How do you encourage quiet or shy students to engage in discussion with peers and yourself in online settings? F2F settings?
- 10. How do you promote collaboration online? F2F?
- 11. How do you format online discussions to make students see different perspectives?
- 12. What specific projects have you prepared that you feel will work best to foster social presence? Why?

- 13. Is collaboration among students important to you? Why?
- 14. How do you foster a feeling of trust from student-to-student and student-to-teacher?
- 15. Describe challenges of transferring social presence from F2F to online.
- 16. What strategies are you currently employing to increase social presence?
- 17. What strategies are you currently employing to build a learning community?
- 18. What strategies are you currently employing to reduce feelings of isolation in students?
- 19. What role do you think the synchronous tools will play to achieve these aims?

#### **APPENDIX B**

## Community of Inquiry Survey Instrument (draft v14)

## **Teaching Presence**

## Design & Organization

- 1. The instructor clearly communicated important course topics.
- 2. The instructor clearly communicated important course goals.
- 3. The instructor provided clear instructions on how to participate in course learning activities.
- 4. The instructor clearly communicated important due dates/time frames for learning activities.

#### **Facilitation**

- 5. The instructor was helpful in identifying areas of agreement and disagreement on course topics that helped me to learn.
- 6. The instructor was helpful in guiding the class towards understanding course topics in a way that helped me clarify my thinking.
- 7. The instructor helped to keep course participants engaged and participating in productive dialogue.
- 8. The instructor helped keep the course participants on task in a way that helped me to learn.
- 9. The instructor encouraged course participants to explore new concepts in this course.
- 10. Instructor actions reinforced the development of a sense of community among course participants.

## Direct Instruction

- 11. The instructor helped to focus discussion on relevant issues in a way that helped me to learn.
- 12. The instructor provided feedback that helped me understand my strengths and weaknesses relative to the course's goals and objectives.

13. The instructor provided feedback in a timely fashion.

### Social Presence

### Affective expression

- 14. Getting to know other course participants gave me a sense of belonging in the course.
- 15. I was able to form distinct impressions of some course participants.
- 16. Online or web-based communication is an excellent medium for social interaction.

## Open communication

- 17. I felt comfortable conversing through the online medium.
- 18. I felt comfortable participating in the course discussions.
- 19. I felt comfortable interacting with other course participants.

## Group cohesion

- 20. I felt comfortable disagreeing with other course participants while still maintaining a sense of trust.
- 21. I felt that my point of view was acknowledged by other course participants.
- 22. Online discussions help me to develop a sense of collaboration.

## Cognitive Presence

## Triggering event

- 23. Problems posed increased my interest in course issues.
- 24. Course activities piqued my curiosity.
- 25. I felt motivated to explore content related questions.

## **Exploration**

- 26. I utilized a variety of information sources to explore problems posed in this course.
- 27. Brainstorming and finding relevant information helped me resolve content related questions.

28. Online discussions were valuable in helping me appreciate different perspectives.

## Integration

- 29. Combining new information helped me answer questions raised in course activities.
- 30. Learning activities helped me construct explanations/solutions.
- 31. Reflection on course content and discussions helped me understand fundamental concepts in this class.

#### Resolution

- 32. I can describe ways to test and apply the knowledge created in this course.
- 33. I have developed solutions to course problems that can be applied in practice.
- 34. I can apply the knowledge created in this course to my work or other non-class related activities.

# 5-point Likert-type scale

1 = strongly disagree, 2 = disagree, 3 = neutral, 4 = agree, 5 = strongly agree

### APPENDIX C

## **Social Presence Faculty Post-Semester Interview Questions**

- 1. Were there any problems that students had with the technology tools, LMS, or activities that you did not expect? Language issues?
- 2. What will you change or adapt for the next ESL hybrid course?
- 3. Do you think students were comfortable having conversations with each other online and/or f2f? Were there any disagreements? If so, how were they handled, both by the students and yourself?
- 4. What do you think went well with collaboration (group work, partner) this semester, both online and F2F? Why? Was there any aspect of collaboration that did not go as planned?
- 5. Do you think your accountability measures were successful?
- 6. Do you feel these students formed a strong community of inquiry? Why or Why not?
- 7. How do you feel that your feedback helped students this semester? What about peer feedback?
- 8. How was the social climate of our f2f and online classroom? What do you feel contributed to that?
- 9. Do you feel that some students became more socially and academically open as the course developed? What do you think contributed to that?
- 10. How did group work go in your classroom? Is there anything that you would do differently for next semester?
- 11. What is your overall impression of class perception towards BL? Collaboration?
- 12. What course activities do you think the students enjoyed most? Least? Why do you think that?

- 13. What specific projects worked best to foster social presence, both online and F2F? Why?
- 14. Was your online discussion format successful in getting students to share their opinions about topics? How did you monitor that?
- 15. Describe any challenges of transferring social presence from F2F to online.

# APPENDIX D

RUBRIC: Your project will be graded as follows. Use this as a checklist to make sure everything required has been included in your project.

SECTION 1:
a. Introduces the title and author, spelled and capitalized correctly(5
points)
<b>b.</b> Explains how advertising creates emotions in consumers (5 points)
SECTIONS 2-6:
a. includes a separate page/slide for each of the five concepts you choose
(1 point for each slice; total 5 points)
b. includes a definition or explanation of each concept, clearly explaining what
the concept means (5 points for each slide; total 25 points)
c. Includes a picture on each slide chosen from our list and explains how the
picture you chose supports the concept on the slide (5 points for each
slide; total 25 points)
SECTION 7:
a. Explains which technique you think is the most effective and why (5
points)
VOCABULARY:

a. Ten vocabulary words are highlighted or underlined within the slides of the
presentation (5 points)
b. Ten vocabulary words are used correctly within the presentation (20
points – 2 each)
c. A numbered list of ten vocabulary words used in the presentation is on the
last slide of the presentation (5 points)

# **TOTAL POINTS: 100**

- Please spell-check your project. If it is not spell-checked, you will lose 10 points off your final grade.
- DUE BY EMAIL on or before March 19th at 12:30 <u>before class time</u> (12:30).

  Late assignments will be marked down 10 points every DAY they are late after 12:30 on March 19th.

#### REFERENCES

- Abdillah, L.A. (2016). Exploring student's blended learning through social media.

  Computer, Mathematics and Engineering Applications, 7, 245-254.
- Abedin, B., Daneshgar, F., & D'Ambra, J. (2010). Underlying factors of sense of community in asynchronous computer supported collaborative learning environments. *Journal of Online Learning and Teaching*, 6(3), 585-597.
- Aimin, L. (2013). The study of second language acquisition under socio-cultural theory.

  \*American Journal of Educational Research 1(5), 162-167.
- Akyol, Z., & Garrison, D. (2011). Understanding cognitive presence in an online and blended community of inquiry: Assessing outcomes and processes for deep approaches to learning. *British Journal of Educational Technology*, 42(2), 233-250. doi:10.1111/j.1467-8535.2009.01029.x
- Akyol, Z., Garrison, D. R., & Ozden, M. Y. (2009). Online and blended communities of inquiry: Exploring the developmental and perceptional differences. *The International Review of Research in Open and Distance Learning*, 10(6), 65–83.
- Albalawi, B.R. (2014). Student attitudes towards intensive English program at CSCED and their effects on academic achievement. *International Journal for Innovation*, *Education*, *and Research*, 10(4), 139-157.
- Al-Fudail, M. & Mellar, H. (2008). Investigating teacher stress when using

- technology. Journal of Computer and Science, 51, 1103-1110.
- Al-Jarf, R. (2004). The effects of web-based learning on struggling EFL college writers.

  King Saud: King Saudi University. *Foreign Language Annual*, *37*(1), 46-45.
- Allen, I. E., Seaman, J., and Garrett, R. (2007). *Blending in: The extent and promise of blended education in the united states*. Needham, MA: The Sloan Consortium.
- Al-Zahrani, A. M. (2015). From passive to active: The Impact of the flipped classroom through social learning platforms on higher education students' creative thinking. British Journal of Educational Technology, 46(6), 1133-1148.
- Amaral, K. E., & Shank, J. D. (2010). Enhancing student learning and retention with blended learning class guides. *Educause Quarterly*, 33(4).
- Ananiadou, K. & Claro, M. (2009). 21st century skills and competences for new millennium learners in OECD countries. *OECD Education Working Papers*, 41. Paris: OECD Publishing.
- Anderson, T., Rourke. L., Garrison, R., Archer, W. (2001). Assessing teaching presence in a computer conferencing context. *Journal of Asynchronous Learning Networks*, 5(2), 1-17.
- Aragon, S. R. & Johnson, E. S. (2008). Factors influencing completion and noncompletion of community college online courses. *The American Journal of Distance Education*, 22(1), 146-158.
- Arbaugh, J. B., Cleveland-Innes, M., Diaz, S. R., Garrison, D. R., Ice, P., Richardson, J. C., & Swan, K. P. (2008). *Developing a community of inquiry instrument: Testing a measure of the Community of Inquiry framework using a multi-institutional*

- sample. Internet and Higher Education, 11(3-4), 133-136.
- Armellini, A., & De Stefani, M. (2016). Social presence in the 21st century: An adjustment to the Community of Inquiry framework. *British Journal of Educational Technology*, 47(6), 1202–1216. doi:10.1111/bjet.12302.
- Aschengrau A. & Seage, G.R. (2013). *Essentials of Epidemiology in Public Health*. Sudbury, MA: Jones and Bartlett.
- Asri Humaira, M., Rasmitadila, R., Widyasari D., & Rachmadtullah, R. (2019). Using experiential learning model (ELM) to slow learner students in the science lesson *Journal of Physics Conference Series*, 1175.
- Aurangzeb, W. (2018). Blended learning classroom environment at university level: A panoramic view of students' perceptions. *Journal of Critical Inquiry*, 16(1), 96-113.
- Baepler, P., Walker, J.D., & Driessen, M. (2014). It's not about seat time:

  Blending, flipping, and efficiency in active learning classrooms. *Computers & Education*, 78, 227–236.
- Bailey, J., Ellis, S., Schneider, C., & Ark, T. V. (2013). Blended learning implementation guide. Retrieved from <a href="http://net.educause.edu/ir/library/pdf/CSD6190.pdf">http://net.educause.edu/ir/library/pdf/CSD6190.pdf</a>
- Bakar, N. A., Latiff, H., & Hamat, A. (2013). Enhancing ESL learners speaking skills through asynchronous online discussion forum. *Asian Social Science*, *9*(9), 224–234. doi:10.5539/ass.v9n9p224
- Baker, J.D. (2004). An investigation of relationships among instructor immediacy and affective and cognitive learning in the online classroom. *The Internet and Higher*

- *Education*, 7, 1-13.
- Bangert, A. (2008). The influence of social presence and teaching presence on the quality of online critical inquiry. *Journal of Computing in Higher Education*, 20(1), 34-61. doi: 10.1007/BF03033431
- Barker, J. (2015). Benefits of hybrid classes in community colleges. *Contemporary Issues In Education Research*, 8(3), 143-146. Retrieved from

  <a href="https://search-proquest">https://search-proquest</a>
  com.mutex.gmu.edu/docview/1696895686?accountid=14541
- Barr, S., Eslami, Z.R., Joshi, M.R., Slattery P., & Hammer, J. (2016). Perceptions and knowledge of literacy constructs by elementary school teachers of ELLs. *The Educational Forum*, 80(1), 61-78, doi: 10.1080/00131725.2015.1102364
- Barr, D., Leakey, J., & Ranchoux, A. (2005). Told like it is! An evaluation of an integrated oral development pilot project. *Language Learning and Technology*, 9(3), 55–78. Retrieved from http://llt.msu.edu/vol9num3/barr/
- Batardiere, M. T. (2015). Promoting critical thinking in online intercultural communication. *The Eurocall Review*, 23(1), 3-9.
- Bauer-Ramazani, C., Graney, J. M., Marshall, H. W., & Sabieh, C. (2016). Flipped learning in TESOL: Definitions, approaches, and implementation. *TESOL Journal*, 7(2), 429–437. https://doi.org/10.1002/tesj.250.
- Beatty, K. (2003). *Teaching and Researching Computer-Assisted Language Learning*.

  London: Pearson Education Limited.
- Becerra, D. (2012). Perceptions of educational barriers affecting the academic

- achievement of Latino K-12 students. *Children & Schools*, 34(3), 167-177. doi:10.1093/cs/cds001
- Behroozizad, S., Nambiar, R. M., & Amir, Z. (2014). Sociocultural theory as an approach to aid EFL Learners. *Reading*, *14*(2).
- Bender, W.N. (2012). Differentiating instruction for students with learning disabilities:

  New best practices for general and special educators (3rd Edition). Thousand
  Oaks, CA: Crowin.
- Bender, T. (2003). Discussion Based Online Teaching to Enhance Student Learning.

  Virginia, Stylus Publishing LLC.
- Benson, V., Anderson, D., & Ooms, A. (2011). Educators' perceptions, attitudes and practices: blended learning in business and management education. *Research in Language Technology*, 19(2), 143-154.
- Bergmann, J. & Sams, A. (2012). Flip your classroom: Reach every student in every class every day. Washington, DC: International Society for Technology in Education.
- Bersin, J. (2004). The blended learning book: Best practices, proven methodologies, and lessons learned. L. Shannon (Ed.). San Francisco, CA: Pfeiffer.
- Bifuh-Ambe, E. Literacy skills acquisition and use: A study of an English language learner in a U.S. university context. (2009). *Adult Basic Education and Literacy Journal*, *3*(1), 24–33.
- Biocca, F., Harms, C. & Burgoon, J. K. (2003). Toward a more robust theory and measure of social presence: review and suggested criteria. *Presence*, 12(5), 456–

480.

- Blake, R. J. (2009). The use of technology for second language distance learning. *The Modern Language Journal*, *93*, 822–835. Retrieved from http://www.jstor.org/stable/25612277
- Blattner, G., and Lomicka, L. (2012). A sociolinguistic study of practices in different social forums in an intermediate French class. *International Journal of Instructional Technology and Distance Learning*, 9(9), 3-24.
- Bleffert-Schmidt, A. (2011). The Blended learning experience of community college students. (Doctoral Dissertation). Nova Southeastern University. Retrieved from NSUWorks. Retrieved from http://nsuworks.nova.edu/gscis\_etd
- Block, D. (2003). *The social turn in second language acquisition*. Edinburgh: Edinburgh University Press.
- Bloomberg, L. D., & Volpe, M. F. (2015). *Completing your qualitative dissertation: A road map from beginning to end* (2nd ed.). Thousand Oaks, CA: Sage.
- Bloor, M. and Wood, F. (2006) Keywords in Qualitative Methods. London: Sage.
- Bluic, A. G. (2007). Research focus and methodological choices in studies into students' experiences of blended learning in higher education. *The Internet and Higher Education*, 10, 31-244.
- Borup, J., West, R., & Graham, C. (2012). Improving online social presence through asynchronous video. The Internet and Higher Education, 15(3), 195-203. doi: 10.1016/j.iheduc.2011.11.001
- Bowen, A. G. (2009). Document analysis as a qualitative research method. *Qualitative*

- *Research Journal*, 9(2), 27-40.
- Bowen, W. G., Nygren, T. I., Lack, K. A., & Chingos, M. M. (2013). Online learning in higher education: Randomized trial compares hybrid learning to traditional course. *Education Next*, *13*, 58-64.
- Brantmeier, C. (2003). Beyond linguistic knowledge: Individual differences in second language reading. *Foreign Language Annals*, 36, 33–43.
- Brazington, A. (2012). Letting go. Campus Technology, 17-20.
- Brinks-Lockwood, R. (2014). Flip it! Strategies for the ESL classroom. Ann Arbor, MI: University of Michigan Press.
- Brock, T., Jenkins, D., Todd, E., Miller, J., Gooden, S., Martin, K., MacGregor, C., & Pih, M. (2007). Building a culture of evidence for community college student success: Early progress in the Achieving the Dream initiative. New York, NY: MDRC and Community College Research Center, Teachers College, Columbia University.
- Bruce, M. A., Young, S., & Kennedy, A. (2012). *Discovering student and*instructor perceptions of trust-building in online courses. Paper presented at the

  30th Annual Conference of the Northern Rocky Mountain Educational Research

  Association, Park City, UT.
- Bueno-Alastuey, M. C. (2009). WebCT design and users' perceptions in English for agriculture. In R. V. Marriott & P. L. Torres (Eds.), *Handbook of research on e-learning methodologies for language acquisition* (pp. 480–496). New York/London: Information Science Reference.

- Calcagno, J., Bailey, T., Jenkins, D., Kienzl, G., & Leinbach, T. (2008). Community college student success: What institutional characteristics make a difference? *Economics of Education Review*, 27(6), 632–645.
- Campbell, N. (2007). Bringing ESL students out of their shells: Enhancing participation through online discussion. *Business Communication Quarterly*, 70(1), 37-43.
- Canagarajah, A. S. (2013). Negotiating translingual literacy: An enactment. *Research in the Teaching of English*, 48(1), 40-6.
- Center for Digital Education. (2015). *Digital Community Colleges Survey*. Retrieved from <a href="http://www.govtech.com/education/higher-ed/2014-2015-Digital-">http://www.govtech.com/education/higher-ed/2014-2015-Digital-</a>
  Community-Colleges-Survey.html
- Century Foundation Press. (2013). Bridging the higher education divide: Strengthening community colleges and restoring the american dream: The report of the century foundation task force on preventing community colleges from becoming separate and unequal, 12.
- Chapelle, C. A. (2009). The relationship between second language acquisition theory and computer-assisted language learning, *The Modern Language Journal*, *93*, 741–753. Retrieved from <a href="http://www.jstor.org/stable/25612271">http://www.jstor.org/stable/25612271</a>
- Chapelle, C. A. (2001). Computer applications in second language acquisition:

  Foundations for teaching, testing and research. Cambridge: Cambridge University

  Press.
- Chapin, H. (1974). Cat's in the Cradle. On *Verities & Balderdash*. Elektra.
- Chang, Y., & Sperling, M. (2014). Discourse and identity among ESL learners: A case

- study of community college ESL classroom. *Research in the Teaching of English*, 49(1), 31–51.
- Chen, L. L. (2016). Impacts of flipped classroom in high school health education. *Journal of Educational Technology Systems*, 44(4), 411–420.
- Chen, B. H., & Chiou, H. H. (2014). Learning style, sense of community and learning effectiveness in hybrid learning environment. *Interactive Learning Environments*, 22(4), 485–496. doi: 10.1080/10494820.2012.680971
- Chen, C.C., & Jones, K.T. (2007). Blended learning vs. traditional classroom settings:

  Assessing effectiveness and student perceptions in an MBA accounting course.

  The Journal of Educators Online, 4(1), 1-15.
- Chen, Y., Wang, Y., Kinshuk, & Chen, N.S. (2014). Is FLIP enough? Or should we use the FLIPPED model instead? *Computers & Education*, 79, 16–27.
- Chenoweth, N. A., Ushida, E., Murday, K., & Ushtoa, E. (2006). Student learning in hybrid French and Spanish courses: An overview of language online. *CALICO Journal*, 24(1), 115–146. Retrieved from http://www.jstor.org/stable/24156297
- Chiasson, K., Terras, K., & Smart, K. (2015). Faculty perceptions of moving a face-to-face course to online instruction. Journal of College Teaching & Learning, 12(3), 231-240.
- Christo-Baker, E. (2004). Distance education leadership in higher education institutions: explored within theoretical frameworks of organizational change and diffusion of innovations theory. In L. Cantoni & C. McLoughlin (Eds), *Proceedings of world*

- conference on educational multimedia, hypermedia and telecommunications 2004 (pp. 251–256). Chesapeake, VA: AACE.
- Clark, R.C. & Lyons, C. (2004). Graphics for learning: Proven guidelines for planning, designing, and evaluating visuals in training materials. San Francisco, CA:

  Pfieffer.
- Clark, M. C., & Rossiter, M. (2008). Narrative learning in adulthood. *New Directions for Adult and Continuing Education*, 119, 61-70.
- Comas-Quinn, A. (2011). Learning to teach online or learning to become an online teacher: An exploration of teachers' experiences in a blended learning course. *ReCALL: The Journal of EUROCALL*, 23(3), 218-232. doi:10.1017/S0958344011000152.
- Community College Consortium for Immigrant Education. (2015). *The U.S. Immigrant Population: Demographics, Education, Labor Force, and the Economy.*
- Community college trends and statistics (2015). Retrieved from http://www.aacc.nche.edu
- Complete College America (2016). Complete College America Common College

  Completion Metrics Technical Guide. Retrieved from www.completecollege.org
- Compton, K.L. (2009). Preparing language teachers to teach language online: A look at skills, roles, and responsibilities, *Computer Assisted Language Learning*, 22(1), 73-99, doi: 10.1080/09588220802613831
- Coryell, J. & Chlup, D.T. (2007). Implementing e-learning components with adult english language learners: Vital factors and lessons learned. Computer Assisted

- Language Learning, 20(3), 263-278, doi: 10.1080/09588220701489333
- Crawford, C., Barker, J., & Seyam, A. (2014). The promising role of hybrid learning in community colleges: Looking towards the future. *Contemporary Issues in Education Research*, 7(3), 237-242.
- Creswell, J. W. (2014). Research design: Qualitative, quantitative, and mixed methods approaches. (5th ed). Thousand Oaks, CA: Sage Publications.
- Creswell, J.W. (2007). Qualitative inquiry and research design: Choosing among five approaches (2nd ed.). Thousand Oaks, CA: Sage.
- Creswell, J. (2015). 30 essential skills for the qualitative researcher. Los Angeles, CA:

  Sage.
- Crookston A. and Hooks G. (2012). Community colleges, budget cuts, and jobs: The impact of community colleges on employment growth in rural U.S. counties, 1976–2004. *Sociology of Education*, 85, 350–372. doi:10.1177/0038040712441376
- Cuban, L. (1986). *Teachers and machines: The classroom use of technology since 1920*.

  New York: Teachers College Press.
- Darvin, R., & Norton, B. (2014). Social class, identity, and migrant students. *Journal of Language, Identity and Education*, 13 (2), 111–117.
- Davies, R., Dean, D., & Ball, N. (2013). Flipping the classroom and instructional technology integration in a college-level information systems spreadsheet course. *Educational Technology Research and Development*, 61, 563–580.
- Deggs, D., Grover, K., & Kacirek, K. (2010). Expectations of adult graduate

- students in an online degree program. College Student Journal, 44(3), 690-699.
- DeNoyelles, A., Zydney, J., & Chen, B. (2014). Strategies for creating a community of inquiry through online asynchronous discussions. *Journal of Online Learning and Teaching*, 10(1), 153-165. Retrieved from
- Denzin, N.K., & Lincoln, Y.S. (2005). Introduction: The discipline and practice of qualitative research. In N.K. Denzin & Y.S. Lincoln (Eds.), *The sage handbook of qualitative research* (2nd ed.). Thousand Oaks, CA: Sage.
- Desrochers, D. M., & Wellman, J. V. (2011). *Trends in College Spending 1999–2009*.

  Washington, DC: Delta Project on Postsecondary Education Costs, Productivity, and Accountability, 52–57.
- Dewar, T. & Whittington, D. (2004). Blended learning research report. *Calliope Learning*, 2(1).
  - http://www.calliopelearning.com/resources/index.html#Research\_Reports
- Dewey, J. (1997). *Experience and education*. New York: Simon and Schuster. (Original work published 1938).
- Diaz, V. & Brown, M. (2010). Blended Learning: A report on the ELI focus session. *Educause Review*, 1-17.
- Dikkers, A.G., Whiteside, A.L., & Lewis, S. (2013). Virtual high school teacher and student reactions to the social presence model. *Journal of Interactive Online Learning*, 12(3), 156-170.
- Dixson, M.D. (2010). Creating effective student engagement in online courses: What do students find engaging? *Journal of the Scholarship of Teaching and Learning*,

- *10*(2), 1-13.
- Dobbins, N., & Rodríguez, C. D. (2013). Providing support for English language learners with behavioral needs. *Intervention in School & Clinic*, 48, 152-158. doi: 10.1177/1053451212454003
- Dörnyei, Z. (2001). New themes and approaches in second language motivation research.

  Annual Review of Applied Linguistics, 21, 43.
- Dörnyei, Z. & Csizér, K. (2002). Some dynamics of language attitudes and motivation:

  Results of a longitudinal nationwide survey, *Applied Linguistics*, 23(4), 421–462.

  https://doi.org/10.1093/applin/23.4.421.
- Downing, C. E., Spears, J., & Holtz, M. (2014). Transforming a course to blended learning for student engagement. *Education Research International*. doi:10.1155/2014/430732
- Draffan, E. A., Rainger, P., & Ltd, F. (2006). A model for the identification of challenges to blended learning. Research in Learning Technology. *Research in Learning Technology*, *14*(1), 55–67. https://doi.org/10.1080/09687760500479787
- Drouin, M. (2008). The relationship between students' perceived sense of community and satisfaction, achievement, and retention in an online course. *Quarterly Review of Distance Education*, 9(3), 267-284.
- Duff, P. (2010). Language socialization into academic discourse communities. *Annual Review of Applied Linguistics*, 30, 169–192.
- Dzakiria, H., Mustafa, C.S., & Abu Bakar, H. (2006). Moving forward with blended learning (BL) as a pedagogical alternative to traditional classroom learning.

- Malaysian Online Journal of Instructional Technology (MOJIT), 3(1), 103-110.
- Egbert, J., Herman, D., and Chang, A. (2014). To flip or not to flip? That's not the question: Exploring flipped instruction in technology supported language learning environments. *International Journal of Computer-Assisted Language Learning and Teaching*, 4(2), 1-10.
- Egbert, J., Herman, D., & Lee, H. (2015). Flipped instruction in english language teacher education: a design-based study in a complex, open-ended learning environment.

  The Electronic Journal for English as a Second Language, 19(2), 1–23.
- Ehlers, U.D., & Schneckenberg, D. (2010). Introduction: Changing cultures in higher education. In U.D. Ehlers & D. Schneckenberg (Eds.), *Changing Cultures in Higher Education: Moving ahead to future learning* (pp. 1-14). New York: Springer.
- Estrada, L., Dupoux, E. & Wolman, C. (2005) The personal-emotional social adjustment of english-language learners to a community college. *Community College Journal of Research and Practice*, 29(7), 557-568.

  http://dx.doi.org/10.1080/10668920590954015
- Evseeva, A., & Solozhenko, A. (2015). Use of flipped classroom technology in language learning. *Social and Behavioral Sciences*, 206, 205-209. doi:10.1016/j.sbspro.2015.10.006.
- Ferreira, A. F., Salinas, J. G., & Morales, S. (2014). Using a task-based approach for supporting a blended learning model for english as a foreign language. *International Journal of Computer-Assisted Language Learning and*

- Teaching, 4(1), 44-62. https://doi.org/10.4018/978-1-4666-8246-7.ch064
- Fotos, S. & Browne, C. (2004). The development of CALL and current options. In S. Fotos, & C. M. Browne (Eds.), *New perspectives on CALL for second language classrooms* (pp. 3-14). Mahwah, NJ: Lawrence Erlbaum Associates.
- Fuchs, C. (2016). "Are you able to access this website at all?" team negotiations and macro-level challenges in telecollaboration, *Computer Assisted Language Learning*, 29(7), 1152-1168, DOI: 10.1080/09588221.2016.1167091
- Fu, D. & Matoush, M. (2006). Writing development and biliteracy. In P.K. Matsuda, C.

  Ortmeier-Hooper & X. You (Eds.), *The Politics of Second Language Writing: In*Search of the Promised Land (pp.5-29). West Lafayette, Indiana: Parlor Press.
- Futch, L., deNoyelles, A., Thompson, K., & Howard, W. (2016). "Comfort" as a critical success factor in blended learning courses. *Online Learning*, 20(3), 140-158. doi:http://dx.doi.org/10.24059/olj.v20i3.978
- Gardner, R. C. (1985). Social psychology and second language learning: The role of attitudes and motivation. London: Edward Arnold.
- Gardner, R. C., & MacIntyre, P. D. (1993). A student's contribution to second language learning. Part II: Affective variables. *Language Teaching*, 26, 1-11.
- Garrett, N. (1991). Technology in the service of language learning: Trends and issues. *Modern Language Journal*, 75, 74-101.
- Garrett, N. (2009). Computer-assisted language learning trends and issues revisited:

  Integrating innovation. *Modern Language Journal*, 93, 719–740. Retrieved from http://www.jstor.org/stable/25612270

- Garrison, D.R. (2011). *E-learning in the 21st century: A framework for research and practice* (2nd ed.). London: Routledge.
- Garrison, D. R. (2009). Implications of online learning for the conceptual development and practice of distance education. *Journal of Distance Education*, 23(2), 93-104.
- Garrison, D. R., Anderson, T., & Archer, W. (2010). The first decade of the community of inquiry framework: A retrospective. *The Internet and Higher Education*, *13*, 5–9. <a href="https://doi.org/10.1016/j.iheduc.2009.10.003">https://doi.org/10.1016/j.iheduc.2009.10.003</a>
- Garrison, D. R., Anderson, T., & Archer, W. (2001). Critical thinking, cognitive presence and computer conferencing in distance education. *American Journal of Distance Education*, 15(1), 7-23.
- Garrison, D. R., Anderson, T. & Archer, W. (2000). Critical inquiry in a text-based environment: Computer conferencing in higher education. *The Internet and Higher Education*, 2(2–3), 1–19.
- Garrison, D.R., Cleveland-Innes, M., Fung, T.S. (2010). Exploring causal relationships among teaching, cognitive and social presence: Student perceptions of the Community of Inquiry Framework. *Internet and Higher Education*, *13*(1), 31-36.
- Garrison, R., & Kanuka, H. (2004). Blended learning: Uncovering its transformative potential in higher education. *The Internet and Higher Education*, 7, 95-105.
- Garrison, D. R., & Vaughan, N. D. (2008). Blended Learning in Higher Education Framework, Principles, and Guidelines. San Francisco, CA Jossey-Bass.
- Gaughan, J. E. (2014). The flipped classroom in world history. *History Teacher*, 47(2), 221-244. Retrieved from

- http://www.societyforhistoryeducation.org/pdfs/F14\_Gaughan.pdf
- Geertz, C. (1973). The Interpretation of Cultures. New York: Basic Books.
- Gillies, R. (2006). Teachers' and students' verbal behaviours during cooperative and small-group learning. *British Journal of Educational Psychology*, 76, 271–287. doi:10.1348/000709905X52337
- Gleason, J. (2013). Dilemmas of blended language learning: Learner and teacher experiences. *The Computer Assisted Language Instruction Consortium Journal*, 30, 323–341.
- Graham, C. (2006). Blended learning systems. Definitions, current trends and future directions. In C. Bonk & C. Graham (Eds.), The handbook of blended learning: Global perspectives, local designs. San Francisco: John Wiley and Sons.
- Graham, C. R., Allen, S., & Ure, D. (2005). Benefits and challenges of blended learning environments. In M. Khosrow-Pour (Ed.), *Encyclopedia of information science* and technology (pp. 253–259). Hershey, PA: Idea Group.
- Graham, C. R. & Robison, R. (2007). Realizing the transformational potential of blended learning: comparing cases of transforming blends and enhancing blends in higher education. In A. G. Picciano & C. D. Dziuban (Eds), *Blended learning: research perspectives* (pp. 83–110). Newburyport, MA: The Sloan Consortium.
- Griffiths, M. E., & Graham, C. R. (2009). The potential of asynchronous video in online education. *Distance Learning*, 6(2), 13–23.
- Grgurović, M. (2014). An application of the Diffusion of Innovations theory to the investigation of blended language learning. *Innovation in Language Learning and*

- *Teaching*, 8(2), 155-170. doi: 10.1080/17501229.2013.789031.
- Gunawardena, C. N. (1995). Social presence theory and implications for interaction and collaborative learning in computer conferences. *International Journal of Educational Telecommunications*, 1(2/3), 147-166.
- Gunawardena, C. N., & McIsaac, M. S. (2004). *Distance education*. In D. H. Jonassen (Ed.), Handbook of research for educational communications and technology (2nd ed., pp. 355–396). Mahwah, NJ: Erlbaum.
- Gunawardena, C.N. & Zittle, F.J. (1997). Social presence as a predictor of satisfaction within a computer-mediated conferencing environment. *American Journal of Distance Education*, 11(3), 8-26, doi: 10.1080/08923649709526970
- Hagedorn, L. S., & Li, R. (2017). English instruction at community colleges: The language bridge to the USA. In Internationalization in Vocational Education and Training (pp. 229–241). Cham, Switzerland: Springer International Publishing.
- Hamdan, N., McKnight, P., McKnight, K., & Arfstrom, K. (2013). *A review of flipped learning*. Retrieved from http://www.flippedlearning.org/review
- Hampel, R., & Stickler, U. (2005). New skills for new classrooms: Training tutors to teach languages online. *Computer Assisted Language Learning*, 18(4), 311-326. https://doi.org/10.1080/09588220500335455.
- Han, Y. J. (2015). Successfully flipping the ESL classroom for learner autonomy. *NYS TESOL Journal*, 2(1), 98-109.
- Hara, N., Bonk, C.J. and Angeli, C. (2000). Content analysis of an online discussion in an applied educational psychology course. *Instructional Science*, 28, 115-52.

- Hill, R.A. (2016). Community college faculty dispositions towards blended learning.
  Doctoral dissertation. Nova Southeastern University. Retrieved from NSUWorks,
  College of Engineering and Computing. http://nsuworks.nova.edu/gscis\_etd/967.
- Hoic-Bozic, N., Mornar, V., & Boticki, I. (2009). A blended learning approach to course design and implementation. Education, *IEEE Transactions on*, 52(1), 19-30.
- Hong, K., & Samimy, K., (2010). The influence of L2 teachers' use of CALL modes on language learners' reactions to blended learning. *CALICO Journal*, 27(2), 328-348. <a href="http://dx.doi.org/10.11139/cj.27.2.328-348">http://dx.doi.org/10.11139/cj.27.2.328-348</a>
- Hostetter, C., & Busch, M. (2013). Community matters: Social presence and learning outcomes. *Journal of the Scholarship of Teaching and Learning*, 13(1), 77-86.
- Hotle, S.L., & Garrow, L.A. (2015). Effects of the traditional and flipped classrooms on undergraduate student opinions and success. *Journal of Professional Issues in Engineering Education and Practice*, 142(1), 1-11.
- Hsieh, J.S., Wu, V., & Marek, M.W. (2017) Using the flipped classroom to enhance EFL learning. *Computer Assisted Language Learning*, 30,1(2), 1-21, doi: 10.1080/09588221.2015.1111910
- Hubackova, S. (2015). Blended learning: New stage in the foreign language teaching. *Social and Behavioral Sciences*, 197, 1957–1961.
- Hubbard, P. (2008). CALL and the future of language teacher education. *CALICO Journal*, 25(252), 175–188. Retrieved from http://www.jstor.org/stable/calicojournal.25.2.175
- Hung, H.T. (2015). Flipping the classroom for English language learners to foster active

- learning. *Computer Assisted Language Learning*, 28(1), 81-96. https://doi.org/10.1080/09588221.2014.967701
- Ice, P., Gibson, A.M., Boston, W. & Becher, D. (2011). An exploration of differences between community of inquiry indicators in low and high disenrollment online courses. *Journal of Asynchronous Learning Networks*, 15(2), 44-69.
- Ice, P., Swan, K., Diaz, S., Kupczynski, L., & Swan Dagen, A. (2010). An analysis of students' perceptions of the value and efficacy of instructors' auditory and text-based feedback modalities across multiple conceptual levels. *Journal of Educational Computing Research*, 43(1), 113-134.
- Ingerson, T. (2011). Helping English language learners (ELLS) achieve success (Passing Grades) in the mainstream classroom. Retrieved from <a href="http://web.b.ebscohost.com.ezproxy.cu-portland.edu/ED518704">http://web.b.ebscohost.com.ezproxy.cu-portland.edu/ED518704</a>.
- Institute of International Education. (2017). Open Doors Report on International Exchange Data.
- Jafarian, K., Soori, A., & Kifipour, R. (2012). The effect of computer assisted language learning (CALL) on EFL high school students' writing achievement. *European Journal of Social sciences*, 27(2), 138-148.
- Jaggars, S.S. (2014). Choosing between online and face-to-face courses: Community college student voices. *American Journal of Distance Education*, 28(1), 27-38.
- Jaggars, S. S. (2011). Online learning: Does it help low-income and underprepared students? *Community College Research Center*, *36*. New York, NY: Columbia University.

- Jang, E.-Y. and Jiménez, R. T. 2011. A sociocultural perspective on second language learner strategies: Focus on the impact of social context. *Theory into Practice*, 50(2), 141–8. doi:10.1080/00405841.2011.558443
- Jensen J. L., Kummer T. A. & Godoy P. D. (2015). Improvements from a flipped classroom may simply be the fruits of active learning. *Life Science Education*, *14*, 1–12.
- Jiménez, R., & Rose, B. (2010). Knowing how to know: Building meaningful relationships through instruction that meets the needs of students learning English. Journal of Teacher Education, 61(5), 403-412.
- Jones, N., & Lau, A. M. S. (2010). Blending learning: Widening participation in higher education. *Innovations in Education and Teaching International*, 47(4), 405-416.
- Jusoff, K. & Khodabandelou, R. (2009). Preliminary study on the role of social presence in blended learning environment in higher education. *International Education Studies*, 2(4), 79-83.
- Kanno, Y., & Kangas, S. E. N. (2014). "I'm not going to be, like, for the AP": English language learners' limited access to advanced college-preparatory courses in high school. *American Educational Research Journal*, *51*, 848–878. doi:10-3102/0002831214544716.
- Kanuka, H., Liam, R. & Laflamme, E. (2007). The influence of instructional methods on the quality of online discussion. *British Journal of Educational Technology*, 38(2), 260-271.
- Kern, R. and Warschauer, M. (2000). Theory and practice of network-based language

- *teaching*, in M. Warschauer and R. Kern (eds.), Network-Based Language

  Teaching: Concepts and Practice, Cambridge University Press, Cambridge, UK,

  1–19.
- Kessler, G. (2018). Technology and the future of language teaching. *Foreign Language*Annals, 51(1), 205–218.
- Kilgore, W., & Lowenthal, P. R. (2015). The Human Element MOOC: An experiment in social presence. In R. D. Wright (Ed.), *Student-teacher interaction in online learning environments* (pp. 389-407). Hershey, PA: IGI Global.
- Kim, E., Byun, H., & Lee, O. (2012). Course redesign using Flipped Instructional Model.

  Retrieved from http://scholar.googleusercontent.com/scholarq=cache:9G5K

  AGn8OD8J:scholar.google.com/+Course+Redesign+using+Flipped+Instructional

  +Model&hl=en&as\_sdt=0,48.
- King, G., Keohane, R. O. and Verba, S. (1994) *Designing Social Inquiry: Scientific Inference in Qualitative Research*. Princeton University Press: Chichester.
- Kitchen, D., & McDougall, D. (1998). Collaborative learning on the Internet. *Journal of Educational Technology Systems*, 27(3), 245.
- Koh, H. L., Herring, S. C., & Hew, K. F. (2010). Project-based learning and student knowledge construction during asynchronous online discussion. *The Internet and Higher Education*, 13, 284-291.
- Kostka, I. & Brinks Lockwood, R. (2015). What's on the internet for flipping English language instruction? *The Electronic Journal for English as a Second Language*, 19(2), 1-12.

- Krashen, S. (2014). Case histories and the comprehension hypothesis. *TESOL Journal*, 9.

  Retrieved from
  - http://www.sdkrashen.com/content/articles/skrashen\_case\_histories.pdf
- Kucuk S. & Sahin, I. (2013). From the perspective of community of inquiry framework:

  An examination of Facebook uses by Preservice teachers as a learning
  environment. *TOJET: The Turkish Online Journal of Educational Technology*, 12,
  142–156.
- Kvashnina, O. S.& Martynko, E. A. (2016). Analyzing the potential of flipped classroom in ESL teaching. *International Journal of Emerging Technologies in Learning*, 11(3), 71–73. doi: http://10.3991/ijet.v11i03.5309.
- Lamport, M. A., & Bartolo, P. J. (2012). Student perceptions of online instructional practices that enhance connectedness: Themes toward the development of an instrument. *Journal of Instructional Research*, 1(26), 23-33.
- Lantolf, J. P., & Thorne, S. L. (2006). Sociocultural theory and the genesis of second language development. Oxford: Oxford University Press.
- Lapointe, D.K., & Gunawardena, C.N. (2004). Developing, testing and refining of a model to understand the relationship between peer interaction and learning outcomes in computer-mediated conferencing. *Distance Education*, 25(1) 83-106.
- Lave, J. (1991). Chapter 4: Situating learning in communities of practice. (pp. 63-82). InLauren Resnick, Levine B., M. John, Stephanie Teasley & D. (eds.), *Perspectives*on Socially Shared Cognition. American Psychological Association.
- Leafman, J.S., Mathieson, K. (2014). Comparison of student and instructor perceptions of

- social presence. Journal of Educators Online, 11(2), p. 1-27.
- Leafman, J.S., Mathieson, K., & Ewing, E. (2013). Students' perceptions of social presence and attitudes toward social media. *International Journal of Higher Education*, 2(1), 67-77.
- Lee, G., & Wallace, A. (2017). Flipped learning in the English as a foreign language classroom: Outcomes and perceptions. *TESOL Quarterly*, 52(1), 62-84. doi:10.1002/tesq.372
- Lee, S.J. & Huang, K. (2018). Online Interactions and social presence in online learning. *Journal of Interactive Learning Research*, 29(1), 113-128.
- Leh, A. S. (2001). Computer-mediated communication & social presence in a distance learning environment. *International Journal of Educational Telecommunications*, 7(2), 109-128.
- Leonard, D.A., & DeLacey, B.J. (2002). Designing hybrid online/in-class learning programs for adults. Harvard Business School.

  http://calsca.com/Writings/walters\_blended\_learning.htm.
- Levy, D. (2017). Online, blended and technology-enhanced learning: Tools to facilitate community college student success in the digitally-driven workplace.

  Contemporary Issues in Education Research, 10(4), 255–262.
- Li, N. (2013). Seeking best practices and meeting the needs of the English language learners: Using second language theories and integrating technology in teaching. *Journal of International Education Research*, 9, 217-222.
- Lin, J. M., Wang, P., & Lin, I. (2012). Pedagogy technology: A two-dimensional model

- for teachers' ICT integration. *British Journal of Educational Technology*, 43(1), 97-108. doi:10.1111/j.1467-8535.2010.01159.x
- Liu, S. Y., Gomez, J., & Yen, C. J. (2009). Community college online course retention and final grade: Predictability of social presence. *Journal of Interactive Online Learning*, 8(2), 165-182.
- Liu, M. H., & Kleinsasser, R. C. (2015). Exploring EFL teachers' CALL knowledge and competencies: In-service program perspectives. *Language Learning & Technology*, 19, 119–138.
- Lofland, J., Snow, D., Anderson, L., & Lofland, L.H. (2006). *Analyzing social settings:*A guide to qualitative observation and analysis. Belmont, CA:

  Wadsworth/Thomson Learning.
- Lowenthal, P. R., & Dunlap, J. C. (2014). Problems measuring social presence in a community of inquiry. *E-Learning and Digital Media*, 11(1), 20–31.
- Lyons, A., Reysen, S., & Pierce, L. (2012). Video lecture format, student technological efficacy, and social presence in online courses. *Computers in Human Behavior*, 28(1), 181-186.
- Marshall, C., & Rossman, G. B. (1989). *Designing Qualitative Research*. Newbury Park, CA: Sage.
- Mate-Martensen, M. (2009). ESL teacher and student perspectives on technology in the community college classroom. In K. M. Bailey & M. G. Santos (Eds.), Research on ESL in U. S. community colleges: People, programs, and potential. Ann Arbor, MI: University of Michigan Press.

- McMillan, D.W. and Chavis, D.M. (1986) Sense of community: A definition and theory. *Journal of Community Psychology, 14*, 6-23.
- McGee, P., & Reis, A. (2012). Blended course design: A synthesis of best practices. *Journal of Asynchronous Learning Networks*, 16(4), 7-22.
- McGuire, B.F. (2016). Integrating the intangibles into asynchronous online instruction: Strategies for improving interaction and social presence. *The Journal of Effective Teaching*, *16*(3), 62 75.
- McKinney, L. & Backscheider-Burridge, A. (2015). Helping or hindering? The effects of loans on community college student persistence. *Research in Higher Education*, 56(4), 299-324.
- Means, B., Toyama, Y., Murphy, R., Bakia, M., & Jones, K. (2009). Evaluation of evidence-based practices in online learning: A meta-analysis and review of online learning studies. *US Department of Education*. Retrieved from http://files.eric.ed.gov/fulltext/ED505824.pdf
- Mehring J.G. (2016). Present research on the flipped classroom and potential tools for the EFL classroom. *Computers in the Schools 33*(1), 1–10.
- Mendieta Aguilar, J. A. (2012). Blended learning and the language teacher: A literature review. *Colombian Applied Linguistics Journal*, *14*(2), 163. <a href="https://doi.org/10.14483/udistrital.jour.calj.2012.2.a10">https://doi.org/10.14483/udistrital.jour.calj.2012.2.a10</a>.
- Meltzer, J., & Hamann, E. (2004). Meeting the needs of adolescent English language learners for literacy development and content area learning, Part 1: Focus on motivation and engagement. Providence, RI: The Education Alliance at Brown

### University

- Miller, F.A., & Alvarado, K. (2005). Incorporating documents into qualitative nursing research. *Journal of nursing scholarship: an official publication of Sigma Theta Tau International Honor Society of Nursing*, 37(4), 348-53.
- Mirriahi, N., Alonzo, D. & Fox, B. (2015). A blended learning framework for curriculum design and professional development. *Research in Learning Technology*, 23, 1–14. doi: 10.3402/rlt.v23.28451
- Mishan, F. (2005). *Designing authenticity into language learning materials*. Bristol, UK: Intellect.
- Moore, M.G. (1993). Theory of Transactional Distance," in Theoretical Principles of Distance Education, ed. Desmond Keegan (New York: Routledge).
- Mullin, C. M. (2012). Student success: Institutional and individual perspectives.

  Community College Review, 40, 126-144. doi: 10.1177/0091552112441501
- Ndlovu, M. & Mostert, I. (2018). Teacher perceptions of moodle and throughput in a blended learning programme for in-service secondary school mathematics teachers, *Africa Education Review*, *15*(2), 131-151. doi: 10.1080/18146627.2016.1241667
- Neumeier, P. (2005). A closer look at blended learning: Parameters for designing a blended learning environment for language teaching and learning. *ReCALL*, *17*(2), 163–178. https://doi.org/10.1017/S0958344005000224.
- Nikula, T. (2002). Teacher talk reflecting pragmatic awareness: A look at EFL and content-based classroom settings. *Pragmatics*, 12(4), 447–67.

- Northern Virginia Community College (NOVA) Office of Institutional Effectiveness.

  (2012-2017). Northern Virginia Community College (NVCC) Factbook, 1-157.

  Retrieved from <a href="https://www.nvcc.edu/oir/files/factbooks2012-2017.pdf">https://www.nvcc.edu/oir/files/factbooks2012-2017.pdf</a>
- Northern Virginia Regional Commission. (n.d). Retrieved from <a href="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion.org/DocumentCenter/View/1734/section\_03\_community\_profile?bidId="https://www.novaregion\_03\_community\_prof
- Office of Institutional Research (2014-2016). NVCC ESL Enrollment Profile.
- Oliver, M., & Trigwell, K. (2005). Can "blended learning" be redeemed? *E-learning*, 2(1), 22-45. Retrieved from <a href="http://journals.sagepub.com.mutex.gmu.edu/doi/pdf/10.2304/elea.2005.2.1.17">http://journals.sagepub.com.mutex.gmu.edu/doi/pdf/10.2304/elea.2005.2.1.17</a>
- Osguthorpe, R. T., & Graham, C. R. (2003). Blended learning systems: Definitions and directions. *Quarterly Review of Distance Education*, 4(3), 227–234.
- Paepe, L.D., Zhu, C., & DePryck, K. (2018). Drop-out, Retention, Satisfaction and Attainment of Online Learners of Dutch in Adult education. *International Journal on e-Learning*, 17(3), 303-323.
- Palloff, R., & Pratt, K. (1999). Building learning communities in cyberspace: Effective strategies for the online classroom. San Francisco, CA: Jossey-Bass.
- Palmer, S., Holt, D., & Bray, S. (2008). Does the discussion help? The impact of a formally assessed online discussion on final student results. *British Journal of Educational Technology*, 39(5), 847-858.
- Palmer, M. M., Shaker, G., & Hoffmann-Longtin, K. (2014). Despite faculty skepticism:

  Lessons from a graduate-level seminar in a hybrid course environment. *College*

- *Teaching*, 62(3), 100-106. doi:10.1080/87567555.2014.912608.
- Panahi, P., Birjandi, P., & Azabdaftari, B. (2013). Toward a sociocultural approach to feedback provision in L2 writing classrooms: The alignment of dynamic assessment and teacher error feedback. *Language Testing in Asia*, *3*(13), 1-10.
- Partridge, H., Ponting, D., & McCay M. (2011). Good practice report: Blended learning australian learning and teaching council. Available at http://eprints.qut.edu.au/47566/1/47566.pdf
- Patton, M. Q. 1990. *Qualitative evaluation and research methods*, (2nd ed.). Newbury Park, CA: Sage.
- Peirce, B. (1995). Social identity, investment, and language learning. *TESOL Quarterly*, 29(1), 9-31. doi:10.2307/3587803.
- Pennington, M. 2004. Cycles of innovation in the adoption of information technology: A view for language teaching. *Computer Assisted Language Learning*, 17(1), 7–33.
- Pezalla, A. E., Pettigrew, J., & Miller-Day, M. (2012). Researching the researcher as instrument: An exercise in interviewer self-reflexivity. *Qualitative Research*, *12*(2), 165–185. doi: 10.1177/1468794111422107.
- Pichon, H.W. (2016). Developing a sense of belonging in the classroom: Community college students taking courses on a four-year college campus. *Community College Journal of Research and Practice*, 40(1), 47-59. doi: 10.1080/10668926.2014.964429
- Pinto-Llorente, A. M., Sánchez-Gómez, M. C., García-Peñalvo, F. J., & Casillas-Martín, S. (2017). Students' perceptions and attitudes towards asynchronous technological

- tools in blended-learning training to improve grammatical competence in English as a second language. *Computers in Human Behavior*, 72, 632–643. https://doi.org/10.1016/J.CHB.2016.05.071
- Porter, W. W., Graham, C.R., Bodily, R.G., & Sandberg, D.S. (2016). A qualitative analysis of institutional drivers and barriers to blended learning adoption in higher education. *The Internet and Higher Education*, 28 17–27.
- Punch, K. (2014). Introduction to social research (3rd ed.). London, UK: Sage.
- Radosavlevikj, N. (2015). Developing a blended learning strategy: Motivating university students to learn English by implementing blended learning activities (instructional media and pedagogical considerations). *Psychological and creative approaches to language teaching*, 14-29.
- Rance-Roney, J.A. (2009). Best practices for adolescent ELLs. *Educational Leadership*, 66(7), 32–37.
- Rance-Roney, J. A. (2010). Reconceptualizing interactional groups: Grouping schemes for maximizing language learning. *English Teaching Forum*, 10(1), 20-26.
- Ravenna, G., Foster, C., & Bishop, C. (2012). Increasing student interaction online: A review of the literature. *Journal of Technology and Teacher Education*, 20(2), 177–203.
- Richards, L. (2015). *Handling qualitative data: A practical guide (3rd ed.)*. London, UK: Sage.
- Riel, J., Lawless, K.A. & Brown, S.W. (2016). Listening to the teachers: Using weekly online teacher logs for ROPD to identify teachers' persistent challenges when

- implementing a blended learning curriculum. *Journal of Online Learning Research*, 2(2), 169-200.
- Roe, C. (2009). Community college ESL learners' access to and perspectives on technology. In K. M. Bailey & M. G. Santos (Eds.), Research on ESL in U. S. community colleges: People, programs, and potential (pp. 62-81). Ann Arbor, MI: University of Michigan Press.
- Roschelle J., & Teasley, S.D. (1995) The construction of shared knowledge in collaborative problem solving. In: O'Malley CE (ed) Computer-supported collaborative learning. Springer, Berlin, pp 69–97.
- Rourke, L., Anderson, T., Garrison, R. D., & Archer, W. (2001). Assessing social presence in an asynchronous text-based computer conferencing. *Journal of Distance Education*, 14, 51-70.
- Rovai, A.P. & Jordan, H.P. (2004). Blended learning and sense of community: A comparative analysis with traditional and fully online graduate courses.

  International Review of Research in Open and Distance Learning, 5(2), 1-17.
- Rovai, A.P. (2002). Building a sense of community at a distance. *International Review of Research in Open and Distance Learning*, 3(1), 1-16.
- Rubin, B., Fernandes, R. & Avgerinou, M. (2013). The effects of technology on the community of inquiry and satisfaction with online courses. *Internet and Higher Education*, 17, 48-57.
- Rubio, F., Thomas, J. M., & Li, Q. (2018). The role of teaching presence and student participation in Spanish blended courses. *Computer Assisted Language Learning*,

- *31*(3), 226-250.
- Ryan, S., Kaufman, J., Greenhouse, J., She, R., & Shi, J. (2016). The effectiveness of blended online learning courses at the community college level. *Community College Journal of Research & Practice*, 40(4), 285-298.

  doi:10.1080/10668926.2015.1044584
- Ruiz, M., Palmer, J., & Fortanet, I. (2005). Carencias comunicativas orales detectadas por el profesorado de lengua inglesa. In E. Hernández & L. Sierra (Eds.), *Lenguas para Fines Específicos (VIII) Investigación y Enseñanza* (pp. 63–74). Madrid: Universidad de Alcalá de Henares.
- Šafranj, J. (2013). Using information technology in English language
  learning procedure: Blended learning. *Social and Behavioral Sciences*, 83, 514–521. https://doi.org/10.1016/j.sbspro.2013.06.099
- Sahin-Kizil, A. (2014). Blended instruction for EFL learners: Engagement, learning and course satisfaction. *Japanese Association for Language Teaching Computer*Assisted Language Learning Journal, 10(3), 175-188.
- Saldaña, J. (2016). The coding manual for qualitative researchers. London, UK: Sage.
- Santiago, D. (1992). *The Somebody*. In Edward Simmen (Ed.), North of the Rio Grande:

  The Mexican-American experience in short Fiction (pp. 212-221). New York,

  NY: Mentor.
- Scherer Bassani, P. B. (2011). Interpersonal exchanges in discussion forums: A study of learning communities in distance learning settings. *Computers & Education*, 56(4), 931- 938. http://dx.doi.org/10.1016/j.compedu.2010.11.009

- Scida, E. & Saury, R. E. (2006). Hybrid courses and their impact on student and classroom performance: A case study at UVA. *CALICO Journal*, *23*(3), 517–531. Retrieved from http://www.jstor.org/stable/24156352.
- Seidman, I. E. 1991. *Interviewing as qualitative research: A guide for researchers in education and the social sciences*, New York: Teachers College Press.
- Senior, R. (2010). Connectivity: A framework for understanding effective language teaching in face-to-face and online learning communities. *Regional Language Center Journal*, 41(2), 137-147.
- Sfard, A. (1998). On two metaphors for learning and the dangers of choosing just one. *Educational Researcher*, 27(2), 4-13.
- Sharma, P. and B. Barrett. (2007). Blended learning: Using technology in and beyond the language classroom. Oxford: Macmillan.
- Sharpe, R., Benfield G., Roberts, G., & Francis, R. (2006) The Undergraduate

  Experience of Blended E-learning: A Review of UK Literature and Practice.

  Available at

  <a href="http://www.heacademy.ac.uk/assets/York/documents/ourwork/research/literature\_reviews/blended\_elearning\_exec\_summary\_1.pdf">http://www.heacademy.ac.uk/assets/York/documents/ourwork/research/literature\_reviews/blended\_elearning\_exec\_summary\_1.pdf</a>.
- Shea, P., & Bidjerano, T. (2014). Does online learning impede degree completion? A national study of community college students. *Computers & Education*, 75, 103-111. doi:10.1016/j.compedu.2014.02.009.
- Silverman, R. & Hines, S. (2009). The effects of multimedia enhanced instruction on the vocabulary of English-language learners and non-English-language learners in

- pre-kindergarten through second grade. *Journal of Educational Psychology*, 101(2), 305–314.
- Shawer, S. (2010). Communicative-based curriculum innovations between theory and practice: Implications for EFL curriculum development and student cognitive and affective change. The Curriculum Journal, 21(3), 333-359.

  doi:10.1080/09585176.2010.506802
- Shimamoto, D. (2012). *Implementing a flipped classroom: An instructional module*.

  Presented at the Technology, Colleges, and Community Worldwide Online
  Conference, Honolulu, HI. Retrieved from

  http://scholarspace.manoa.hawaii.edu/handle/10125/22527.
- Slavkov, N. (2015). Sociocultural theory, the L2 writing process, and google drive: Strange bedfellows? *TESL Canada Journal*, *32*(2), 80. doi:https://doi.org/10.18806/tesl.v32i2.1209
- Slomanson, W. R. (2014). Blended learning: A flipped classroom experiment. *Journal of Legal Education*, 64(1), 93. Retrieved from <a href="http://ssrn.com/abstract=2381282">http://ssrn.com/abstract=2381282</a>
- Snart, J. (2017). Hybrid learning at the community college. *New Directions for Teaching* and Learning, 149, 59-67.
- So, H.J. & Brush, T. A. (2008). Student perceptions of collaborative learning, social presence and satisfaction in a blended learning environment: Relationships and critical factors. *Computers & Education 51*, 318-336.
- Sorden, S. D., & Munene, I. I. (2013). Constructs related to community college student

- satisfaction in blended learning. *Journal of Information Technology Education*Research, 12(12), 251–270.
- Staker H. & Horn, M. (2012). *Classifying K-12 blended learning*. Retrieved from the Innosight Institute website http://files.eric.ed.gov/fulltext/ED535180.pdf
- Stracke, E. (2007). A road to understanding: A qualitative study into why learners drop out of a blended language learning (BLL) environment. *ReCALL*, *19*(1), 57-58 <a href="https://doi.org/10.1017/S0958344007000511">https://doi.org/10.1017/S0958344007000511</a>
- Sung, E. & Mayer, R. (2012). Five facets of social presence in online distance education.

  Computers in Human Behavior, 28, 1738-1747.
- Swan, K., Garrison, D. R. & Richardson, J. C. (2009). A constructivist approach to online learning: The community of inquiry framework. In Payne, C. R. (Ed.) *Information Technology and Constructivism in Higher Education: Progressive Learning Frameworks*. Hershey, PA: IGI Global, 43-57.
- Takeuchi, O., Griffiths, C., & Coyle, D. (2007). Applying strategies to contexts: The role of individual, situational, and group differences. In A. D. Cohen & E. Macaro (Eds.), *Language learner strategies* (pp. 69–92). New York: Oxford University Press.
- Tay, H. Y. (2016). Investigating engagement in a blended learning course. *Cogent Education*, 3(1), 1-13. doi: 10.1080/2331186X.2015.1135772.
- Tayebinik, M., & Puteh, N. (2012). Blended and e-learning. *International Magazine on Advances in Computer Science and Telecommunications*, 3(1), 103–110.
- Tavangarian, D., Leypold, M. E., Nölting, K., Röser, M., & Voigt, D. (2004). Is e-

- Learning the solution for individual learning? *Electronic Journal of E-Learning*, 2, 273–280.TESOL International Association. (2018). English as a Foreign Language. Retrieved from <a href="http://www.tesol.org/connect/interest-sections/english-as-a-foreign-language">http://www.tesol.org/connect/interest-sections/english-as-a-foreign-language</a>
- Tetnowski, J. 2015. Qualitative case study research design. *Perspectives on Fluency and Fluency Disorders*, 25(1), 39-45. doi:10.1044/ffd25.1.39.
- Thorne, K. (2003). *Blended learning: How to integrate online and traditional learning*. London: Kogan Page.
- Tinto, V. (1993) Leaving College: Rethinking the Causes and Cures of Student Attrition, 2nd(ed.), Chicago: University of Chicago Press.
- Tran, T.T. (2012). The Perceptions and Attitudes of International Students Towards

  Plagiarism. *The Australian Council Journal for Private Education and Training,*1(2), 13-21.
- Tsai, C. (2010). Facilitating students to earn computing certificates via blended learning in online problem solving environment: A cross-course-orientation comparison.

  International Journal of Information and Communication Technology Education, 6(2), 11-23.
- Tsang, W. K. (2004). Teacher's personal practical knowledge and interactive decisions.

  \*Language Teaching Research, 8(2), 163-198.
- Tu, C. H., Sujo-Montes, L., Yen, C. J., Chan, J. Y., & Blocher, M. (2012). The integration of personal learning environments & open network learning environments. *TechTrends*, 56(3), 13-19. doi:10.1007/s11528-012-0571-7

- U.S. Census Bureau (2018). *Census Quickfacts*. Retrieved from <a href="https://www.census.gov/quickfacts/VA">https://www.census.gov/quickfacts/VA</a>
- U.S. Census Bureau (2017). *American Community Survey*. Retrieved from <a href="https://www.census.gov/programs-surveys/acs">https://www.census.gov/programs-surveys/acs</a>
- Vanek, J., King, K., & Bigelow, M. (2018). Social presence and identity: Facebook in an English language classroom. *Journal of Language, Identity, and Education*.
- Van Veen, B. (2013). Flipping Signal-Processing Instruction [SP Education]. *Signal Processing Magazine*, *IEEE*, *30*(6), 145-150. doi: 10.1109/MSP.2013.2276650
- Vaughan, N. (2014). Student engagement and blended learning: Making the assessment connection. *Education Sciences*, *4*(4), 247–264.
- Vesely, P., Bloom, L., & Sherlock J. (2007). Key elements of building online community: Comparing faculty and student perceptions. *Journal of Online Learning and Teaching*, 3(3), 234-246.
- Vernadakis, N., Antoniou, P., Giannousi, M., Zetou, E., & Kioumourtzoglou, E. (2011).

  Comparing hybrid learning with traditional approaches on learning the Microsoft

  Office Power Point 2003 program in tertiary education. *Computers & Education*,

  56(1), 188–199.
- Volchok, E. (2018) Toward a predictive model of community college student success in blended classes. *Community College Journal of Research and Practice*, 42(4), 274-288, DOI: 10.1080/10668926.2017.1287607
- Vygotsky, L. S. (1978). *Mind in society: The development of higher psychological* processes Cambridge, Mass.: Harvard University Press.

- Walshe, C., Ewing, G., & Griffiths, J. (2012). Using observation as a data collection method to help understand patient and professional roles and actions in palliative care settings. *Palliative Medicine* 26(8), 1048–1054.
- Wang, M.J. (2010). Online collaboration and offline interaction between students using asynchronous tools in blended learning. *Australasian Journal of Educational Technology*, 26(6), 830-846.
- Wanner, T., & Palmer, E. (2015). Personalising learning: Exploring student and teacher perceptions about flexible learning and assessment in a flipped university course.

  \*Computers & Education\*, 88, 354–369. doi:10.1016/j.compedu.2015.07.008
- Warren, E., & Miller, J. (2015). Supporting English second-language learners in disadvantaged contexts: Learning approaches that promote success in mathematics. *International Journal of Early Years Education*, 23(2), 192-208.
- Warschauer, M. 1996. Comparing face-to-face and electronic discussion in the second language classroom. *CALICO Journal*, 13: 7–26.
- Watanabe, Y. & Swain, M. (2008). Perception of learner proficiency: Its impact on the interaction between an ESL learner and her higher and lower proficiency partners.
  Language Awareness, 17(2), 115-130, DOI: <a href="https://doi.org/10.1080/09658410802146651">10.1080/09658410802146651</a>
- Webb, M., Doman, E., & Pusey, K. (2014). The flipped and non-flipped EFL classroom:

  Initial reactions from Chinese university students. *Thai TESOL Journal*, 27(1),

  13-43.
- Weinel, M., Bannert, M., Zumbach, J., Hoppe, H. U., & Malzahn, N. (2011). A closer look on social presence as a causing factor in computer-mediated collaboration.

- Computers in Human Behavior, 27(1), 513-521.
- Wen, Q. F. (2008). On the output-driven hypothesis and reform of English-skill courses for English majors. *Foreign Language World* 2, 2-9.
- Wenger, E. (1998). *Communities of practice: Learning, meaning, and identity*.

  Cambridge, U.K.; New York, N.Y.: Cambridge University Press.
- Whiteman, J. A. M. (2002). Interpersonal communication in computer mediated learning.

  Retrieved from http://eric. ed.gov/PDFS/ED465997.pdf
- Whiteside, A. L. (2015). Introducing the social presence model to explore online and blended learning experiences. *Online Learning*, 19(2), 53–72. Retrieved from <a href="http://mutex.gmu.edu/login?url=http://search.ebscohost.com/login.aspx?direct=tru">http://mutex.gmu.edu/login?url=http://search.ebscohost.com/login.aspx?direct=tru</a>
  e&db=ehh&AN=101822892&site=ehost-live
- Wicks, D. A., Craft, B. B., Mason, G. N., Gritter, K., & Bolding, K. (2015). An investigation into the community of inquiry in blended classrooms by a faculty learning community. *The Internet and Higher Education*, 25, 53–62.
- Woltering, V., Herrler, A., Spitzer, K., Spreckelsen C. (2009). Blended learning positively affects students' satisfaction and the role of the tutor in the problem-based learning process: Results of a mixed-method evaluation. *Advances in Health Sciences Education*, 14(5), 725-738.
- Woodley, C. & Meredith, C. (2012). Supporting student transition through social media.

  \*American Journal of Distance Education, 26, 86-95.
- Wu, S., & Alrabah, S. (2009). A cross-cultural study of Taiwanese and Kuwaiti EFL students' learning styles and multiple intelligences. *Innovations in Education and*

- *Teaching International*, 46(4), 393-403. doi:10.1080/14703290903301826
- Wu, H., Tennyson, R.D., & Hsia, T. (2010). A study of student satisfaction in a blended e-learning environment system. *Computers in Education*, 55, 155-164.
- Xu, D. & Jaggars, S. S. (2011). Online and hybrid course enrollment and performance in Washington state and technical colleges. *Community College Research Center*,31. Teachers College, Columbia University.
- Xu, D., & Jaggars, S. S. (2013). The impact of online learning on students' course outcomes: Evidence from a large community and technical college system.
   Economics of Education Review, 37, 46-57.
   doi:10.1016/j.econedurev.2013.08.001
- Yalçınkaya, D. (2015). Why is blended learning for vocationally oriented language teaching? *Social and Behavioral Sciences*, *174*, 1061-1068. https://doi.org/10.1016/j.sbspro.2015.01.795
- Yang, Y.F. (2014) Preparing language teachers for blended teaching of summary writing.

  Computer Assisted Language Learning, 27(3), 185-206.
- Yin, R.K. (2013). Validity and generalization in future case study evaluations. *Evaluation*, 19(3), 321 332. https://doi.org/10.1177/1356389013497081
- Yin, R.K. (2003). *Case study research: Design and methods*. Sage: Thousand Oaks, California.
- Youmans, R. J. 2011. Does the Adoption of Plagiarism-detection Software in Higher Education Reduce Plagiarism? *Studies in Higher Education*, *36*(7), 749. 761.10.1080/03075079.2010.523457

- Yu, S., & Lee, I. (2016). Exploring Chinese students' strategy use in a cooperative peer feedback writing group. *System*, *58*, 1–11.
- Yueh-ching, C. (2016). Discourses, identities and investment in english as a second language learning: Voices from two U.S. community college.

  students. *International Journal of Education & Literacy Studies*, 4(4), 38-47.
- Yuen, A. H. K. (2011). Exploring teaching approaches in blended learning. *Research and Practice in Technology Enhanced Learning*, 6(1), 3-23. Retrieved from <a href="http://www.apsce.net/RPTEL/RPTEL2011MarIssue-Article1">http://www.apsce.net/RPTEL/RPTEL2011MarIssue-Article1</a> %28p3-p23%29R1.pdf
- Zapata, G. C. (2004). Computer assisted language learning second language instructors and CALL: A multidisciplinary research framework. *Computer Assisted Language Learning*, 17(17), 3–4. https://doi.org/10.1080/0958822042000319610
- Zappe, S., Leicht, R., Messner, J., Litzinger, T., & Lee, H.W. (2009). Flipping the classroom to explore active learning in a large undergraduate course. Paper presented at 2009 Annual Conference & Exposition, Austin, TX. Retrieved from https://peer.asee.org/4545
- Zhao, N. & McDougall, D. (2008). Cultural influences on Chinese students' asynchronous online learning in a Canadian university. *The Journal of Distance Education*, 22(2), 59-79.
- Zielinski, D.E. (2017). The use of collaboration, authentic learning, linking material to personal knowledge, and technology in the constructivist classroom: Interviews with community college faculty members. *Community College Journal of*

Research and Practice, 41(10), 668-686. doi: 10.1080/10668926.2016.1220338

### **BIOGRAPHY**

Allison Omohundro graduated from Oakton High School, Vienna, Virginia, in 2001. She received her Bachelor of Liberal Studies from Longwood University in 2005 and Masters in Special Education in 2006. She has been employed as a teacher in Fairfax County Public Schools for ten years.